

Pritchard, David John (1999) The development of multianalyte biosensors based upon patterning biological molecules using a photoactivatable compound. PhD thesis

http://theses.gla.ac.uk/6644/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

THE DEVELOPMENT OF MULTIANALYTE BIOSENSORS BASED UPON PATTERNING BIOLOGICAL MOLECULES USING A PHOTOACTIVATABLE COMPOUND

A Thesis submitted to the Faculty of Engineering

of the University of Glasgow

for the degree of

Doctor of Philosophy

by

David John Pritchard

March 1999

DECLARATION

The author declares that the work of this thesis has

not been previously submitted for any degree or award

Signature _____ Date <u>30/3/44</u>

SUMMARY

The primary aim of this work was the construction of a multianalyte amperometric immunosensor. In the course of this work a new technique was developed for the selective immobilisation of organic molecules at defined areas on a variety of surfaces, including silica, glass, gold and carbon. The immobilisation technique comprised of binding avidin to the surface, and addition of photobiotin (a photactivatable biotin analogue) which bound to the avidin. This resulted in surfaces that were resistant to non-specific binding, and that allowed site specific immobilisation of biological molecules by exposure of selected areas of the surface to light, in the presence of the species to be immobilised. The application of this patterning technique was demonstrated for the immobilisation of antibodies, antigens, enzymes, lectins and nucleic acids.

In order to further demonstrate the application of this technology, a simultaneous quantitative electrochemical immunoassay was developed for the measurement of follicle stimulating hormone and luteinising hormone in human plasma. The assay showed good specificity and sensitivity, and values obtained for clinical samples correlated well with results obtained from an established commercial assay. This was the first report of an electrochemical multianalyte immunosensor for high molecular weight analytes.

Further work demonstrated the applicability of the selective immobilisation method to the construction of multianalyte enzymic biosensors and immunosensors, using ultramicroelectrodes 10 µm in diameter. It was shown that previously reported advantages of using microspot immunoassays were not applicable to these amperometric assays, as non-planar diffusion of species to ultramicroelectrodes effectively greatly increased the

i

instruments "field of view". Investigations into "cross-talk" between ultramicroelectrodes showed that it would be possible to fit over 350 sensors into a 1mm² area.

Finally, the applicability of the immobilisation method in the detection of multiple nucleic acid sequences was also demonstrated, and an assay was developed for the detection of the Factor V_{leiden} genetic polymorphism.

The model systems described above, demonstrated both the efficiency of the protein patterning techniques developed, and its application to multianalyte sensing systems. The technology that has been developed is applicable to a generic range of sensors, and fabrication methods are compatible with the requirements of larger scale production and miniaturisation, utilising processes common in the microelectronics industry.

PUBLICATIONS ARISING FROM THIS WORK

Morgan, H., Pritchard, D. J. & Cooper, J. M. "Selective immobilisation and patterning of molecules on a surface.", UK Patent Application No 9325100.7 (1993)

Pritchard, D. J., Morgan, H. & Cooper, J. M. "Micron scale patterning of biological molecules.", *Angew. Chemie. Int. Ed. Engl.*, **35**, 1-93 (1995)

Pritchard, D. J., Morgan, H. & Cooper, J. M. "Simultaneous determination of follicle stimulating hormone and luteinising hormone using a multianalyte immunosensor.", *Anal. Chim. Acta.*, **310**, 251-256 (1995)

Pritchard, D.J., Morgan, H., and Cooper, J.M. "Patterning and regeneration of surfaces with antibodies". *Anal. Chem.* 67, 3605-3607, (1995)

Morgan, H., Pritchard, D. J. & Cooper, J. M. "Photo-patterning of sensor surfaces with biom0lecular structures: Characterisation using AFM and fluorescence microscopy.", *Biosens. Bioelectron*, **10**, 841-846 (1995)

ACKNOWLEDGEMENTS

My sincere thanks go to my supervisors Dr J. M. Cooper and Dr H. Morgan for all the assistance they gave me in the course of this work.

I am indebted to other members of the Bioelectronics group, particularly Mr W. Monaghan, Ms M. Robertson, Dr K. Ryder, Mr A. Griffith and Ms F. Young. I would also like to express my appreciation of the assistance in photolithographic techniques provided by Mrs Lois Hobbs.

I would like to thank Mr. S. Jones, Clinical Chemistry, Southmead Hospital, Bristol, UK. for assistance with comparison of the immunoassay results with those from an established method. I would also like to thank Dr David Stirling Of Edinburgh Royal Infirmary for the provision of DNA extracts from people of known Factor V genotype.

Finally, I wish to thank my family and friends for their support, especially my wife Catherine without whose sacrifice and encouragement this work would not have been possible.

Contents

| Section | Page |
|------------------------------------------------------|------|
| Summary | i |
| Acknowledgments | iii |
| Contents | iv |
| Index of Figures | xi |
| Index of Tables | XV |
| Abbreviations | xvi |
| Chapter 1 | 1 |
| 1 Study Aims | 2 |
| 1.1 Biosensors:- General Principles. | 2 |
| 1.1.1 Electrochemical transducers | 4 |
| 1.1.2 Optical Transducers | 6 |
| 1.1.3 Mass Detecting transducers | 7 |
| 1.1.4 Calorimetric Transducers | 9 |
| 1.2 Immunosensors | 10 |
| 1.3 Principles of immunoassay | 10 |
| 1.3.1 Solid Phase Immunoassay | 10 |
| 1.3.2 Antibody-antigen binding. | 12 |
| 1.3.3 Labels in Immunoassay | 12 |
| 1.3.4 Enzyme Immunoassay | 13 |
| 1.4 Applications of Biosensors | 14 |
| 1.4.1 Biosensors in Clinical Chemistry | 14 |
| 1.5 Multianalyte Biosensors | 20 |
| 1.6 Protein Patterning on Sensor Surfaces | 21 |
| 1.7 Non Specific Binding | 22 |
| 1.8 Immobilisation | 24 |
| 1.8.1 Desirable properties of immobilisation method. | 24 |
| 1.9 Review of possible immobilisation strategies | 26 |
| 1.9.1 Physical deposition techniques | 26 |
| 1.9.2 Photolithographic "lift-off" techniques | 27 |

| 1.9.3 Entrapment in Polymers | 30 |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| 1.9.4 Electrodeposition followed by covalent cross-linking. | 31 |
| 1.9.5 Techniques using deep ultraviolet light | 31 |
| 1.9.6 Langmuir-Blodgett | 32 |
| 1.9.7 Self assembled monolayers | 32 |
| 1.9.8 Masking using thioalkanes | 34 |
| 1.9.9 Photoactivation and photodeactivation techniques | 36 |
| 1.9.10 Miscellaneous Techniques | 38 |
| Chapter 2 | 40 |
| 2 Introduction to Initial Experimental Work on Protein Patterning Techniques. | 41 |
| 2.1 The use of a Photoactivable Binding Agent for Immobilisation of Biological Molecules. | 41 |
| 2.1.1 Depositing a protein layer on the surface, adding photobiotin followed by the immobilisation of a protein through avidin-biotin interactions | 44 |
| 2.1.2 Depositing a protein layer on the surface, and add avidinated protein with photobiotin attached | 46 |
| 2.1.3 Depositing an avidin layer on the surface, adding photobiotin followed by the addition of the species to be immobilised. | 46 |
| 2.1.4 Investigation into resolution of avidin-photobiotin immobilisation procedure | 49 |
| 2.1.5 Investigation into timescale of avidin-photobiotin immobilisation procedure. | 49 |
| 2.1.6 Light activated patterning of multiple proteins on a silicon dioxide surface. | 53 |
| 2.1.7 Demonstration of light dependent coupling of enzymes and antibodies to gold, glass and carbon surfaces | 53 |
| 2.1.8 Investigation into reducing non-specific binding on gold surfaces. | 54 |
| 2.1.8.1 Immobilisation of avidin on gold using carbodiimide coupling to a N-acetyl-cysteine self assembled monolayer. | 54 |
| 2.1.8.2 Immobilisation of avidin on gold using DTSSP | 56 |
| 2.1.9 Regeneration of avidinated surfaces | 57 |
| 2.1.10 Characterisation of biomolecular patterns using atomic force microscopy and fluorescence microscopy. | 57 |
| 2. 2 Materials | 60 |
| 2.3 Methods for Patterning of Proteins using Avidin and Photobiotin. | 61 |
| 2.3.1 Patterning of an avidin-protein conjugate by depositing a protein layer on the surface, adding photobiotin followed by the addition of the avidinated protein. | 61 |

| 2.3.1.1 Patterning avidin by depositing a protein layer on the surface, adding photobiotin followed by the addition of avidin | 62 |
|--------------------------------------------------------------------------------------------------------------------------------------|----|
| 2.3.2 Patterning a protein by depositing a protein layer on the surface, and add avidinated protein with photobiotin attached | 63 |
| 2.3.2.1 Spectrophotometric determination of alkaline phosphatase activity. | 64 |
| 2.3.2.2. Assessment of cross reaction between photobiotin conjugates in solution. | 64 |
| 2.3.3 Patterning of a protein on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 65 |
| 2.4 Method for Investigation into Achievable Resolution of Protein Patterning using | 66 |
| Avidin-Photobiotin Technique. | |
| 2.5 Investigation into Timescale of Avidin-Photobiotin Immobilisation Procedure. | 66 |
| 2.5.1. Monitoring effect of exposure time on patterning process using fluorescence microscopy. | 66 |
| 2.5.2. Monitoring effect of exposure time on patterning process using colourimetric enzyme assay. | 66 |
| 2.6 Methods for Light Activated Patterning of Multiple Proteins onto Silicon Dioxide Surfaces | 67 |
| 2.6.1 Patterning of two immunoglobulins on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 67 |
| 2.6.2 Patterning of five functional antibodies on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 68 |
| 2.7 Demonstration of Light Dependent Coupling of Proteins to Gold Surface | 69 |
| 2.7.1 Demonstration of light dependent coupling of enzyme to gold surface | 69 |
| 2.7.2 Demonstration of light dependent coupling of antibody to gold surface | 70 |
| 2.8. Method for Light Dependent Coupling of a Protein to a Glass Surface | 71 |
| 2.9. Method for Light Dependent Coupling of a Protein to a Carbon Surface | 72 |
| 2.10 Methods for Investigation into Reducing Non-Specific Binding on Gold Surfaces. | 73 |
| 2.10.1 Investigation into reducing non-specific binding on gold surfaces. | 73 |
| 2.10.1.1 Immobilisation of avidin on gold by adsorption | 73 |
| 2.10.1.2 Immobilisation of avidin on gold using NAC and EDC | 73 |
| 2.10.1.3 Immobilisation of avidin on gold using DTSSP | 74 |
| 2.10.1.4 Treatment with blocking agent and testing for non specific binding. | 74 |
| 2.11 Methods to Investigate Regeneration of Avidinated Surfaces | 75 |
| 2.11.1 Methods to investigate regeneration of avidinated silicon dioxide surfaces | 75 |
| 2.11.2. Method to investigate regeneration of avidinated gold surfaces | 76 |

| 2.12 Method for Characterisation of Biomolecular Patterns using Atomic Force Microscopy | 77 |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------|
| and Fluorescence Microscopy. | |
| 2.13 Results and Discussion for Patterning of Proteins using Avidin and Photobiotin. | 79 |
| 2.13.1 Patterning method using an avidin-protein conjugate by depositing a protein layer on the surface, adding photobiotin followed by the addition of the avidinated protein | 79 |
| 2.13.2 Patterning a protein by depositing a protein layer on the surface, and add avidinated protein with photobiotin attached | 82 |
| 2.13.3 Patterning of a protein on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 85 |
| 2.14 Results and Discussion for Investigation into Achievable Resolution of Protein | 87 |
| Patterning using Avidin-Photobiotin Technique. | |
| 2.15 Results and Discussion for Investigation into Timescale of Avidin-Photobiotin | 9 0 |
| Immobilisation Procedure. | |
| 2.15.1 Monitoring effect of exposure time on patterning process using fluorescence microscopy. | 90 |
| 2.15.2 Monitoring effect of exposure time on patterning process using colourimetric enzyme assay. | 90 |
| 2.16 Results and Discussion for Light Activated Patterning of Multiple Proteins onto Silicon Dioxide surfaces | 92 |
| 2.16.1 Patterning of two immunoglobulins on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 92 |
| 2.16.2 Patterning of five antibodies on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct | 93 |
| 2.17 Results and discussion for Demonstration of Light Dependent Coupling of Proteins to a | 95 |
| Gold surface | |
| 2.17.1 Demonstration of light dependent coupling of an enzyme to a gold surface | 95 |
| 2.17.2 Demonstration of light dependent coupling of an antibody to a gold surface | 9 7 |
| 2.18 Results and Discussion for Light Dependent Coupling of a Protein to a Glass Surface | 98 |
| 2.19 Results and Discussion for Light Dependent Coupling of a Protein to a Carbon Surface | 99 |
| 2.20 Results and Discussion for Investigation into Reducing Non-Specific Binding on Gold | 100 |
| Surfaces. | |
| 2.21 Results and Discussion to Investigate Regeneration of Avidinated Surfaces | 102 |
| 2.21.1 Regeneration of avidinated silicon dioxide surfaces | 102 |
| 2.21.2 Regeneration of avidinated gold surfaces | 105 |
| 2.22 Results and Discussion for Characterisation of Biomolecular Patterns using Atomic Force Microscopy and Fluorescence Microscopy. | 106 |

| Chapter 3 | 110 |
|------------------------------------------------------------------------------------------|-----|
| 3 Simultaneous Determination of FSH and LH using a Multianalyte Immunosensor | 111 |
| 3.1 Follicle Stimulating Hormone and Luteinising Hormone | 111 |
| 3.1.1 Physiological function of FSH and LH. | 112 |
| 3.1.2 Diagnostic utility of FSH and LH | 115 |
| 3.2 Review of Assay Methods Available for the Assay of FSH and LH | 117 |
| 3.2.1 In vivo bioassays | 117 |
| 3.2.2 In vitro bioassays | 117 |
| 3.2.3 Haemagglutination assays | 118 |
| 3.2.4 Radioimmunoassay | 119 |
| 3.2.5 Immunoradiometric assay | 121 |
| 3.2.6 Dual Radioisotopic assay | 122 |
| 3.2.7 Non Isotopic immunoassays | 122 |
| 3.2.7.1 Enzymatic immunoassay | 122 |
| 3.2.7.2 Fluorescence Immunoassay | 125 |
| 3.2.7.3 Luminescence immunoassay | 126 |
| 3.2.8 Near-patient tests | 127 |
| 3.3 Amperometry | 128 |
| 3.3.1 Chronoamperometry | 130 |
| 3.4 Materials | 132 |
| 3.5 Method | 133 |
| 3.5.1 Photolithographic Lift Off patterning techniques for the fabrication of electrodes | 133 |
| 3.5.2 Electrode fabrication | 135 |
| 3.5.3 Antibody immobilisation | 137 |
| 3.5.4 Immunoassay procedure | 138 |
| 3.5.5 Behaviour of calibration curve over time | 140 |
| 3.6 Results | 141 |
| 3.6.1 Electrochemical Characterisation of Electrodes | 141 |
| 3.6.2 Response of immunosensors to aqueous and clinical samples | 143 |
| 3.6.3 Behaviour of calibration curve over time | 147 |
| 3.7 Discussion | 150 |

| Chapter 4 | 154 |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| 4 Introduction to microspot immunoassay | 155 |
| 4.1 Antibody antigen reactions at surfaces | 156 |
| 4.2 Microspot immunoassay | 157 |
| 4.2.1. Influence of working electrode area as proportion of total area on the specific/non specific signal ratio. | 163 |
| 4.3 Materials | 168 |
| 4.4. Methods | 169 |
| 4.4.1 Fabrication of electrodes and electrode arrays using photolithography | 169 |
| 4.4.2. Avidin-photobiotin immobilisation procedure. | 171 |
| 4.4.3. Amperometric response of electrodes. | 171 |
| 4.4.4 Investigation into effect of hemispherical diffusion field upon response of ultramicroelectrodes. | 171 |
| 4.4.4.1 Enzyme immobilisation procedure | 171 |
| 4.4.4.2 Amperometric analysis. | 173 |
| 4.4.5 Investigation into analytical "cross-talk" caused by mass transport of electrochemical species generated at one electrode to other electrodes in an array. | 173 |
| 4.4.5.1 Enzyme immobilisation procedure | 173 |
| 4.4.5.2 Amperometric analysis | 174 |
| 4.4.6 Construction of multianalyte enzymic biosensor using ultramicroelectrodes. | 174 |
| 4.4.6.1 Electrode fabrication. | 174 |
| 4.4.6.2 Enzyme immobilisation procedure. | 175 |
| 4.4.6.3 Amperometric analysis. | 175 |
| 4.4.7 Construction of multianalyte immunosensor using ultramicroelectrodes. | 175 |
| 4.4.7.1 Electrode Fabrication | 175 |
| 4.4.7.2 Antibody immobilisation | 175 |
| 4.4.7.3 Immunoassay procedure. | 176 |
| 4.5 Results and Discussion | 177 |
| 4.5.1 Investigation into effect of hemispherical diffusion field upon response of ultramicroelectrodes. | 177 |
| 4.5.2 Investigation into analytical "cross-talk" caused by mass transport of electrochemical | 180 |
| species generated at one electrode to other electrodes in an array. | |
| 4.5.3 Construction of multianalyte enzymic biosensor using ultramicroelectrodes. | 182 |
| 4.5.4 Construction of multianalyte immunosensor using ultramicroelectrodes. | 184 |

| Chapter 5 | 186 |
|--------------------------------------------------------------------------------------------------------------------|-----|
| 5 Further applications of avidin-photobiotin immobilisation procedure | 187 |
| 5.1 Patterning of nucleic acids | 187 |
| 5.1.1 DNA amplification | 190 |
| 5.1.2 Principle of assay for detection of Factor V_{leiden} genotype | 192 |
| 5.2 Materials | 195 |
| 5.3 Methods | 196 |
| 5.3.1 Patterning of nucleic acids using avidin and photobiotin. | 196 |
| 5.3.2 Utilising the avidin-photobiotin immobilisation technique in the determination of Factor V genotypes | 196 |
| 5.3.2.1 Immobilisation of oligonucleotide primers. | 196 |
| 5.3.2.2 PCR amplification of DNA | 197 |
| 5.3.2.3 Detection of PCR products | 197 |
| 5.4. Results and Discussion | 198 |
| 5.4.1 Patterning of nucleic acids using avidin and photobiotin. | 198 |
| 5.4.2 Utilising the avidin-photobiotin immobilisation technique in the determination of Factor V genotypes | 199 |
| 5.5 Review of related work from other groups since publication of the avidin-photobiotin immobilisation technique. | 200 |
| 5.6 Future possibilities | 202 |
| 5.6.1 Multianalyte Enzymic biosensors | 203 |
| 5.6.2 Multianalyte Immunosensors | 203 |
| 5.6.3 Integrated devices for analysis of nucleic acids. | 204 |
| 5.6.4 Direct sensing | 205 |
| 5.6.5. Cell Guidance | 205 |
| 5.6.6 Other applications of patterning technique | 207 |
| 5.6.7 Photoactivable Reagents other than Aryl Azides | 207 |
| 5.6.8 Very High Resolution Patterning | 208 |
| References | 209 |

Figures

| Figure | Title | Page |
|--------|--------------------------------------------------------------------------------------------------------------------------------------|------|
| 1.1 | Schematic representation of the ELISA "Sandwich" assay. | 15 |
| 1.2 | Amperometric measurement of glucose using GOD and a mediator. | 19 |
| 1.3 | Classification of techniques for immobilisation of biomolecules. | 25 |
| 1.4 | Molecular patterning using photoresist lift-off. | 29 |
| 1.5 | Constructing protein pattern by means of stamping SAM films. | 33 |
| 1.6 | Patterning of proteins by DUV irradiation of SAM films. | 35 |
| 1.7 | Schematic diagram of light-directed spatially addressable chemical synthesis. | 37 |
| 1.8 | Use of benzophenone as a photoactivable cross linker for protein immobilisation. | 39 |
| 2.1 | Structure of long arm photobiotin. | 43 |
| 2.2 | Scheme for depositing a protein layer on the surface adding photobiotin and immobilising proteins through avidin-biotin interactions | 45 |
| 2.3 | Scheme for depositing a protein layer on the surface and adding an avidinated protein with photobiotin attached | 47 |
| 2.4 | Scheme for immobilisation procedure coating the surface with avidin and photobiotin, | 48 |
| | followed by the addition of the protein to be patterned in the presence of light. | |
| 2.5 | Structure of azidobenzene. | 50 |
| 2.6 | Equilibrium between bonds in aryl azide. | 50 |
| 2.7 | Some of the reaction of aryl azides upon activation. | 52 |
| 2.8 | Reaction scheme for the promotion of peptide bond formation between amino and | 55 |
| | carboxylic acid residues in two proteins by use of carbodiimides. | |
| 2.9 | Reaction scheme for the formation of acyl ureas from carbodiimides attached to a carboxylic acid. | 55 |
| 2.10 | Structure of DTSSP (3,3'-dithiobis(sulfosuccinimidylpropionate)). | 56 |
| 2.11 | Coupling of protein to a DTSSP activated gold surface. | 56 |
| 2.12 | Diagram showing principle of operation of scanning tunnelling microscope | 59 |
| 2.13 | Diagram showing principle of operation of atomic force microscope | 59 |
| 2.14 | Photomicrograph of silicon dioxide patterned using a BSA / photobiotin / avidin / biotin-immunoglobulin construct | 80 |
| 2.15 | Plot of relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.14. | 80 |
| 2.16 | Photomicrograph of silicon dioxide patterned using a BSA / photobiotin / TRITC labelled avidin construct | 81 |

| 2.17 | Plot of relative fluorescent intensity against distance for cross sectional profile through a section of the pattern in Figure 2.16 | 81 |
|------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| 2.18 | Photomicrograph of silicon dioxide patterned using a BSA / photobiotin-avidin - immunoglobulin construct | 83 |
| 2.19 | Relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.18 | 83 |
| 2.20 | Normalised A_{280} from fractions from chromatographic separation of a photobiotin / avidin / protein solution, pre and post exposure to light. | 84 |
| 2.21 | Photomicrograph of silicon dioxide patterned with TRITC labelled IgG, using an avidin-photobiotin IgG construct | 86 |
| 2.22 | Plot of relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.21 | 86 |
| 2.23 | Photomicrograph of SiO ₂ patterned with TRITC labelled IgG, using an avidin-photobiotin IgG construct. | 88 |
| 2.24 | Plot of relative fluorescent intensity against distance for cross sectional profile of different width lines formed using the patterning of proteins as previously described | 89 |
| 2.25 | Plot of enzyme activity versus time of exposure of avidin - photobiotin coated SiO_2 to an ALP solution in the presence or absence of light. | 91 |
| 2.26 | Photomicrograph of Silicon dioxide patterned with two fluorescently labelled antibodies | 92 |
| 2.27 | Series of fluorescence photomicrographs demonstrating the sequential patterning of different antibodies on a silicon dioxide surface | 94 |
| 2.28 | Chronoamperometric responses to glucose of avidin - photobiotin coated gold electrodes which have been incubated with GOD in the presence or absence of light | 96 |
| 2.29 | Photomicrograph of glass patterned with a fluorescently labelled antibody | 98 |
| 2.30 | Fluorescent photomicrographs of attempts to regenerate avidinated SiO ₂ surfaces. | 103 |
| 2.31 | 2 - dimensional fluorescence micrograph of TRITC labelled IgG bound by a patterned complementary antibody. | 107 |
| 2.32 | Cross sectional profile through the pattern between points A and B marked in Figure 2.31. | 107 |
| 2.33 | 2 - dimensional AFM image of the same pattern as in Figure 2.31 | 108 |
| 2.34 | Cross sectional profile through the pattern between points A and B marked in Figure 2.33. | 108 |
| 2.35 | Three dimensional reconstruction from AFM data of an area of a protein patterned surface. | 109 |
| 3.1 | Schematic diagram illustrating control of FSH and LH release by GRH | 112 |
| 3.2 | Typical plasma FSH and LH concentrations during the menstrual cycle | 113 |

| 3.3 | Semi-logarithmic dose response curve obtained in radioimmunoassay | 119 |
|-------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| 3.4 | Semi-logarithmic dose response curve of an immunoradiometric assay | 121 |
| 3.5 | Typical semi-logarithmic dose response curve for a "sandwich" ELISA | 123 |
| 3.6 | Typical semi-logarithmic dose response curve for an EMIT assay | 124 |
| 3.7 | Scheme for the chemiluminescent oxidation of luminol. | 127 |
| 3.8 | Typical input and ouput chronoamaperometry waveforms | 131 |
| 3.9 | Fabrication of electrodes using photolithographic lift-off techniques | 134 |
| 3.10. | Diagrammatic representation of the fabricated electrode array for use in the construction of a multianalyte immunosensor. | 136 |
| 3.11 | Principle of the "sandwich" ELISA used in the determination of LH and FSH | 138 |
| 3.12 | Schematic representation of electrochemical detection of HRP using ferrocene derivatives. | 139 |
| 3.13 | Cyclic voltammograms of ferrocenemonocarboxylic acid obtained using fabricated electrode pair and Commercial (BAS) electrodes | 142 |
| 3.14 | Response of multianalyte sensor to FSH concentration in buffer | 144 |
| 3.15 | Response of multianalyte sensor to LH concentration in buffer | 144 |
| 3.16 | Comparison of sensor results for FSH in clinical samples (serum) with an established DELFIA technique. | 145 |
| 3.17 | Comparison of sensor results for LH in clinical samples (serum) with an established DELFIA technique | 146 |
| 3.18 | Response of multianalyte sensor to FSH concentration in buffer over a 60 day period. | 148 |
| 3.19 | Response of multianalyte sensor to LH concentration in buffer over a 60 day period. | 149 |
| 4.1 | Fractional antibody binding-site occupancy (F) related to antibody concentration for several different antigen concentrations. | 158 |
| 4.2 | Percentage of antigen bound to antibody related to antibody concentration for several different antigen concentrations. | 159 |
| 4.3 | Diagram demonstrating changes in field of view, antibody-coated area and coated antibody density. | 160 |
| 4.4 | Theoretical plot of specific/non-specific signal ratio versus electrode size for several systems with differing degrees of non-specific binding in a 2 electrode system | 164 |
| 4.5 | Transverse sections of a thin layer cell, microelectrode and a macroelectrode demonstrating differences in diffusion. | 165 |
| 4.6 | Diagrammatic representation of a photolithographically fabricated gold electrode with an electrode 2 mm in diameter and surrounded by a gold non-electrode surface. | 170 |

| 4.7 | Diagrammatic representation of a photolithographically fabricated gold electrode with an | 170 |
|-------|-----------------------------------------------------------------------------------------------------------------|-----|
| | electrode 10 μ m in diameter and surrounded by a gold non-electrode surface | |
| 4.8 | Diagrammatic representation of a photolithographically fabricated electrode array with | 170 |
| | electrodes either 1 mm or 10 μ m in diameter. | |
| 4.9 | Current as a function of electrode distance from the generator electrode for 1 mm | 181 |
| | diameter electrodes. | |
| 4.10. | Current as a function of electrode distance from the generator electrode for $10 \ \mu m$ diameter electrodes. | 181 |
| 4.11 | Response of the glucose oxidase electrode to solutions containing glucose and glutamate. | 183 |
| 4.12 | Response of the glutamate oxidase electrode to solutions containing glucose and glutamate. | 183 |
| 5.1 | Simplified version of the coagulation cascade | 189 |
| 5.2 | Biochemical pathways of the conversion of homocysteine to methione and cysteine. | 189 |
| 5.3 | Principle of the polymerase chain reaction method of amplifying DNA | 191 |
| 5.4 | Principle of the assay for identification of Factor V genotype | 194 |
| 5.5 | Photomicrograph the patterning of nucleic acids by use of the avidin-photobiotin | 198 |
| | technique. | |
| 5.6 | Surface plasmon resonance microscopy image of IgG patterned on to a surface using the avidin-photobiotin method | 200 |

Tables

| Table | Title | Page |
|-------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------|
| 1.1 | Physicochemical changes and possible detection methods in biosensors. | 4 |
| 1.2 | Applications of immunosensors for medical and environmental monitoring. | 12 |
| 1.3 | Commonly requested clinical chemistry investigations from different sources. | 19 |
| 2.1 | Enzyme activity versus time of exposure of avidin - photobiotin coated SiO_2 to an ALP solution in the presence or absence of light | 91 |
| 2.2 | Currents obtained from avidin - photobiotin coated gold electrodes incubated with GOD in the presence or absence of light. | 95 |
| 2.3 | Currents obtained from avidin - photobiotin coated glassy carbon electrodes which have been incubated with GOD in the presence or absence of light | 99 |
| 2.4 | Non specific binding to treated gold electrodes after exposure to IgG - ALP conjugate. | 101 |
| 2.5 | Summary of results from procedures to attempt regeneration of avidinated surfaces | 104 |
| 2.6 | Currents obtained from gold electrodes pre and post repatterning using guanidine. | 105 |
| 4.1 | Examples of specific / non-specific signal ratios for differently antibody deposition and field of view areas. | 162 |
| 4.2 | Examples of specific / non-specific signal ratios for different antibody deposition and antibody surface densities. | 162 |
| 4.3 | Areas of sensor surfaces exposed to light in presence of GOD during patterning process and those areas exposed to DUV radiation following the patterning process | 172 |
| 4.4 | Areas of sensor surfaces containing specifically and non-specifically bound GOD | 173 |
| 4.5 | Currents obtained from 2 mm diameter electrodes in the presence of glucose when no potential was applied to the surrounding gold surface. | 179 |
| 4.6 | Currents obtained from 10 μ m diameter electrodes in the presence of glucose when no potential was applied to the surrounding gold surface | 179 |
| 4.7 | Currents obtained from the 2 mm diameter electrodes in the presence of glucose when a potential of +650 mV was applied to the surrounding gold surface as well as to the working electrode. | 179 |
| 4.8 | Currents obtained from the 10 μ m diameter electrodes in the presence of glucose when a potential of +650 mV was applied to the surrounding gold surface as well as to the working electrode. | 180 |
| 4.9 | Currents obtained in presence of antigens for ultramicroelectrode immunoassay when antibody was immobilised only on electrode. | 184 |
| 4.10. | Currents obtained in presence of antigens for ultramicroelectrode immunoassay when antibody was immobilised on surrounding area as well as on electrode | 185 |
| 5.1 | Results obtained from a DNA sensor for patients with different Factor V genotypes | 199 |

ABBREVIATIONS

| Г | Density of antigen binding sites on surface |
|---------------|-----------------------------------------------------------------------|
| Θ | Fractional binding of antigen |
| Θ_{eq} | Fractional binding of antigen at equilibrium |
| A | Electrode area |
| Ae | Area of specific electrode |
| A_o | Area of other electodes |
| A_{i} | Total area |
| A & E | Accident and emergency (Casualty) |
| Ab | Antibody |
| [<i>Ab</i>] | Antibody concentration |
| AbAg | Antibody-antigen complex |
| [AbAg] | Antibody-antigen complex concentration |
| a.c. | Alternating current |
| Ag | Antigen |
| Ag* | Labelled antigen |
| [Ag] | Antigen concentration |
| $[Ag]_0$ | Initial antigen concentration |
| $[Ag]_{eq}$ | Antigen concentration at equilibrium |
| ALP | Alkaline phosphatase |
| APC | Activated protein C |
| AST | Aspartate transaminase |
| BSA | Bovine serum albumin |
| С | Concentration (mol cm ⁻³) |
| CCU | Coronary care unit |
| СК | Creatine kinase |
| D | Diffusion coefficient (cm ² s ⁻¹) |
| Des | Density of signal producing molecules specifically bound to electrode |

| D_{en} | Density of signal producing molecules non-specifically bound to electrode | | | |
|---------------------------------------------------|---------------------------------------------------------------------------|--|--|--|
| D_{on} | Density of signal producing molecules bound to other electrodes | | | |
| D_{nn} | Density of signal producing molecules bound to non-electrode areas | | | |
| DELFIA | Dissociation enhanced lanthanide fluoroimmunoassay | | | |
| DMF | Dimethylformamide | | | |
| DNA | Deoxyribonucleic acid | | | |
| dNTP | dinucleotide phosphate | | | |
| DTSSP 3,3'-dithiobis(sulfosuccinimidylpropionate) | | | | |
| DUV | Deep ultraviolet | | | |
| EDC | 1-ethyl-3-(3-dimethylaminopropyl)carbodiimide | | | |
| EIA | Enzyme immunoassay | | | |
| ELISA | Enzyme-linked immunosorbent assay | | | |
| EMIT | Enzyme-multiplied immunoassay | | | |
| F | Fractional occupancy | | | |
| F | 96,500C Faraday ⁻¹ | | | |
| Fa | Apparent concentration of functional group of ligand | | | |
| FITC | Fluorescein isothiocyanate | | | |
| FSH | Follicle stimulating hormone | | | |
| GLOD | Glutamate oxidase | | | |
| GOD | Glucose oxidase | | | |
| GP | General Practitioner | | | |
| GRH | Gonadotrophin release hormone | | | |
| hCG | Human chorionic gonadotrophin | | | |
| HRP | Horseradish peroxidase | | | |
| i | Current | | | |
| ICSH | Interstitial cell stimulating hormone (LH) | | | |
| Ig | Immunoglobulin | | | |
| IgG | Immunoglobulin G | | | |
| ISE | Ion selective electrode | | | |

| ISFET | Ion-sensitive field effect transistor | | |
|----------------|-------------------------------------------------------------------|--|--|
| ITU | Intensive therapy unit (Intensive care) | | |
| J | Joules | | |
| J | Mass transport flux (mol cm ⁻² s ⁻¹) | | |
| k_a | Adsorption rate constant | | |
| k_d | Desorption rate constant | | |
| К | Antibody affinity constant | | |
| K _d | Dissociation constant for antigen binding | | |
| LB | Langmuir-Blodgett | | |
| LDH | Lactate dehydrogenase | | |
| LFT | Liver function tests | | |
| LH | Luteinising hormone | | |
| ln | Logarithm to base e | | |
| MGIA | Microgravimetric immunoassay | | |
| MTHFR | Methylenetetrahydrofolate reductase | | |
| NAC | N-acetyl-cysteine | | |
| NFOSM | Near-field optical scanning microscope | | |
| NHS | N-hydroxy succinimide ester | | |
| NPT | Near-patient testing | | |
| NSB | Non-specific binding | | |
| Nvoc | Nitroveratryloxycarbonyl | | |
| PBS | Phosphate buffered saline pH 7.4 | | |
| PCR | Polymerase chain reaction | | |
| pI | Isoelectric point | | |
| pNPP | para nitrophenol phosphate | | |
| QCM | Quartz crystal microbalance | | |
| R | Molar gas constant (8.31441 J mol ⁻¹ K ⁻¹) | | |
| RIA | Radioimmunoassay | | |
| Snsb | Signal due to non-specific components | | |

| S_{sb} | Signal due to specific components | | |
|----------------|-------------------------------------|--|--|
| SAM | Self assembled monolayer | | |
| SAW | Surface acoustic wave | | |
| SCBU | Special care baby unit | | |
| SEM | Scanning electron microscopy | | |
| SPR | Surface plasmon resonance | | |
| STM | Scanning tunnelling microscopy | | |
| t | Time | | |
| t _% | Half life | | |
| TBS | Tris buffered saline | | |
| TDM | Therapeutic drug monitoring | | |
| TFT | Thyroid Function tests | | |
| Tris | Tris(hydroxymethyl)methylamine | | |
| TRITC | Tetramethylrhodamine isothiocyanate | | |
| TSH | Thyroid stimulating hormone | | |
| v_a | Rate of antigen adsorption | | |
| v_d | Rate of antigen desorption | | |
| x | Position | | |

CHAPTER 1

The Road goes ever on and on Down from the door where it began. Now far ahead the road has gone, And I must follow if I can, Pursuing it with eager feet, Until it joins some larger way where many paths and errands meet. And whither then? I can not say.

> J. R. R. Tolkien Lord of the Rings

1 Study Aims

The primary aim of this work was the construction of a multianalyte amperometric immunosensor, capable of quantitatively measuring two or more high molecular weight analytes simultaneously. An integrated electrochemical multianalyte immunosensor had not previously been reported in the literature, although electrochemical multianalyte immunosensing systems have been reported where more than one sensor or parts of sensors have been fabricated separately and later integrated to make an array.^{1,2} In general, in the miniaturisation of sensors, it is preferable that the sensor should be constructed in such a manner that facilitates fabrication, namely from a single piece of material and employing techniques suitable for mass production.

Miniaturisation is desirable for a number of reasons including the ability to perform the analyses with a very small sample, faster response times, high throughput and reduced cost as well as the potential for implanting sensors into biological tissues.^{3,4} In this study two glycoproteins, follicle stimulating hormone (FSH) and luteinising hormone (LH) were chosen as the model analytes, the measurement of FSH and LH being selected as, in clinical analysis, the determination of these hormones is usually requested simultaneously. Additionally, FSH and LH are normally assayed by heterogeneous immunoassay, the approach to be used in this work.

<u>1.1 Biosensors:- General Principles.</u>

Biosensors are analytical devices that use biological material to detect chemical or biological species directly, without the need for complex sample processing. Biosensors consist of a biological 'receptor' element that has the ability to bind to a very narrow range of analytes (conferring specificity) and a 'transducing' element that provides a quantitative

indication that the analyte and the biological receptor have combined. Measurement of the target analyte is achieved by selective transduction of an aspect of the biomolecule-analyte reaction into a signal. Generally, biosensors are distinguished from other biologically based analytical devices, in that the biological element is immobilised so that it is in intimate contact with, or incorporated within the transducer element. As analytical devices, they may also possess several other attractive features such as compact size, low cost, fast response time, specificity of response, simplicity in use and absence of radioactivity.^{3,5}

A variety of materials have been used as the biological receptor element including enzymes,^{6,7} antibodies,⁸ antigens,⁹ organelles,¹⁰ cell membrane receptors,¹¹ complete cells¹² and slices of tissue.^{13,14} Of these biological materials, enzymes and antibodies are the most frequently employed, which is a reflection of their widespread use in conventional analytical techniques. One great advantage of the use of biological molecules is that they possess discriminatory powers equal to or better than the most powerful of the non-biological analytical techniques.¹⁵ For example, both enzymes and antibodies are able to distinguish between such closely related molecules as optical and geometrical isomers. An additional advantage of the use of biological molecules in analyses, is that it may be possible to incorporate a functional amplification system, which can allow improvement of both detection limits and sensitivity by several orders of magnitude.¹⁰

The transducer element of a biosensor detects a change in physiochemical properties upon binding of the analyte to its biological receptor. This interaction may result in a change of pH, electron transfer, refractive index, heat transfer, or uptake or release of specific ions or gases. Transducer mechanisms include electrochemical,^{6,17} optoelectronic,¹⁸ piezoelectric¹⁹ and thermometric^{20,21} devices. Changes in physiochemical properties and possible detection methods used in biosensors are summarised in Table 1.1

| Physicochemical Change | Possible Detection Methods | |
|---------------------------|-----------------------------------------------------------------------------------------------------------------|--|
| Heat | Calorimetric | |
| Optical | Absorbance, luminescence, fluorescence, reflective, ellipsometric, Surface Plasmon Resonance, optoelectronic | |
| Electrons | Amperometric | |
| Ions | Potentiometric, conductimetric | |
| Gases | Amperometric, potentiometric, piezoelectric, conductimetric | |
| Mass | Piezoelectric | |

Table 1.1. Physicochemical changes and possible detection methods in biosensors.

1.1.1 Electrochemical transducers

Electrochemical biosensors combine the analytical power of electrochemical techniques with the selectivity and specificity of biological molecules. Clark and Lyons²² first introduced the concept of the "soluble enzyme" electrode, but the earliest incorporation of an enzyme into an electrode was by Updike and Hicks,⁶ who immobilised glucose oxidase in a gel over an oxygen electrode to measure the glucose concentration of biological fluids. The majority of electrochemical biosensors can be classified into one of two groups; amperometric or potententiometric devices. In amperometric devices⁵ an electrical potential is applied to an electrode system, which facilitates reduction or oxidation of electroactive species at the electrode surface. This "redox" reaction involves the transfer of electrons between the electrode and an electroactive species, with a current being produced which is proportional to the concentration of the electrochemical species at the electrode surface. Amperometric transducers have been very widely used in biosensing, particularly systems based upon the detection of oxygen or hydrogen peroxide. The amperometric approach has high sensitivity and a linear concentration dependence (compared with a logarithmic relationship in potentiometric systems), and improved selectivity can be obtained by careful selection of the electrode potential. Many molecules (such as proteins) are not intrinsically electroactive, therefore it is frequently necessary to introduce enzymic labels that can catalyse redox reactions and produce electroactive species. For example, amperometric sensing of immunological reactions can be performed by using an immobilised antibody to bind the analyte, and then adding a second enzyme labelled antibody which is also directed against the analyte.²³⁻³⁰

The first potentiometric enzyme electrode was reported by Guilbault and Montalvo⁷ for the determination of urea. In this technique, the fact that the Nernst potential across a membrane was dependent upon the concentration of electroactive species on either side of the membrane was exploited; thus if the concentration on one side was kept constant then the potential generated across the membrane (measured against a reference electrode maintained at zero current flow) was proportional to the concentration on the other side (the concentration - potential relationship being logarithmic). The use of ion-selective membranes introduced specificity into the system. Examples of potentiometric sensors are ion-selective electrodes (ISEs) including the pH electrode³¹ and ion-sensitive field effect transistors (ISFETs).³² Potentiometric electrodes have been used as the transducer in biosensors employing immunological reactions, employing a number of approaches,³³⁻³⁶ one of which, is to measure the change in potential caused by the alteration in the ionophoric properties of an immobilised antigen (either ionophoric in its own right, or conjugated to an ionophore) when selectively bound by antibodies.^{37,38} Due to their small size, their generic

technology and ease of manufacture ISFETs are very attractive in biosensor technology, one example of their usage is in detecting changes in charge densities and isoelectric points that take place upon the formation of an antibody-antigen complex.³⁹

A third electrochemical transduction system that has been used in biosensors is the measurement of conductance. An example is the measurement of urea using urease which results in the production of NH₄⁺ causing an increase in the conductance of the solution.¹⁷ Conductimetric biosensors utilising immunological reactions for the measurement of pesticides⁴⁰ and drugs of abuse⁴¹ have also been developed. The use of conductance has not been widely used in biosensors, partially because conductimetric techniques tends to suffer from poor specificity of measurement as the resistance of a solution is dependent upon all of the ions present, and there are variations in ionic strength and buffering capacity between samples.

1.1.2 Optical Transducers

The measurement of absorbance or emission of electromagnetic radiation, by either reactants or products of a biological system has been widely used in biosensors. Optical transducers are able to respond to changes in absorbance, fluorescence, luminescence, reflectance, resonance, interference or refractive index. Early optical sensing techniques generally required the use of labels, and these indirect optical sensing techniques are still very widely used in biosensor technology. In some of the earliest optical systems, reactants were immobilised in columns or on tubing and optical readings (such as absorbances) were taken downstream in a flow through system.⁴² Fibre-optic technology enables the fabrication of "optrodes", where the reagent phase is immobilised on a single optical fibre, so that changes in optical properties of the reagent phase attributable to the analyte can be

measured. Antibodies can be labelled with a number of materials in order to allow optical measurement. Enzymes can be used as a label to eliminate substrates or generate products that absorb light,⁴³⁻⁴⁷ fluoresce⁴⁸ or luminesce,⁴⁹⁻⁵¹ and these have been widely used in biosensor technology. Antibodies can also be directly labelled with fluorescent compounds⁵² or coloured material such as red blood cells⁵³ or colloidal gold.⁵⁴

Optical techniques for the direct monitoring of changes in adsorption, fluorescence, refractive index or light scattering have been used in biosensors. These direct optical sensing techniques (those that do not require a label) include attenuated total internal reflection,⁵⁵ ellipsometry,⁵⁶ surface plasmon resonance⁵⁷ and waveguides.⁵⁸ In most of these systems, light entering the device is directed towards the sensing surface and then reflected back, with the light emerging from the device revealing information regarding the events occurring at the sensor surface. These direct sensing techniques are particularly useful for real time applications such as investigating reaction kinetics, although they tend to be affected to a greater extent by non-specific interference than do indirect sensing techniques. Attenuated total internal reflection,⁵⁹ ellipsometry,⁶⁰ surface plasmon resonance⁶¹ and waveguides⁶² have all been used to monitor immunochemical reactions.

1.1.3 Mass Detecting transducers

In 1880 Jacques and Pierre Curie discovered that a mechanical stress applied to the surface of quartz (and a number of other crystals) resulted in an electrical potential across the crystal; the piezoelectric effect, the magnitude of which was proportional to the mechanical stress applied.⁶³ The piezoelectric effect only exists in materials that crystallise in noncentrosymmetric space groups. The quartz crystal microbalance (QCM) makes use of this phenomenon, and comprises a thin quartz crystal sandwiched between two electrodes. An alternating current (a.c.) field is established across the crystal causing vibrational motion

of the crystal at its resonant frequency. One of the parameters that the resonant frequency is sensitive to, is a change in the mass of absorbate on the crystal. In piezoelectric biosensors, crystals are coated with an adsorbent that selectively interacts with the analyte of interest, subsequent binding increases the mass of the coated crystal and alters its frequency of oscillation. Monitoring the change in oscillation frequency enables determination of the change in mass, which is proportional to analyte concentration. However, in thick films of adsorbed layer (as is generally the case in biosensing), the change in resonant frequency is not only caused by the addition of mass, but by change in the viscoelastic response of the sensor system.⁶⁴ Thus, crystals employing thick elastic layers which "couple" into the supporting fluid medium (e.g. multilayer sensors containing organic molecules), which makes interpretation of data difficult.

Notwithstanding this, the QCM has excellent sensitivity, being capable of measuring mass changes corresponding to submonolayer adsorption and desorption. Consequently it has been employed in the analysis of gaseous environmental pollutants,⁶⁵⁻⁶⁷ herbicides,^{68,69} medical diagnostic assays^{70,71} and in monitoring antibody-antigen interactions.¹⁹

Surface Acoustic Wave (SAW) devices have also been used as transducers in biosensors. In these devices, an acoustic wave is generated by the application of an alternating voltage across an interdigital transducer (a pattern of interlaced metal electrodes). The acoustic signal produced is detected by a second interdigital transducer located a few millimetres away. The adsorption of analyte to the crystals slows the acoustic wave, and the change in velocity (which is proportional to the analyte concentration) is recorded. Again SAW devices have seen application in the analysis of gaseous environmental pollutants,⁷² and in monitoring antibody-antigen interactions.⁷³

Surface plasmon resonance (SPR) is a quantum optical-electrical phenomenon arriving from the interaction of light with a suitable metal or semiconductor surface, where under certain conditions the photon's energy is transferred to packets of electrons called plasmons.⁵⁷ In SPR the angle of incident monochromatic light is varied. At most angles the light is reflected but at a particular angle the plasmons are excited and the incident light is almost completely absorbed. The angle of incidence that causes resonance is dependent upon the refractive index of the metal, the nature of the metal's surface (i.e. roughness) and the refractive index of the medium in contact with the metal surface, therefore changes in mass at the surface result in a change to the resonance characteristics.

A disadvantage of direct immunosensing techniques (e.g. QCM, SAW and SPR) is that any protein that is adsorbed will interfere with the assay, whilst for indirect immunosensing techniques, only proteins that contribute to the signal as generated by the label will interfere with the assay (unless the non-specific binding interferes with the molecular recognition between immobilised molecule and analyte).

1.1.4 Calorimetric Transducers

Calorimetric biosensors are constructed by attaching the biological component to a thermistor, which detects the heat generated by enzymatic reactions.^{74,75} These devices have been used for enzymatic measurements⁷⁶ and have also been used for immunological analysis by labelling antibodies with enzymes.^{77,78}

1.2 Immunosensors

The primary aim of this work was the construction of a multianalyte amperometric immunosensor. Immunosensors are biosensors where the biological receptor component immobilised at the transducer is an antigen or an antibody. Immunosensors set out to perform an immunoassay without the requirement for the multiple, time-consuming, procedures that are usually associated with this technique. Immunoassay may be defined as a technique based on the reaction between an antigen and an antibody for measuring the concentration of either reactant in solution. In solid phase immunoassay, the antibody or the antigen is immobilised on a solid insoluble matrix.⁷⁹ An indirect immunosensor uses a separate labelled species that is detected after binding, e.g. enzyme linked immunosorbent assay (i.e. a heterogeneous immunoassay), whilst direct immunosensors can monitor binding in real time, by measuring parameters such as changes in mass or optical properties such as refractive index. Immunosensors have been used for a number of applications in medical, environmental and biotechnology measurements, some of which are summarised in Table 1.2.

1.3 Principles of immunoassay

1.3.1 Solid Phase Immunoassay

Antigens and antibodies can be coupled to an insoluble matrix with the retention of immunological activity,⁷⁹ which facilitates the separation and washing steps required to differentiate bound and free fractions of the label in heterogeneous assays. The chemical composition and the form of the solid phase of the immunoadsorbent can be varied in many different ways, solid phases that have been used include fine powder particles,⁸² thin discs,⁸³ magnetic particles,^{84,85} microtitre wells⁸⁶ and coated tubes.⁸⁷

| Transduction technique | Medical Applications | Environmental applications |
|------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|
| Amperometric | hCG, ^{23,29} LDH5, ³⁰ theophylline, ²⁴ von Willebrand Factor, ²⁵ apolipoprotein E, ²⁶ α1-acid glycoprotein ²⁷ | Herbicides ²⁸ |
| Potentiometric | Penicillin, ³³ Insulin, ³⁴ hCG, ³⁵ HBsAg ³⁶ | |
| Conductometric | Methamphetamine ⁴¹ | Pesticides ⁴⁰ |
| Absorbance | hCG, ⁴⁴ C-reactive protein, ⁴⁵ theophylline, ⁴⁶ insulin ⁴⁷ | |
| Fluorescence | β2 microglobulin ⁴⁸ | |
| Luminescence | Oestradiol, ^{49,50} α-interferon, ^{49,50} hCG, ⁸¹ IgG ^{49,50} | Mercury ⁵¹ |
| Ellipsometric | γ interferon ⁶⁰ | |
| SPR | hCG ⁶¹ | |
| Waveguides | Microbial toxin ⁶² | |
| Interferometric | HBs Ag ⁸⁰ | |
| Piezoelectric | Drugs of abuse, ⁷⁰ HIV antibodies ⁷¹ | Gaseous pollutants ^{65, 66} herbicides ^{66,69} |
| SAW | IgG ⁷³ | Gaseous pollutants ⁷² |
| Calorimetric | Proinsulin, ⁷⁷ albumin, ⁷⁸ gentamicin ⁷⁸ | |

Table 1.2 Examples of applications of immunosensors for medical and environmental monitoring.

1.3.2 Antibody-antigen binding.

Antibody-antigen binding is due to the summation of a number of relatively weak non covalent forces including Coulombic interactions, van der Waals forces and hydrophobic and hydrogen bonding.⁸⁸ The binding of an antigen (Ag) to the antibody (Ab) is reversible and the strength of the interaction is dependent upon the affinity of the antibody and the number of binding sites on the antigen. Antibody-antigen binding can be expressed as an equilibrium as shown in Equation 1.1

$$Ab + Ag \Leftrightarrow AbAg (1.1)$$

The equilibrium point is dependent upon the concentration of the antibody and antigen ([Ab] and [Ag]) respectively and upon the strength of their interaction. This strength of interaction can be expressed as an affinity (or association) constant, K_a (Equation 1.2). Values for this constant range between 50 and 10⁹ M⁻¹.⁸⁹

$$K_{a} = \frac{[AgAb]}{[Ab][Ag]}$$
 (1.2)

The affinity is independent of the number of binding sites on the antigen. Avidity is the term used to express the total binding strength of the antibody to all of the binding sites on the antigen. A typical IgG molecule binds at least four orders of magnitude more strongly to a multivalent antigen if both binding sites are engaged, than if only a single site is bound.⁸⁹ Antibody-antigen reactions at surfaces are discussed more fully in Chapter 4.

1.3.3 Labels in Immunoassay

Early immunoassay techniques used radioactive labels to quantify the degree of binding. The first of these techniques to incorporate radiolabels was radioimmunoassay (RIA) which was developed and applied to measurement of substances in humans by Berson and Yalow.^{90,91} In recent years there has been a move away from the use of radioactive labels, and they have largely been replaced by amongst others luminescent,⁹² fluorescent,⁹³ and enzymatic labels.^{94,95} These non-isotopic labels possess a number of advantages such as longer shelf life, shorter assay time, easier automation and improved detection limits. They also introduce the possibility of developing disposable devices that can be used to perform near-patient testing.^{96,97} The main objective of this study was to establish a multianalyte immunosensor based upon the electrochemical detection of the formation of products catalysed by enzymic labels. Immunoassays utilising enzyme labels are generally classified into one of two classes, heterogeneous and homogeneous assays.

1.3.4 Enzyme Immunoassay

In heterogeneous enzyme immunoassay, after the antibody and antigen have been incubated together, a step is required to separate the free antigen and antibody from the antibody-antigen complexes. The enzymic activity of one or both of these fractions is then assessed. The most common heterogeneous enzyme immunoassay is Enzyme-Linked Immuno-Sorbent Assay (ELISA), in this technique either the antigen or the antibody can be labelled with the enzyme. In the so called 'Sandwich' ELISA assay (Figure 1.1), an excess of immobilised antibody is incubated with the sample (Figure 1.1a), and the antigen binds with the immobilised antibody. After washing away the unbound sample constituents (Figure 1.1b), the immobilised antibody-antigen complex is incubated with excess enzyme-labelled antibody. This enzyme labelled antibody binds to a second antigenic site (Figure 1.1c). Excess enzyme-labelled antibody is washed away. Therefore, the remaining enzyme activity is directly proportional to the antigen concentration (Figure 1.1d).

In homogeneous enzyme immunoassays (enzyme multiplied immunoassay technique (EMIT)), a separation process is not required. In this technique antigen-coupled enzyme activity changes when bound to an antibody. Upon the addition of a sample, less

antigen-enzyme conjugate will bind to the antibody, as it is in competition for the limited numbed of antibody binding sites with free antigen from the sample. Therefore, when antigen from the sample is added to the reaction mixture, the change in enzymic activity is inhibited, and the degree of change is inversely proportional to concentration of the analyte in the sample. ELISA is generally applicable to the measurement of almost any antigen, and can detect substances at much lower concentrations than can EMIT. EMIT is mainly used for the analysis for substances of relatively low molecular weight, including many drugs.

1.4 Applications of Biosensors

Biosensors have applications in a wide variety of fields including defence,⁹⁸ safety,⁹⁹ agricultural and industrial analysis,¹⁰⁰ process monitoring,¹⁰¹ environmental monitoring,^{102,103} food monitoring,¹⁰⁴ drug screening¹⁰⁵ and medical diagnosis and monitoring.¹⁰⁶ Commercially, biosensors have not as yet achieved as significant percentage of sales in the diagnostics market as in some predictions, but there is the potential of hugely increased sales.¹⁰⁷ Specific examples of the applications of immunosensors are shown in table 1.2.

1.4.1 Biosensors in Clinical Chemistry

The use of biosensors in medical diagnosis and assessment introduces an increased capacity for decentralised testing, i.e. performing the analyses outside of the central laboratory. Dependent upon the analyte this can be important in special care units such as intensive therapy (ITU), cardiac care (CCU) and special care baby units (SCBU), on hospital wards, outpatient clinics, General Practitioner's surgeries or even for patient self-monitoring at home. The main criterion for determining if decentralised testing is effective, is whether obtaining the result immediately will aid in the treatment of the patient, other factors are whether a reduction in cost or inconvenience will result. Sensors have a greater potential for



Figure 1.1. Schematic representation of the ELISA 'Sandwich' assay. An excess of immobilised antibody is incubated with the sample (a), and binds the antigen therein. After washing away the unbound sample constituents (b), the immobilised antibody-antigen complex is incubated with excess enzyme-labelled antibody. This enzyme labelled antibody binds to a second antigenic site (c). Excess enzyme-labelled antibody is subsequently washed off, and the remaining enzyme activity is directly proportional to the antigen concentration (d).
the de-skilling of analysis than of any previous technology.¹⁰⁸ Table 1.3 shows frequently requested tests from various sources. Whilst this list is clearly not definitive, it illustrates potential applications of biosensors for near patient testing.

It can be seen from Table 1.3, that only a small proportion of the frequently performed assays on units where the patient is critically ill (i.e. ITU and SCBU) are amenable to assay using biosensors. By contrast, biosensors can play a much more important role in accident and emergency (A&E) departments, where their ease of use and rapid response time can be invaluable in diagnosis, particularly in the case of poisonings. However, the speed of obtaining a result is not only important in life or death situations. In tests performed on outpatients (in clinics, GP's surgery and patient self-testing), a rapidly obtained result may assist in the treatment of the disorder. Examples of this are self-monitoring of blood glucose by diabetics, who can then adjust insulin dosage and/or diet accordingly, and therapeutic drug monitoring where circulating levels of the drug are determined, such that if necessary the dosage of the drug can be changed so as to bring the circulating levels within the effective range. In addition, obtaining a result rapidly, can obviate the need for extra visits, saving time and money for both the patient and the health service.

Near-patient testing (NPT) is not without problems, as errors can occur in the operation of even the most user-friendly tests and unskilled or semi-skilled staff are often unaware of these and thus they may go unnoticed. The risk of this occurring can be reduced by skilled laboratory staff providing proper training and support and the introduction of a Quality Control System.¹⁰⁹

| <u>ITU</u> | <u>SCBU</u> | <u>CCU</u> | <u>A&E</u> | Gener | ral Wards |
|------------------------------------------------|-------------------------------------------|------------------------------|-----------------------------------------------------------------------------------------|-----------------------------------------------|-----------------------------------------------|
| pH pO2 pCO2 Na+ K+ Cl- HCO3- | as ITU plus Bilirubin Intralipid | K+ LDH * CK * AST * | as ITU plus Paracetamol * Salicylate * Ethanol * Amylase* Uric acid * | Na+ K+ Urea Creat Gluco Uric a | * inine * ose * acid * |
| Urea * | | | Drugs of abuse * | | |
| Glucose * | | | | | |
| Total Protein | | | | | |
| Albumin | | | | | |
| Calcium | | | | | |
| Out-patient clinics | GP.s surgery | | Patient Self monitoring Screening Tests | | Screening Tests |
| Glucose * | as Outpatient clinics | | Glucose * | | alpha-foeto protein * |
| Urea * | plus | | TDM * | | Phenylalanine * |
| Creatinine * | hCG * | | | | TSH * |
| Cholesterol * | LDH * | | | | |
| Triglyceride * | CK * | | | LFT = | = Liver function tests (mainly enzyme assays) |
| LFT * | AST * | | | TFT = | = Thyroid function tests (hormone assays) |
| TFT * | | | | Infert | ility Tests are mainly hormone assays |
| Infertility Tests * | | | | TDM | = Therapeutic drug monitoring |
| Uric acid * | | | | | |
| TDM * | | | | | |

Table 1.3. Some of the most frequently requested clinical chemistry investigations from different sources. Assay marked * are those frequently performed utilising functionality of biological molecule.

At present most biosensors employ the biological functionality of enzymes or antibodies to impart specificity to the analytical system. The greatest market in biosensors for medical applications is in devices for the self-monitoring of glucose by sufferers of diabetes mellitus. Self-monitoring allows diabetics to better control glucose levels by adjusting their diet and/or insulin dosage in response to the results. It has been demonstrated that at least in some diabetic groups improved control of glucose concentration results in a better prognosis, with a reduced risk of the disease's complications such as peripheral neuropathy and retinopathy.¹¹⁰

The measurement of glucose provides a good example of the varying degrees of complexity that are possible in biosensor technology. These biosensors are based upon the concentration of glucose being the rate limiting parameter in an enzyme catalysed reaction, the enzyme employed usually being glucose oxidase (GOD), which can be obtained as large amounts of a pure preparation relatively cheaply. The simplest of these sensors are the so called "stick tests." In these glucose acts as the substrate for GOD resulting in the formation of hydrogen peroxide, which oxidises a chromaphore resulting in a colour change. The resultant colour intensity is proportional to the glucose concentration which can be determined by comparison of the colour with a chart or by the use of a simple reflectance spectrophotometer.^{111,112}

A successful commercial product for the determination of glucose is the Exactech device marketed by Abbott (Chicago, IL, USA). In this device GOD oxidises glucose to gluconolactone and in the process is itself reduced. GOD is reoxidised by means of a mediator, which in the process of being reoxidised at an electrode surface results in the production of a current, proportional to the glucose concentration. This process is illustrated in Figure 1.2.



Figure 1.2. Scheme for amperometric measurement of glucose using glucose oxidase and a mediator. Glucose oxidase (GOD) oxidises glucose to gluconolactone and in the process is itself reduced. GOD is reoxidised by means of a mediator, which in the process of being reoxidised at an electrode surface results in the production of a current, proportional to the glucose concentration.

A glucose biosensor can be incorporated into more complex instruments which are capable of measuring several analytes simultaneously.¹¹³ Work is in progress to incorporate biosensors into implantable artificial organs to provide feedback loops, for example an *in vivo* glucose biosensor connected to an insulin pump has the potential of acting as an artificial pancreas, maintaining glucose homeostasis in diabetics.¹¹⁴

For sensors employing antibodies as the biological element providing analytical specificity, a number of qualitative assays based upon the visual observation of agglutination reactions are on the market. Most common are tests for human chorionic gonadotrophin in urine which is an indicator of pregnancy.¹¹⁵ These types of tests are discussed more fully in Chapter 3.

1.5 Multianalyte Biosensors

As already mentioned individual biosensors can be incorporated into more complex instruments which are capable of measuring several parameters simultaneously.¹¹³ A major problem in the construction of a multianalyte biosensor fabricated as a single element, is the selective deposition of different biological molecules at predetermined sites (discussed in detail later in this chapter). A number of enzyme based multianalyte biosensors have been reported, using a variety of protein immobilisation methods.¹¹⁶⁻¹¹⁹ These multianalyte sensors have been employed for the simultaneous determination of small organic molecules of biological importance such as glucose, urea and galactose. A single electrochemical multianalyte immunosensing systems have been reported where several sensors or parts of sensors have been fabricated separately and joined to make an array.^{1,2}

Existing multianalyte immunosensors tend to be qualitative assays, examples of such systems are the Triage[™] test for qualitative determination of drugs of abuse,¹²⁰ and the Spectral Diagnostics assay for markers of myocardial infarction.¹²¹ These assays are based upon immunochromatography reactions, which are observed visually and are generally unsuitable for quantitative assay. Multianalyte immunosensors using optical transduction have also been reported,^{122,123} based upon glass optical waveguides or fibre-optics, where a change in absorbance or fluorescent intensity is related to the concentration of a labelled antibody close to the transducer surface, and therefore to the presence of antigen at the surface.

Simultaneous multianalyte immunoassays, where several analytes are measured by a simple procedure, are required in a number of clinical situations, such as the measurement of

hormones related with thyroid function and the measurement of gonadotrophins for the investigation of infertility.

1.6 Protein Patterning on Sensor Surfaces

An important requirement for the success of this study was the development of a method for the specific immobilisation of biochemical species at individual electrodes within an array. A problem in trying to specifically immobilise species on one electrode in an array, is in preventing non-specific adsorption of the protein to the other electrodes and the surrounding area, as the majority of proteins will adsorb very strongly to a variety of surfaces.^{124,125} Prevention of non-specific adsorption to the area surrounding the electrode is particularly important when using microelectrodes, as there is an increased diffusion to the electrode from the surrounding area with these very small electrodes when compared to macroelectrodes (see Chapter 4).

Most proteins are highly surface active, that is they have a strong tendency to accumulate at interfaces. Proteins usually have a hydrophilic surface with a hydrophobic inner, thus they tend to "unfold" on hydrophobic surfaces or stick to hydrophilic surfaces. A number of factors play a role in the adsorption process; hydrophobic interactions with non-polar surfaces, electrostatic interactions between the protein and the surface, surface heterogeneity, the solvent used, other species present in solution and the structure and possible denaturation of the protein.¹²⁶

Protein adsorption is frequently irreversible (often due to denaturation as a result of the protein unfolding), but an adsorbed protein, may be displaced by another (usually larger) protein even if the adsorbed protein does not desorb in pure solvent. Protein adsorption tends to the formation of a monolayer (although its structure may be complex) as the

process involves the loss of some of the molecules conformational and hydration energy (which is thermodynamically favourable) when it encounters an interface.¹²⁵ The result is that at the interface, the protein tends to undergo a conformational change from their globular conformation to an extended chain conformation. Protein multilayers tend not to form because, as protein is deposited on the surface, the interface becomes less distinct and it becomes less thermodynamically favourable for subsequent molecules to undergo changes in conformation and hydration. Relatively simple thermodynamic descriptions such as the Langmuir Isotherm¹²⁷ can not readily be applied to protein adsorption, as it is a complex phenomenon involving a number of processes that occur simultaneously.^{124,128} These processes include structural alterations of molecules at the interface and interactions with other adsorbed molecules. A number of more complex descriptions of protein adsorption kinetics have been suggested.^{124,126,128}

<u>1.7 Non-specific Binding</u>

Non-specific binding (NSB) is an inherent property of proteins (including antibodies) and is considered the major obstacle to the development of immunosensors.¹²⁹ In solid-phase enzyme immunoassays for single analytes, non-specific binding can be caused by three mechanisms. In the first of these mechanisms, which will be referred to as "direct reagent binding," the second antibody-enzyme conjugate binds to the surface of the solid support. This type of binding results in elevated signals that are usually relatively unaffected by the sample composition. Direct reagent non-specific binding results in a diminished ability of the assay to distinguish the "true" analyte derived signal from the elevated background signal, resulting in a poorer assay sensitivity.

The second mechanism which will be referred to as "sample mediated binding," is where the antibody-enzyme conjugate binds to the solid support via a sample derived species other

than the analyte. If this "cross-linking" species is present in equal quantities in all samples (including standards), then the effect will be similar to that observed for direct reagent binding, i.e. there is a diminished ability of the assay to distinguish the "true" analyte derived signal from the elevated background signal. If, however, the "cross-linking" species is present in differing quantities in any samples, variation of the observed signal will occur unrelated to the analyte concentration, resulting in a reduced assay specificity.

The third mechanism is "direct sample binding." This is where enzymes catalysing the same reactions as those catalysed by the enzymes conjugated to the antibodies, bind to the surface of the solid support. Direct sample binding results in poorer assay sensitivity and specificity. In multianalyte sensing systems there is a fourth type of non-specific binding; "capture antibody non-specific binding." This occurs where a capture antibody binds non-specifically to a sensor element designed for the analysis of a different antigen. This "inappropriately" bound antibody is then able to bind its antigen, and subsequent binding of a labelled second antibody can occur. This results in poorer assay specificity, and can easily lead to the production of incorrect results.

Prevention of non-specific binding is of great importance in developing a technique for immobilising organic molecules (including biomolecules) in a predetermined pattern on a surface. Non-specific adsorption of reagents or of sample components, may lead to interference and incorrect results in the assay. The techniques for reduction of non-specific adsorption are generally based upon either, coating the surface with a substance that will decrease the tendency of proteins in solution to bind to the surface, or to mask areas of the surface with a substance that can be stripped off along with any protein, leaving the bare surface exposed. A number of approaches that have the potential to solve this problem have

been reported in the literature including, pre-coating the electrode surface with protein or amino acids,¹²⁵ and using electrochemical adsorption and desorption of thioalkanes and thioalcohols.¹³⁰ These techniques are discussed in greater detail later in this chapter.

1.8 Immobilisation

A large number of techniques have been employed to immobilise the biologically active moiety of electrochemical biosensors on or in close proximity to the electrode. These can be categorised as encapsulation, entrapment, adsorption, cross-linked adsorption, electrostatic attraction, covalent attachment and affinity binding as illustrated in Figure 1.3.

1.8.1 Desirable properties of immobilisation method.

An immobilisation technique should not adversely modulate activity of the biological molecule, and should not limit access of non immobilised reactants to the immobilised molecule. This is particularly important where reactants of relatively high molecular weights (e.g. antigens) are involved. In addition, there is a requirement for an efficient sensing surface i.e.; a high surface density of functional molecules, the absence of non-specific binding, stability and durability. Several parameters such as pH, ionic strength and the surface chemistry of the solid surface influence the amount of immobilised protein at a solid surface. These parameters can be expected to influence the orientation and conformation and thereby the biological activity of the protein.^{131,132}

In immunosensors with high molecular mass analytes, those methods that immobilise the antibody by trapping it by virtue of its size (i.e. entrapment and encapsulation) will also effectively exclude the analyte from the majority, if not all, of the binding sites. Therefore, for such devices it is preferable, if not essential, to immobilise the antibody at the sensor/ sample interface.



Figure 1.3 Classification of techniques for immobilisation of biomolecules. Encapsulation involves retaining molecules that are in solution or a gel, by the use of a semipermeable membrane. Entrapment involves the trapping of the biological molecule in a porous insoluble polymeric matrix. In adsorption the biological molecule sticks to the surface by a number of physical processes such as hydrophobic interactions with non-polar surfaces, this can be further stabilised by covalent cross-linking of the adsorbed molecules. Immobilisation can also by achieved by electrostatic interactions between the charge on the biological molecule and a charged surface. In covalent attachment the molecule is tethered to the surface via covalent chemical bonds, whilst affinity binding makes use of specific interactions between biological molecules to attach the desired molecule to the surface.

1.9 Review of possible immobilisation strategies

1.9.1 Physical deposition techniques

Some workers have tried to overcome the problem of specific immobilisation by casting different membranes (e.g. cellulose acetate), each containing a specific antibody. These membranes were then placed on the electrodes.¹ This is a labour intensive technique, and would be very difficult to apply to microelectrodes. Alternatively, one can construct the electrodes separately and place them together in an array² but again, this technique is labour intensive and difficult to apply to microelectrodes.

Micropositioning has been used to deposit biological materials on specific areas of a surface. In one of its simplest forms this simply involved the micropipetting of the species into etched wells in a glass substrate.¹²² Constructing micron-scale patterns on a planar surface using this technique is difficult as it is hard to accurately and reproducibly control the spread of a liquid on surfaces particularly if the surface is hydrophilic. Other micropositioning methods include screen-printing and ink-jet printing techniques.

Screen-printing has been widely used for the selective deposition of biological molecules in the fabrication of biosensors.¹³³⁻¹³⁶ Site specific immobilisation is achieved using screen printing by placing a stencil on top of the surface to be patterned. An ink which contains the species to be immobilised is coated over the stencil, and passes through holes onto the surface below. The resolution of the technique is limited to approximately 50 μ m by a number of factors, particularly the nature of the surface and the rheology of the ink. This technique is generally used for immobilisation of proteins in relatively large patterns, but is not suitable for micron-scale patterning. The molecules tend not to be deposited as a

monolayer, which means that a considerable proportion may be inaccessible to the analyte, particularly in the case of antibody/antigen reactions.

Ink-jet printing technology has been used to specifically deposit biological molecules (principally enzymes) on surfaces.¹³⁷⁻¹³⁹ In this technique an inlet from a reservoir allows an enzyme containing solution access into a pressure chamber, which is surrounded by a piezoelectric element. The application of an electrical pulse to the piezoelectric element induces contraction of the pressure chamber, thus forcing the solution through a small nozzle (typically 50 µm diameter) onto the surface. A 50 µm diameter hole results in a surface patterned area with a diameter of between 150 and 200 µm, and the distribution of the enzyme within this area tends to be uneven (typically much greater at the periphery of the patterned area). The use of smaller nozzles coupled with treatment of surfaces so that they are hydrophobic and thus resistant to spreading of solutions, has enabled an improvement by an order of magnitude for the minimum area to be patterned.¹³⁹ Air-jet printing is a similar technique, where an aerosol produced by mixing a solution stream with an air-jet and passing though a nozzle, is directed at the sensor surface.¹⁴⁰

1.9.2 Photolithographic "lift-off" techniques

Molecular patterning has been achieved on planar surface using photolithographic processes.^{141,142} This can be achieved in two ways as shown in Figure 1.4. In the first technique, the surface is coated with photoresist, which is patterned and developed. The surface is then coated with a hydrophobic silane, and "lift-off" is performed leaving a hydrophobic pattern on the surface. The remainder of the surface is coated with a hydrophilic silane to which proteins may subsequently be attached.

In the second technique, the surface is coated with aminosilane and then photoresist which is patterned and developed. A protein species is added and "lift-off" is performed leaving the patterned protein. A second protein species can be added to complete a 2 protein pattern. Details of creating photoresist patterns and the use of lift off techniques are given in Chapter 3. There are two major problems with this technology: firstly that a maximum of two proteins can be immobilised; and secondly that of these, only one is likely to be functional as the first patterned is exposed to organic solvent during lift-off, which can cause denaturation of the protein.^{143,144}

A modified lift-off technique has been described for the patterning of enzymes in a multianalyte biosensor.¹⁴⁵ An enzyme solution containing glutaraldehyde was spin-coated onto a wafer covered with patterned photoresist. The resultant immobilised enzyme layer was removed by ultrasonic vibration in acetone, except in areas where there was no resist. Photoresist was again coated over the entire surface and repatterned. A second enzyme solution containing glutaraldehyde was spin coated onto the wafer, and again lift-off was used. A disadvantage of this process was that most of the enzyme solution was consumed without contributing to membrane fabrication, therefore the application was limited to use in sensors which utilise relatively inexpensive enzymes.¹³⁷



Figure 1.4 Techniques for molecular patterning using photoresist lift-off. In the first technique, the surface is coated with photoresist, which is patterned and developed. The surface is then coated with a hydrophobic silane, and "lift-off" is performed leaving a hydrophobic pattern on the surface. The remainder of the surface is coated with a hydrophilic silane to which proteins may subsequently be attached. In the second technique the surface is coated with aminosilane and then with photoresist which is patterned and developed. A protein species is added and "lift-off" is performed leaving the patterned protein. A second protein species can be added to complete a 2 protein pattern.

<u>1.9.3 Entrapment in Polymers</u>

A successful method of specifically immobilising enzymes at electrodes is to entrap them in an electropolymerised matrix,^{12,116,146-155} a technique which offers the possibility of the entrapment matrix reducing the effect of interfering compounds by limiting access of potential interferents to the electrode surface.^{146,153-155} These matrices allow access of low molecular weight substrates to the protein, but unless the films were very thin (thinner than the dimensions of the protein molecule) they would prevent high molecular weight antigen reaching the antibody's active site, and therefore would not be a suitable immobilisation procedure for immunosensors.

Polymerisation can also be achieved by exposure of certain electrodes to electromagnetic radiation. X-rays, γ -rays and UV and visible light have all been used.¹⁵⁶⁻¹⁵⁸ This method of entrapping protein suffers from the same disadvantages as the electropolymerised matrix, but additionally the protein can be severely damaged by the radiation rendering it non-functional.

A third approach is to form a polymer film on the electrode (by electro- or photopolymerisation) and then to adsorb or covalently link the antibody to the polymer.^{159,160} Polymers have been used to influence the orientation of molecules and can provide direct electronic communication between a redox protein and an electrode.¹⁶¹ A major problem with polymerisation techniques for patterning biological molecules for use in multianalyte immunosensors, is in preventing non-specific binding to the polymer or the uncoated electrodes.

1.9.4 Electrodeposition followed by covalent cross-linking.

The migration of proteins to electrically charged electrodes, followed by cross-linking with glutaraldehyde has been used, as a means of selectively depositing enzymes on electrode surfaces.^{144,162} The migration of proteins in an electric field is dependent upon the isoelectric point (pI) of the protein and the pH of the supporting buffer. At pH values greater than the pI the protein molecule possesses a net negative charge, and will migrate towards a positively charged electrode. Once the protein has been deposited at the electrode, it can be secured by forming covalent linkages using glutaraldehyde as a crosslinker. The technique results in the formation of protein multilayers at the electrode, the thickness of which can be controlled by variation of the immobilisation conditions. It is generally considered that the protein layer that is in direct contact with the electrode surface is denatured and that protein functionality resides in ensuing layers.¹⁶³ A multianalyte sensor for the simultaneous determination of glucose and lactate has been fabricated using this technique, but it has not been applied to immunoassays due to a number of problems; firstly the technique is rather wasteful of the molecule to be immobilised as it is in a relatively thick layer, part of which has been denatured;¹⁶³ secondly with high molecular weight analytes, many of the antibody binding sites within the deposited layer will be unavailable to the antigen.

1.9.5 Techniques using deep ultraviolet light

The coating of the entire surface with protein followed by selective removal using deep ultraviolet (DUV) irradiation ($\lambda = 193$ nm) has been used to pattern IgG onto a surface.¹⁴³ The exact mechanism was not elucidated, but the irradiation caused either protein cleavage or denaturation in exposed areas. It was found, however, that the functionality of the remaining IgG was variable, and non-specific binding remained a major problem.

1.9.6 Langmuir-Blodgett

Orientation of immunoglobulins has been demonstrated using Langmuir-Blodgett (LB) films, thus maximising the immunoglobulin's efficacy.^{164,165} Electrostatic interaction between proteins and Langmuir-Blodgett films has also been used as an immobilisation technique.¹⁶⁶ However, LB technology has not been successfully extended to patterning surfaces in a predefined manner, and the problems of specific immobilisation remain.

1.9.7 Self assembled monolayers

Patterns of hydrophilic and hydrophobic SAMs (each element 20µm in size) have been constructed on gold by using a stamp to attach an alkanethiol to selected areas of a gold surface (sulphur readily binds to gold),¹⁶⁷ a mercaptoalcohol is then attached to the remainder of the surface. SAMs have also been used to pattern a protein on a surface. In particular SAMs formed by the chemisorption of thiol alcohols, especially oligo(ethylene glycol)-terminated alkanethiols (HS(CH₂)_m(OCH₂CH₂)_nOH), have been shown to resist the adsorption of a variety of proteins.¹⁶⁸

Patterns of pyruvate kinase, with resolution of a few μ m, have been formed on gold by stamping the surface with HS(CH₂)₁₅CH₃ and then derivatising the rest of the surface with HS(CH₂)₁₁(OCH₂CH₂)₆OH, followed by protein adsorption as shown in Figure 1.5. A polydimethylsiloxane (PDMS) stamp was fabricated using a resist master pattern, this mask was peeled away and inked with an alkanethiol (hydrophobic) which was transferred as a pattern onto the gold surface by stamping. The surface was then exposed to a solution containing HS(CH₂)₁₁(OCH₂CH₂)₆OH (hydrophilic), and upon subsequent exposure to protein, adsorption occurred at the hydrophobic but not the hydrophilic areas.

This technique has the potential of constructing very high resolution patterns, as resolution is not limited by the wavelength of light, however it is limited to the patterning of a single protein. Patterns of SAMs could be formed in a similar way using resist based lift-off techniques (lift-off described in Chapter 3). The use of electron beam lithography would allow patterns to be formed whose resolution is not limited by the wavelength of light.



Figure 1.5. Schematic diagram for the construction of a protein pattern by means of stamping SAM films. (a) A polydimethylsiloxane (PDMS) stamp was fabricated using a resist master pattern, (b), this mask was peeled away and (c) inked with an alkanethiol (hydrophobic) which was (d) transferred as a pattern onto the gold surface by stamping. (e) The surface was then exposed to a solution containing $HS(CH_2)_{11}(OCH_2CH_2)_6OH$ (hydrophilic), and (f) upon subsequent exposure to protein, adsorption occurred at the hydrophobic but not the hydrophilic areas.

It has been suggested that self assembly combined with LB techniques could be used to pattern proteins on gold.¹⁶⁹ A mixture of palmitic acid and a sulphur containing lipid (thiolipid) was spread on the water/air interface of a Langmuir trough. On compression the components of the floating monolayer separated and domains were formed which

predominantly contained palmitic acid. The film was transferred to a gold substrate where the thiolipids covalently bound to the gold and the physisorbed palmitic acid was washed away to leave domains of bare gold to which other species such as proteins could be attached. This technique would have the advantage that resolution of patterning would not be limited by the wavelength of light as in photoactivation/deactivation or photolithographic techniques. However, methods for precisely controlling the size, geometry and position of this pattern have not yet been reported, additionally this technique would only be of value for patterning a single protein.

Protein patterning has been performed using DUV irradiation of SAM films of organosilanes.^{143,170} This procedure is represented diagramatically in Figure 1.6. DUV irradiation ($\lambda = 193$ nm) of the thiol groups results in their oxidation to sulphonate groups. Protein is then attached to the remaining thiol groups (on areas of the surface that have not been irradiated) either directly or via a heterobifunctional cross-linker. This method achieved a resolution of less than 10 µm, but it is limited to the patterning of a single species and at best non-specific binding was reduced to 15%.

1.9.8 Masking using thioalkanes

It has been suggested that it is possible to mask electrodes to prevent antibody adsorption, whilst leaving other electrodes exposed so that the antibody can be immobilised, by the use of species (for example thioalkanes) that are adsorbed or desorbed at particular electrode potentials.¹³⁰ The successful application of this technique to patterning proteins on surfaces has not been reported. A potential disadvantage of these techniques is that the use of organic solvents is required which may affect the structural integrity and therefore the function of proteins.¹⁴⁴ Additionally, these techniques would be wasteful of biomolecules.



Figure 1.6 Schematic diagram for the patterning of proteins by DUV irradiation of SAM films. DUV irradiation ($\lambda = 193$ nm) of the thiol groups results in their oxidation to sulphonate groups. Protein is then attached to the remaining thiol groups (on areas of the surface that have not been irradiated) either directly or via a heterobifunctional cross-linker. Protein does not attach to the sulphonate groups.

1.9.9 Photoactivation and photodeactivation techniques

Fodor and co-workers have reported the light-directed spatially addressable parallel chemical synthesis of oligopeptides and oligonucleotides.^{171,172} They have combined the use of solid phase chemistry, photolabile protecting groups and photolithography to synthesise a range of chemical products. The principle of the technique is illustrated in Figure 1.7. The substrate possesses amino groups that are blocked with a photolabile protecting group such as nitroveratryloxycarbonyl (Nvoc), illumination of specific regions through a lithographic mask results in photodeprotection. The exposed amino groups are thus available for coupling to another species (which has a photolabile protecting group attached) using conventional solid phase synthesis methods.¹⁷³ A different mask is used to photodeprotect different regions of the substrate, and a different photolabile protected group containing species is attached to the exposed amino groups. Additional cycles of photodeprotection and coupling are performed to obtain the desired set of products.

This technology is capable of rapidly synthesising large numbers of oligopeptides and oligonucleotides. With a simple binary mask, 256 different species can be synthesised in only 8 steps, the use of more complex masks can increase the number of species synthesised.¹⁷⁴ Resolution of between 10 and 20 µm has been achieved within the arrays of synthesised oligopeptides.¹⁷⁵

The formation of a covalent linkage between the protein and the substrate by means of a photoactivatable link has previously been reported.¹⁷⁶ In some of these techniques, a photoactivatable function provided by an aryl azide (e.g. 3-nitro-4-aminophenyl azide) was attached directly to a silica surface. The modified surface was covered with an enzyme containing solution, and designated areas were exposed to light by using a mask. Assay of



Figure 1.7 Schematic diagram for light-directed spatially addressable chemical synthesis. The substrate possesses amino groups that are blocked with a photolabile protecting group such as Nvoc. Illumination of specific regions through a lithographic mask results in photodeprotection. The exposed amino groups are thus available for coupling to another species; A (which has a photolabile protecting group (x) attached) using conventional solid phase synthesis methods. A different mask is used to photodeprotect different regions of the substrate, and a different photolabile protected group containing species is attached to the exposed amino groups. Additional cycles of photodeprotection and coupling are performed to obtain the desired set of products.

potters one protein with a resolution of approximately 100 µm. The activity of the solution of approximately 100 µm. The activity of the surface tends to be non-uniform possibly due to deposition of increasing the surface tends to be non-uniform.

enzyme activity was performed by monitoring the formation of fluorescent products, the observed fluorescence reproduced the pattern of the mask. This procedure did not address the problem of preventing non-specific binding, and was only used to pattern one protein at a time.

Benzophenone has been used as a photoactivated cross-linker to pattern fluorescently labelled proteins to glass (Figure 1.8),¹⁷⁷ although this method suffered greatly from non-specific binding which at best was 20%. The method was not used to attach more than one species to the surface.

1.9.10 Miscellaneous Techniques

Another method of selective attachment is to immobilise the antigen, and then add a multivalent antibody. Some of the antibody binding sites will bind to the immobilised antigen, whilst others will be available to bind the antigen that is in solution.¹⁷⁸ The problem of how to selectively immobilise the antigen remains, and for high molecular mass antigens such as in this study, it is no easier than to immobilise the antibody itself.

Finally, functional antibodies and enzymes have been patterned on to surfaces by laser induced plasma deposition. In this technique biomolecules are vapourised and ionised by the laser, the ionised proteins are guided by electric fields towards a substrate on which they are deposited. Patterning can be achieved by placing a mask (copper grid) on the substrate, or by using spatially well defined electric fields. This technique has only been used to pattern one protein with a resolution of approximately 100 μ m. The activity of the deposited protein over the surface tends to be non-uniform possibly due to deposition of large clusters consisting of several molecules.^{179,180}



Figure 1.8 Use of benzophenone as a photoactivatable cross linker for protein immobilisation. Benzophenone is attached to the surface via aminopropyltroethoxysilane (note that this is not 100% efficient). The benzophenone-derivatized surface is treated with a solution of the protein (Ig) and then irradiated through a lithographic mask. The protein molecules are bound covalently only in areas that are exposed to light.

CHAPTER 2

Se vogliamo che tutto rimanga come è, bisogna che tutto cambi

(If we want things to stay where they are, things will have to change)

> Guiseppe di Lampedusa Il Gattopardo (The Leopard)

2 Introduction to Initial Experimental Work on Protein Patterning Techniques.

Initial investigations were performed into the feasibility of using a variety of techniques for the site specific immobilisation of antibodies. Of the techniques studied, which included employing very thin polymer films and masking of electrodes using thioalkanes (data not shown), the use of a photoactivable binding agent appeared to be the most promising. Section 2.1 provides an introduction to the theory and principles, of the protein patterning methods, and of work carried out in investigating these techniques. Sections 2.2 to 2.12 provide materials used and detailed experimental methods. The remainder of the chapter (sections 2.13 to 2.22) presents the results obtained along with discussion of their significance.

2.1 The use of a Photoactivable Binding Agent for Immobilisation of Biological Molecules.

The possibility of patterning proteins by the use of photoactivable binding agents was investigated, particularly the use of a photoactive biotin analogue (photobiotin), in conjunction with avidin, a molecule with which biotin exhibits a specific affinity. Avidin is a protein with a molecular mass of approximately 60,000 found in egg whites (a microbial form streptavidin is obtained from Streptomyces avidinii) and possesses four binding sites for biotin (vitamin H) with which it forms a very stable complex, with an association constant $K_a = 10^{15}$ M⁻¹ ¹⁸¹⁻¹⁸³ corresponding to a free energy of association of about 330 KJ M⁻¹ of tetramer (81 Kcal M⁻¹).^{184,185} The role of avidin - biotin bonding in nature, and the significance of the bond strength is unknown, although avidin and biotin exhibit the highest known affinity between a protein and ligand, with a binding energy which is comparable to that of a covalent linkage.

Both avidin and streptavidin consist of 4 subunits, each of which is an 8 stranded antiparallel β sheet forming a β -barrel. The biotin binding site is situated near the end of the barrel and involves a highly stabilised network of polar and hydrophobic interactions.¹⁸¹ Biotin can be conjugated to proteins or other biologically active species, and this complex can be attached to immobilised avidin.¹⁸⁶⁻¹⁸⁹ The rationale in using the avidin - biotin system is based on the premise that if a biologically active species is modified with biotin through its side chain, the biological and physicochemical properties of the biotin modified molecule will not be significantly affected, and that the biotin is still available to bind to avidin or streptavidin. This system has been extensively used in immunoassay techniques.^{189,190} Avidin tends to show a high degree of non-specific binding to substances which is usually attributed to two inherent characteristics of the native molecule; a high pl and the presence of carbohydrate residues. Streptavidin, which has a lower carbohydrate content, and deglycosylated avidin, both exhibit reduced non-specific binding as compared to intact avidin.^{185,191} The high isoelectric point (pI 10.0 - 10.5) of avidin, as well as the presence of carbohydrate (which accounts for about 10% of avidin's total molecular weight) on the protein contribute to non-specific binding in applications involving avidin. Due to its more favourable isoelectric point and lack of carbohydrate, streptavidin has historically been the accepted substitute for applications where the non-specific binding characteristics of avidin can not be tolerated. More recently, modified forms of avidin have become available, with pI values closer to 7.0 and reduced carbohydrate content leading to improved non-specific binding characteristics.^{192,193} Streptavidin and deglycosylated avidin have been utilised as coatings on immunoassay microtitre plates to minimise non-specific binding.194

Photobiotin is a photoactivable biotin analogue (N-(4-azido-2-nitrophenyl)-N'-

(N-d-biotinyl-3-aminopropyl)-N'-methyl-1,3-propanediamine), which was developed as a nucleic acid hybridisation probe,¹⁹⁵ and has also been used to label proteins.¹⁹⁶ Photobiotin contains an aryl azide group which is stable in the dark, but upon exposure to ultraviolet / blue light (340 - 375 nm) a highly reactive aryl nitrene group is generated which will react with proteins (and other species).¹⁹⁶ The potential effects of steric hindrance preventing binding between an activated aryl nitrene group and a protein can be minimised by introducing a spacer arm into the photobiotin molecule.¹⁹⁵ Such a spacer arm contains basic tertiary amino groups, which are positively charged at neutral pH and give a high solubility in water. The structure of the long arm photobiotin used in this work is shown in Figure 2.1



Figure 2.1 Chemical structure of long arm photobiotin. Photobiotin is a photoactivable biotin analogue, containing an aryl azide group which is stable in the dark, but upon exposure to ultraviolet / blue light (340 - 375 nm) a highly reactive aryl nitrene group is generated which will bind non-specifically to organic molecules. The introduction of a spacer arm minimises steric hindrance, and this spacer arm contains basic tertiary amino groups, which are positively charged at neutral pH and give a high solubility in water.

A number of approaches to the use of photobiotin and avidin in the immobilisation of molecules were examined;

a) Depositing a protein layer on the surface, adding photobiotin in the presence of light, followed by the attempted immobilisation of a second protein through avidin-biotin interactions.

b) Depositing a protein layer on the surface, adding avidinated protein with photobiotin attached and exposing to light;

c) Depositing an avidin layer on the surface, adding photobiotin followed by the addition of the species to be immobilised in the presence of light.

2.1.1 Depositing a protein layer on the surface, adding photobiotin followed by the immobilisation of a protein through avidin-biotin interactions

In this technique bovine serum albumin (BSA) was deposited on a surface which was subsequently exposed to light in the presence of photobiotin. This was followed by the sequential addition of avidin and a biotinylated protein, the rationale being that the avidin would bind to the immobilised biotin moiety and to the biotinylated protein resulting in the biotinylated protein's immobilisation. This is represented diagramatically in Figure 2.2. Initially, BSA was immobilised on to the surface (Figure 2.2a), photobiotin was added and exposure of selected areas of the surface to light through a mask resulted in activation of the photobiotin molecule (Figure 2.2b). Unbound photobiotin was removed by washing, and avidin was added which bound to the immobilised photobiotin groups (Figure 2.2c). A biotinylated protein was then added which bound to the immobilised avidin (Figure 2.2d).



Figure 2.2 Scheme for depositing a protein layer on the surface adding photobiotin and immobilising proteins through avidin-biotin interactions. BSA was immobilised on to the surface (a), photobiotin was added and exposure of selected areas of the surface to light through a mask resulted in activation of the photobiotin molecule (b). Unbound photobiotin was removed by washing, and the surface was incubated with a solution of avidin which bound to the immobilised photobiotin groups (c). A biotinylated protein was then added which bound to the immobilised avidin (d).

2.1.2 Depositing a protein layer on the surface, and add avidinated protein with photobiotin attached

In this technique BSA was deposited on a surface, followed by the addition, in the presence of light, of an avidinated protein which had been preincubated with photobiotin. The rationale behind this method was that the photobiotin (and as a consequence of the avidin biotin interaction, the avidinated species), would bind to the protein coated surface. This is depicted diagramatically in Figure 2.3. Initially a protein - avidin - photobiotin complex was formed (Figure 2.3a), which was then placed in contact with a surface on which BSA had been immobilised (Figure 2.3b). Exposure of selected areas of the surface to light through a mask resulted in activation of the photobiotin molecule which attached to the surface BSA, thus immobilising the attached protein (Figure 2.3c).

2.1.3. Depositing an avidin layer on the surface, adding photobiotin followed by the addition of the species to be immobilised.

A third approach to protein patterning developed in this study was the use of photoactivable binding using photobiotin, which was bound to an avidin covered surface,¹⁹⁷⁻¹⁹⁹ the immobilisation procedure is shown in Figure 2.4. Initially, avidin was immobilised on the surface and photobiotin was added and bound to the avidin (Figure 2.4a). Exposure of selected areas to light through a mask resulted in activation of the photobiotin molecule and a protein in the solution was immobilised specifically (Figure 2.4b). Unbound material was removed by washing, and the procedure repeated with a second protein (Figure 2.4c), the entire surface was then exposed to light in the presence of an agent such as casein in order to block previously unreacted photobiotin (Figure 2.4d).



Figure 2.3 Scheme for immobilising an avidinated protein by depositing a protein layer on the surface and adding the avidinated protein with photobiotin attached. A protein - avidin - photobiotin complex was formed (a), which was then placed in contact with a surface on which BSA had been immobilised (b). Exposure of selected areas of the surface to light through a mask resulted in activation of the photobiotin molecule which attached to the surface BSA, thus immobilising the attached protein (c).





2.1.4 Investigation into resolution of avidin-photobiotin immobilisation procedure.

Investigations were performed to determine the resolution of patterned proteins that were achievable using the patterning technique that was selected as being the best of the methods (the criteria being relative efficiency of the immobilisation process, and contrast between patterned and non-patterned areas, results shown later in this chapter). This was done by patterning of proteins using a photolithographic mask with equal mark:space ratio lines of different widths. The patterned proteins were visualised by the addition of fluorescently labelled antibodies, and resolution was examined by measurement of the relative fluorescent intensity across the pattern.

2.1.5 Investigation into timescale of avidin-photobiotin immobilisation procedure.

The aim of this work was to determine the time required to complete the patterning process, under defined conditions (photobiotin being activated by light from a a 100W HB-10101AF super high pressure mercury vapour lamp (Nikon, Tokyo, Japan) situated 185 mm from the substrate (irradiance $\sim 9 \text{ mW cm}^{-2}$)). The distance of 185 mm was chosen, as this was the height of a chamber that was constructed to house samples being irradiated and providing protection to the user. Irradiance was directly measured by the use of a calibrated photovoltaic cell.

2.1.5.1 Activation of Photobiotin

The photoactivation of the photobiotin molecule (Figure 2.1) is due to the incorporation of an aryl azide. Aryl azides are derivatives of benzene and fused ring analogues, in which the azide (N_3) function is attached to a carbon that is part of the aromatic ring, i.e. all aryl

azides contain the group shown in Figure 2.5. The bonds in aryl azides are in equilibrium, as shown in Figure 2.6.



Figure 2.5 Structure of azidobenzene, the structure common to all aryl azides



Figure 2.6 Equilibrium between bonds in aryl azide.

Aryl nitrenes are generated by thermolysis or photolysis of aryl azides which readily lose 2 nitrogen atoms as N_2 as shown in Equation 2.1

Aryl nitrenes are extremely reactive and will undergo a large variety of reactions, some of which are shown in Figure 2.7.²⁰⁰ Aryl nitrenes are very short lived species (half lives less than a second) but they may rearrange to give less reactive species, with longer half lives. Azidobenzene itself absorbs short - wavelength light ($\lambda_{max} = 250 \text{ nm } \varepsilon = 10,000$) having a spectra with characteristic shoulders at 277 and 286-288 nm, but in aryl azides irradiation of the aromatic system allows transfer of energy to the azide function.^{201,202} Nitro substitution of the aromatic ring leads to the appearance of an absorbance band at longer wavelengths so that light with a λ of around 350 nm will cause the production of nitrenes from the nitro substituted aryl azide present in photobiotin.

As the aryl nitrene group will bind to any organic molecules, the attachment of photobiotin to molecules should not be performed in buffers that contain organic material such as Tris(hydroxymethyl)methylamine (Tris).²⁰³ Additionally, azides are rapidly converted to amines by thiol groups, commonly used in biochemical buffers in order to produce a reducing environment. The presence of electron withdrawing groups on the aromatic ring may be used to increase the reactivity of aryl nitrenes. The inclusion of such groups also affects the specificity of the reaction, making reactions with nucleophiles much more rapid at the expense of insertion reactions.²⁰⁴


Further Rearrangements

Figure 2.7 Some of the reaction of aryl azides upon activation.²⁰⁰ They can react with amine groups and saturated and unsaturated carbons. Alternatively they can react with oxygen or acids, dimerise or undergo spontaneous rearrangements.²⁰⁰

2.1.6 Light activated patterning of multiple proteins on a silicon dioxide surface.

Once a patterning method had been selected (criteria being relative efficiency of the immobilisation process, and contrast between patterned and non-patterned areas, results shown later in this chapter), further work was performed to ascertain the possibility of patterning multiple proteins using this technique. Initially this was using just two proteins, but further work was performed to demonstrate the application of the developed patterning technique to the site-specific attachment of six different proteins (including 5 functional antibodies) within an area of $200 \times 50 \,\mu\text{m}$.

2.1.7 Demonstration of light-dependent coupling of enzymes and antibodies to gold, glass and carbon surfaces

Once a patterning method for the immobilisation of proteins on silicon dioxide had been selected, further work was performed to demonstrate the application of the immobilisation procedure to a range of surfaces, and to examine further characteristics of the technique. The aim of this section of work was to demonstrate the application of the patterning technique to the immobilisation of enzymes (GOD) and/or antibodies on to gold, glass and carbon surfaces. Successfully patterning enzymes and antibodies onto gold and/or carbon would allow the technique to be used for the fabrication of multianalyte amperometric enzymic biosensors and immunosensors.

2.1.8 Investigation into reducing non-specific binding on gold surfaces.

For the protein patterning to work efficiently, and in order to use the technique to develop a multianalyte electrochemical immunosensor, it was necessary to minimise non-specific binding of proteins to the surface. The effect upon non-specific binding of using different forms of avidin, different avidin immobilisation methods and different blocking reagents were examined. The types of avidin used were streptavidin (Vector) and two deglycosylated forms of avidin; avidin D (Vector) and Neutravidin (Pierce). The avidin immobilisation methods employed were adsorption, using N-acetyl-1-cysteine (NAC) with 1-ethyl-3-(3-dimethyl- aminopropyl)- carbodiimide (EDC), and using 3,3'-dithiobis(sulfosuccinimidylpropionate) (DTSSP). The three potential blocking agents examined were BSA, casein and Superblock[™].

2.1.8.1 Immobilisation of avidin on gold using carbodiimide coupling to a N-acetyl-cysteine self assembled monolayer.

In this procedure NAC was assembled onto a gold surface through chemisorption of the thiol group. Subsequent attachment of the avidin to the NAC was by means of a carbodiimide. Carbodiimides comprise a group of compounds with the general formula R-N=C=N-R', where R and R' can be aliphatic or aromatic. This group of compounds has long been used for the synthesis of peptide bonds.²⁰⁵ The carbodiimide used in this work was 1-ethyl-3-(3-dimethylaminopropyl)carbodiimide (EDC).

The reaction of carbodiimides may be represented as a dehydration via an intermediary as shown in Figure 2.8. ²⁰⁵⁻²⁰⁷ (Temperatures of around 0°C favours the formation of peptide bonds, at higher temperatures the reaction shifts towards the formation of acyl ureas ²⁰⁷ (Figure 2.9)).



Figure 2.8 Reaction scheme for the promotion of peptide bond formation between amino and carboxylic acid residues in two proteins by use of carbodiimides.





2.1.8.2 Immobilisation of avidin on gold using DTSSP

The structure of the 3,3'-dithiobis(sulfosuccinimidylpropionate) (DTSSP) is shown in Figure 2.10. DTSSP possesses two N-hydroxysuccinimide (NHS) esters connected by a disulphide linkage. The sulphur atoms in the disulphide bond will chemisorb to gold surfaces with a stability greater than that given by covalent silane bonds with glass. NHS esters react with primary amine - containing species to from stable amide (peptide) bonds (Figure 2.11). NHS esters also react with thiols.



Figure 2.10 Structure of DTSSP (3,3'-dithiobis(sulfosuccinimidylpropionate)). The molecule possesses two N-hydroxysuccinimde (NHS) esters connected by a disulphide linkage.



Figure 2.11 Coupling of protein to a DTSSP activated gold surface. The sulphur atoms in the disulphide bond chemisorb to gold surfaces with great stability. NHS esters react with primary amine - containing species (e.g. proteins) to from stable amide (peptide) bonds.

2.1.9 Regeneration of avidinated surfaces

Investigations were performed to elucidate if it was possible to break the avidin photobiotin bond, whilst the avidin remained immobilised on the surface. pH extremes and chemicals (urea and guanidine) which are known to disrupt non-covalent bonds,²⁰⁸⁻²¹⁰ were examined. Being able to break the avidin-photobiotin bond, whilst leaving the immobilised avidin intact would introduce the possibility of regenerating and repatterning the surface, which could be advantageous if one wished to repattern and reuse the transducer element of an immunosensor (which is desirable if using expensive transducer elements).

2.1.10 Characterisation of biomolecular patterns using atomic force microscopy and fluorescence microscopy.

Atomic force microscopy was utilised to ascertain if information on the avidin-photobiotin patterning technique could be obtained, and comparing this information with data from fluorescent microscopy. Since its invention a little over a decade ago²¹¹ scanning probe microscopy (SPM) has rapidly become a powerful technique for the imaging and analysis of surfaces. Scanning probe microscopy, (including atomic force microscopy (AFM) and scanning tunnelling microscopy (STM)) has been used to study the structure of immobilised biological molecules.²¹²⁻²¹⁴

The principle of scanning tunnelling microscopy (STM) is simple, when a bias voltage (0.1-1.0 V) is applied to a metallic tip, and advanced towards a conducting surface a tunnelling current is initiated at an appropriate distance from the surface (0.1-10 nm), due to electrons tunnelling from the tip through the gap to the surface. The size of the current is dependent upon on the work function of the tip and the bias voltage and is very sensitive to

the gap distance. Typically a change in distance of 0.1 nm results in an order of magnitude change in current. This critical relationship between distance and current gives rise to the vertical resolution of the technique.

Typically, in an experiment the tip is rastered across the surface controlled by a piezoscanner and the height of the tip above the surface is kept constant by means of feedback control (constant current mode), thus providing an image of the surface topography. Alternatively the tip is kept at a constant absolute height and changes in the current are recorded (constant height mode). STM requires that both the tip and the surface to be conducting. The advantage of STM over scanning electron microscopy (SEM) is primarily that of resolution, STM being capable of atomic resolution for flat surface,²¹⁵ and molecular resolution when imaging biomolecules.²¹⁴ Additionally, STM unlike SEM does not require the use of a high vacuum and images can be obtained at ambient temperature and pressure.

In AFM the elastic response of molecules to the force exerted by a scanning tip are measured. As the tip is rastered over the surface, changes in the surface cause small deflections in a spring mounted tip which can be measured and processed to give a 3 - dimensional image of the surface. The great advantage of AFM as compared with STM is that it can be used to image non-conducting surfaces. The main problem in imaging "soft" organic and biological material with an AFM is that the relatively high force exerted by the cantilever tip onto the sample may cause distortion or it may even be destructive.^{216,217}



Figure 2.12. Diagram showing principle of operation of scanning tunnelling microscope. The tip is rastered across the surface controlled by a piezoscanner and the height of the tip above the surface is kept constant by means of feedback control (constant current mode), thus providing an image of the surface topography (represented by the red line). Alternatively the tip is kept at a constant absolute height and changes in the current are recorded (constant height mode).

Spring deflection sensor



Figure 2.13. Diagram showing principle of operation of atomic force microscope. The tip is rastered over the surface, changes in the surface cause small deflections in the spring mounted tip (represented by the red line) which can be measured and processed to give a 3 - dimensional image of the surface.

2.2 Materials

Photobiotin, avidin D and Streptavidin were obtained from Vector Laboratories, Peterborough, England. IgG from chicken, guinea pig, goat, mouse, rabbit and rat, all antibodies and their conjugates (fluorescent and enzymic), ferrocenemonocarboxylic acid, 1-ethyl-3-(3-dimethylaminopropyl)carbodiimide, N-acetyl-1-cysteine, sodium cyanoborohydride, guanidine, urea, Sephadex G25 & G200, p-nitrophenol phosphate and glucose oxidase were obtained from Sigma, Fancy Road, Poole, Dorset, England. Neutravidin, bovine serum albumin, casein, N-succinimidyl 4-(n-maleimido- methyl)cyclohexane-1-carboxylate and Superblock[™] were obtained from Pierce Warriner, Chester, England. 1,3-trimethoxysilylpropyl-ethylene diamine, ethanol, sodium dihydrogen phosphate, disodium hydrogen phosphate, sodium chloride, glutaraldehyde, Tris, hydrochloric acid, dimethylformamide, zinc chloride and magnesium chloride were obtained from BDH, The Old Brickyard, Poole, Dorset, England.

Positive S1400-31 photoresist and Microposit developer were obtained from Shipley Europe, Coventry, England. Silicon wafers were obtained from Shin Etsu, Livingston, Scotland; NiChrome (80:20 Ni:Cr) and gold were obtained from Goodfellow Cambridge Ltd, Cambridge, England.

Bioanalytical Systems (BAS) gold electrodes, glassy carbon electrodes (BAS) and BAS RE4 Ag/AgCl reference electrodes were obtained from Biotech Instruments Ltd (Luton, England).

Instrumentation used comprised a EG&G 273A potentiostat (EG & G, Sunninghill, England), a Cecil Series 5000 spectrophotometer, Cambridge, England and a 100W HB-10101AF super high pressure mercury vapour lamp, Nikon, Tokyo, Japan.

2.3 Methods for Patterning of Proteins using Avidin and Photobiotin.

The use of photobiotin and avidin in the site specific immobilisation of molecules were examined using three different formats as outlined in sections 2.1.1 to 2.1.3.

2.3.1 Method for patterning of an avidin-protein conjugate by depositing a protein layer on the surface, adding photobiotin followed by the addition of the avidinated protein.

A SiO₂ wafer was immersed in 1% 1,3-trimethoxysilylpropyl-ethylene diamine in 95:5 (v/v) ethanol / distilled water for 120 seconds and briefly rinsed in 95:5 (v/v) ethanol / distilled water before heating at 120°C for 30 minutes. The wafer was immersed in 2% glutaraldehyde in 100 mM phosphate buffered saline pH 7.4 (PBS) for 15 minutes, and in 40mM sodium cyanoborohydride, 0.2mg ml⁻¹ BSA in PBS for 30 minutes. The SiO₂ substrate was washed in PBS after this and all subsequent steps. The BSA-modified wafer was incubated in 5 ml of 10 μ g ml⁻¹ long arm photobiotin in PBS whilst exposed to light from a 100 W high pressure mercury vapour lamp sited 185 mm from the wafer, for 15 minutes through a 10 μ m interval equal mark space ratio mask (that had been fabricated photolithographically), following which the wafer was incubated in 20 μ g ml⁻¹ avidin in PBS for 60 minutes. After washing in PBS, the wafer was incubated in 20 μ g ml⁻¹

The wafer was then incubated in 10 μ g ml⁻¹ tetramethylrhodamine isothiocyanate (TRITC) labelled goat anti-rabbit IgG in PBS, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescence microscopy. Results are shown in Section 2.13.1, a photomicrograph of the resultant surface

is shown in Figure 2.14 and a cross sectional profile showing fluorescent intensity is shown in Figure 2.15.

Profiles of relative fluorescent intensity were obtained by digitising images by scanning into a computer using Photoshop[™] software. A plot of relative fluorescent intensity was then produced by assigning a value of 100 to the brightest point (i.e. greatest fluorescence) on a line across the pattern. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). It should be noted that this zero value was based upon the software definition of black, and that it was not obtained from the photographic image. This was done so that a more objective measure of contrast and non-specific binding could be obtained.

2.3.1.1 Method for patterning avidin by depositing a protein layer on the surface, adding photobiotin followed by the addition of avidin

BSA was coated on to a SiO₂ surface, and photobiotin was subsequently attached as described in Section 2.3.1, following which the wafer was incubated in 20 μ g ml⁻¹ avidin-TRITC in PBS for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescence microscopy. Results are shown in Section 2.13.1.

2.3.2 Method for patterning a protein by depositing a protein layer on the surface, and add avidinated protein with photobiotin attached

Avidin D was conjugated to rabbit IgG by using a maleimide as the conjugation reagent. 8 mg ml⁻¹ avidin D was dissolved in 50 mM sodium borate buffer, pH 7.6 containing 1mM MgCl₂ and 0.1 mM ZnCl₂. To 1.0 ml of this avidin solution was added 1 mg of N-succinimidyl 4-(n-maleimido- methyl)cyclohexane-1-carboxylate (dissolved in 50 μ l of N, N-dimethylformamide), and the mixture was incubated at room temperature for 1 hour. The reaction mixture was then applied to a Sephadex G-25 column (1.0 × 10 cm) equilibrated with 100 mM Tris-HCl buffer pH 7.0. Reaction products were eluted using the same buffer, and 1.0 ml fractions were collected. The protein content of these fractions was assessed by the measurement of absorbance at 280 nm using a Cecil Series 5000 spectrophotometer, and protein containing fractions were pooled and concentrated to 1.0 ml. Subsequently, 2.5 mg of rabbit IgG was dissolved in 0.5 ml of 50 mM sodium acetate buffer, pH 5.0 and this was added to the maleimide- avidin preparation and incubated at 4°C for 20 hours. 20 µl of 100 mM 2-mercaptotoethylamine was added to the reaction mixture and incubated at room temperature for 30 minutes to block remaining maleimide groups.

In order to separate the conjugated and unreacted material, the reaction mixture was applied to a Sephadex G-200 column $(1.0 \times 10 \text{ cm})$ equilibrated with 100 mM Tris-HCl buffer pH 7.0. Elution of reaction products was undertaken using the same buffer, and 1.0 ml fractions were collected. The relative protein content of these fractions was assessed by the measurement of absorbance at 280 nm and those that were earliest eluting (therefore having the highest molecular weight) were used as the avidin-IgG conjugate. The integrity of the conjugate was validated by use of a colourimetric ELISA, where the capture antibody was directed against avidin and the conjugate was a goat anti-rabbit IgG-alkaline phosphatase.

BSA was immobilised on a SiO₂ wafer using the silanisation procedure described in Section 2.3.1. 2.5 ml of 20 μ g ml⁻¹ avidinated rabbit IgG was incubated with 2.5 ml of 20 μ g ml⁻¹ long arm photobiotin in PBS for 20 minutes. This solution was then placed on the protein coated SiO₂ wafer and exposed to light as described in Section 2.3.1. The wafer was then incubated in 10 μ g ml⁻¹ TRITC labelled goat anti-rabbit IgG in PBS, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy. Results are presented in Section 2.13.2.

2.3.2.1 Spectrophotometric determination of alkaline phosphatase activity.

Alkaline phosphatase (ALP) (E.C.3.1.3.1) activity was determined spectrophotometrically using p-nitrophenol phosphate (pNPP) as a substrate. pNPP is essentially colourless in solution, but when the phosphate is removed by the action of ALP the resultant p-nitrophenol gives an intense yellow colour with an absorbance maxima at 404 nm.

ALP was assayed by the addition to the sample of a 100 mM Tris-HCl buffer pH 9.8, containing 1mM MgCl₂ and 10 mM pNPP. Samples were incubated at room temperature for 15 minutes before reading the absorbance at 404 nm (proportional to ALP activity).

2.3.2.2. Assessment of cross reaction between photobiotin conjugates in solution.

To assess whether crosslinking occurred between the avidinated species-photobiotin complexes in solution, 2.5 ml of 20 μ g ml⁻¹ avidinated rabbit IgG was incubated with 2.5 ml of 20 μ g ml⁻¹ long arm photobiotin in PBS for 20 minutes. This solution was then split into two equal portions and half was exposed to light as it would be in the immobilisation procedure, whilst the other portion was kept in the dark. Both portions were then analysed using size exclusion chromatography using a 10 × 1 cm Sephadex G200 column, eluting using 10 mM phosphate buffer pH 7.4. 1.0 ml fractions were collected and their protein content was determined by measurement of absorbance at 280 nm. The resultant elution profiles are shown in Section 2.13.2.

2.3.3 Method for light activated patterning of a protein on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct

Avidin was immobilised on a SiO₂ wafer using the method described in Section 2.3.1 but replacing the BSA with avidin D. The avidin-modified wafer was incubated in 5.0 ml of 10 μ g ml⁻¹ long arm photobiotin in PBS for 20 minutes, this and all subsequent stages were performed under dark room conditions. The wafer was covered with 10 μ g ml⁻¹ rabbit 1gG in PBS and exposed to light as described in Section 2.3.1. The temperature of the wafer was monitored throughout the exposure period and did not increase by any more than 5°C. The wafer was covered with 10 mg ml⁻¹ casein in PBS and exposed to light for 15 minutes in the absence of a mask. The wafer was then incubated in 10 μ g ml⁻¹ TRITC labelled goat anti-rabbit IgG in PBS, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy. Results are presented in Section 2.13.3.

2.4 Method for Investigation into Achievable Resolution of Protein Patterning using Avidin-Photobiotin Technique.

The methodology of this Section of work was identical to that in Section 2.3.3 apart from the lithographic mask used, which consisted of series of lines (equal mark:space ratio) ranging from 1µm to 10µm in width. Results are presented in Section 2.14.

2.5 Investigation into Timescale of Avidin-Photobiotin Immobilisation Procedure. 2.5.1. Method for monitoring effect of exposure time on patterning process using fluorescence microscopy.

Avidin D and photobiotin were immobilised onto wafers of SiO₂ as described in Section 2.3.3. Wafers were then flood exposed to light from a mercury vapour lamp (irradiance ~ 9 mW cm⁻²) in the presence of 10 mg ml⁻¹ casein in PBS for varying periods of time. After washing in PBS all samples were exposed to light for 15 minutes through a photolithographic mask in the presence of 10 mg ml⁻¹ chicken IgG in PBS. Following washing in PBS, samples were incubated with TRITC labelled rabbit anti - chicken IgG for one hour, and examined using fluorescence microscopy.

2.5.2. Method for monitoring effect of exposure time on patterning process using colourimetric enzyme assay.

Avidin D and photobiotin were immobilised onto wafers of SiO_2 as described in Section 2.3.3. Wafers were then flood exposed to light from a mercury vapour lamp (irradiance ~ 9 mW cm⁻²) in the presence of 100 µg ml⁻¹ ALP in PBS for varying periods of time. Other "control" samples were exposed to the enzyme for the same periods of time but were not

exposed to light, in order to monitor any variation in non-specific binding as a function of exposure time. After washing in TBS, the ALP activity of all samples was assayed as described in Section 2.3.2.1. The SiO₂ wafers were weighed and the enzyme activity related per gram of SiO₂ was calculated in order to correct for any variation in wafer size. The results are presented in Section 2.15.

2.6 Methods for Light Activated Patterning of Multiple Proteins onto Silicon Dioxide Surfaces

2.6.1 Method for light activated patterning of two immunoglobulins on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct

Avidin and photobiotin were immobilised on a SiO₂ wafer using the silanisation process described in Section 2.3.1. The wafer was covered with 10 μ g ml⁻¹ rabbit IgG in PBS and exposed to light through a mask with 10 μ m interval equal mark:space ratio lines as described in Section 2.3.1. The wafer was then covered with 10 μ g ml⁻¹ rat IgG in PBS and exposed to light for 15 minutes through the same mask rotated through 90°. The wafer was covered with 10 mg ml⁻¹ casein in PBS and exposed to light for 15 minutes in the absence of a mask. The wafer was then incubated in 10 μ g ml⁻¹ TRITC labelled goat anti-rabbit IgG in PBS, for 60 minutes, washed and covered with 10 μ g ml⁻¹ fluorescein isothiocyanate (FITC) labelled rabbit anti-rat IgG, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy. A photomicrograph of this surface is shown in Section 2.16.1.

2.6.2 Method for patterning of five functional antibodies on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct

Avidin (Neutravidin[™] in place of Avidin D (note that the manufacturers of Neutravidin claim that it shows less non-specific binding than Avidin D, but Neutravidin was not available at the start of the study, hence the two type of avidin used)) and photobiotin were immobilised on a SiO₂ wafer using the silanisation process described in Section 2.3.1. The SiO₂ wafer was attached to a microscope so that it was in a fixed position below the microscope's objective lens with the top of the wafer level with the microscope stage. The five different antibodies were sequentially bound to the surface by selective light activation of distinct areas using a 100W high pressure mercury vapour lamp and a series of five photolithographic masks, each with a common registration mark (+) and a different number between 1 and 5 (50 μ m × 25 μ m in size) as a pattern. Polyclonal rabbit antibodies (10 μ g ml⁻¹), raised against chicken, guinea pig, goat, mouse and rat were patterned in each case realigning the patterned substrate by means of the registration mark and a CCD camera attached to the microscope (the wafer was thoroughly washed in PBS between each exposure). The surface was then flood - exposed with light in the presence of 10 mg m^{-1} casein in PBS, so immobilising the "blocking" protein at any previously unreacted photobiotin sites. After thorough washing with PBS, the wafer was incubated in a mixture of complementary antigens (chicken IgG, guinea pig IgG, goat IgG, mouse IgG and rat IgG, 100 μ g ml⁻¹ of each) in PBS at room temperature for one hour.

Following thorough washing with PBS, the wafer was incubated for one hour in 100 μ g ml⁻¹ rabbit anti - chicken IgG - TRITC in PBS and washed in PBS. Excess liquid was removed using a gentle stream of nitrogen, and the sample was examined using fluorescent

microscopy recording the image photographically (using a single lens reflex camera with automatic exposure times). This procedure was repeated sequentially, replacing the addition of rabbit anti - chicken IgG - TRITC with TRITC conjugates of anti - guinea pig IgG, rabbit anti - goat IgG, rabbit anti - mouse IgG or rabbit anti-rat IgG, on each occasion. Results are presented in Section 2.16.2.

2.7 Demonstration of Light Dependent Coupling of Proteins to Gold Surface

2.7.1 Method for demonstration of light dependent coupling of enzyme to gold surface

Avidin D was adsorbed on to two identical 2mm diameter gold electrodes (Bioanalytical systems (BAS)) by placing the electrodes in 5 ml of a 0.2mg ml⁻¹ solution of avidin D in PBS for one hour at room temperature. After extensive rinsing with PBS the electrodes were then incubated in 5 ml of 10 μ g ml⁻¹ solution of long arm photobiotin in PBS for 20 minutes under dark room conditions. After extensive rinsing with PBS electrodes were incubated with 50 μ l of identical solutions of glucose oxidase (GOD) (50U ml⁻¹) in PBS, one electrode was retained in dark room conditions whilst the other was exposed to light from a high pressure mercury vapour lamp for 15minutes. After extensively rinsing both electrodes with PBS under dark room conditions, 50 μ l of 10 mg ml⁻¹ BSA in PBS was added to each electrode and both were exposed to light from a high pressure mercury vapour lamp for 15 minutes.

An amperometric enzyme based assay was performed using the modified gold surface as a working electrode, with an Ag/AgCl electrode as the reference and a bare platinum flag as a counter electrode. Chronoamperometry was performed in a working solution containing

10 mM sodium phosphate, 140 mM NaCl, 25 mM KCl, pH 7.4, or with the same solution but with the addition of 100 mM glucose. Initially, the working electrode was poised at a potential of 0V for 300 seconds, after which the potential was "stepped" to 650mV for 120 seconds during which time the current was monitored. Resultant traces and currents obtained 30 seconds after the application of the 650 mV potential are presented in Section 2.17.1.

2.7.2 Method for Demonstration of light dependent coupling of antibody to gold surface

Avidin D and photobiotin were immobilised onto two gold electrodes as described in Section 2.7.1. After extensive rinsing with PBS each electrode was incubated with $50 \ \mu l \ of 100 \ \mu g \ ml^{-1}$ rabbit anti-rat IgG in PBS whilst being exposed to light from a high pressure mercury vapour lamp for 15 minutes.

Electrodes were incubated in either 5 ml of 10 μ g ml⁻¹ rat IgG in PBS for 60 minutes, whilst the other was placed in 5 ml of 10 μ g ml⁻¹ rabbit IgG in PBS for 60 minutes. Following this both electrodes were extensively washed with PBS. Background currents were obtained by performing an amperometric measurement using the modified gold surface as a working electrode, with a Ag/AgCl electrode as the reference and a bare platinum flag as a counter electrode. After thorough rinsing, the electrodes were placed in 15ml of 25 mM KCl, 10mM hydrogen peroxide, 1 mM ferrocenemonocarboxylic acid. The potential was stepped to +150 mV for 120 seconds, during which time the current was monitored. After extensive washing with PBS the electrodes were incubated in 5 ml of a solution of 20 μ g ml⁻¹ horseradish peroxidase labelled rabbit anti-rat IgG at room temperature for 60 minutes, and

following thorough rinsing, amperometric measurements were made as described above. Results are presented in Section 2.17.2.

2.8. Method for Light Dependent Coupling of a Protein to a Glass Surface

A clean glass slide was immersed for 30 seconds in a 1% solution of 1,3-trimethoxysilylpropylethylene diamine in 95% ethanol 5% distilled water, pH adjusted to 5.0 with glacial acetic acid. After removing the slide from this solution it was rinsed briefly in 95% ethanol 5% distilled water before being heated at 120°C for 30 minutes. The slide was immersed in 2% glutaraldehyde in PBS for 15 minutes, and then in a solution of 40mM sodium cyanoborohydride, 0.2mg ml⁻¹ avidin D in PBS for 30 minutes at room temperature. After extensive rinsing with PBS the wafer was incubated in 5 ml of 10µg ml⁻¹ long arm photobiotin in PBS for 20 minutes under dark room conditions. After extensive rinsing with PBS the wafer was covered with 10µg ml⁻¹ rat IgG in PBS and exposed to light from a high pressure mercury vapour lamp for 15 minutes through a patterned chrome mask. Following further rinsing with PBS the wafer was covered with 10mg ml⁻¹ BSA in PBS and the entire surface was exposed to light from the mercury vapour lamp for 15 minutes. The slide was incubated with 10µg ml⁻¹ TRITC labelled rabbit anti-rat IgG in PBS, for 60 minutes at room temperature. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using a fluorescent microscope. A photomicrograph of the surface is shown in Section 2.18.

2.9. Method for Light Dependent Coupling of a Protein to a Carbon Surface

The carboxyl groups on two freshly polished glassy carbon electrodes (GCEs) were activated by the mixed anhydride method,¹⁸⁸ which allows primary amines in the avidin molecule to bind directly to activated carboxyl groups on the GCE surface. This procedure involved incubation of the GCEs in 10% HNO₃ for 10 seconds at +2V vs. Ag/AgCl followed by incubation in dimethylformamide (DMF) containing 1% N-methylmorpholine and 1.5% isobutylchloroformiate for 60 minutes (no electrical potential applied). The electrodes were then incubated in 50 mM sodium phosphate buffer pH 7.0 containing 100µg ml⁻¹ avidin D for 2 hours. After extensive rinsing with PBS the electrodes were then incubated in 5 ml of a 10 µg ml⁻¹ solution of long arm photobiotin in PBS for 20 minutes under dark room conditions.

After extensive rinsing with PBS each electrode had 50 μ l of identical solutions of GOD in PBS aliquotted onto it. One electrode was subsequently retained in dark room conditions, whilst the other was exposed to light from a high pressure mercury vapour lamp for 15 minutes. After extensively rinsing both electrodes with PBS under dark room conditions, 50 μ l of a 10 mg ml⁻¹ solution of BSA in PBS was added to each electrode and both were exposed to light from a high pressure mercury vapour lamp for 15 minutes.

An amperometric enzyme based assay was performed using the modified carbon surface as a working electrode, with an Ag/AgCl electrode as the reference and a bare platinum flag as a counter electrode. Chronoamperometry was performed in a working solutions containing either 0 or 100 mM glucose solutions in 10 mM sodium phosphate, 140 mM NaCl, 25 mM KCl, pH 7.4. Initially, the working electrode was poised at a potential of 0V for 300 seconds, after which the potential was stepped to 650mV for 120 seconds during which time the current was monitored. Results are presented in Section 2.19

2.10 Methods for Investigation into Reducing Non-Specific Binding on Gold Surfaces. 2.10.1 Method for investigation into reducing non-specific binding on gold surfaces.

Gold was evaporated onto a SiO₂ wafer as described in Section 3.5.1. The gold coated wafer was cut into portions of uniform size (10×4 mm), so that results obtained were directly comparable. 3 forms of avidin were examined, two deglycosylated preparations (Avidin D and Neutravidin) and the microbial type (streptavidin).

2.10.1.1 Immobilisation of avidin on gold by adsorption

The different forms of avidin were adsorbed onto gold by incubation of 100µg ml⁻¹ avidin in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C, followed by extensive washing in PBS.

2.10.1.2 Immobilisation of avidin on gold using NAC and EDC

Gold surfaces were incubated in 2 mM NAC in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature, followed by 120 minutes incubation in 1% (w/v) EDC in 10 mM phosphate buffer (pH 7.0). The gold surfaces were then incubated in 100 μ g ml⁻¹ of the different forms of avidin in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C

2.10.1.3 Immobilisation of avidin on gold using DTSSP

Gold surfaces were incubated in 2 mM DTSSP in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature. The gold surfaces were then incubated in $100\mu g$ ml⁻¹ avidin in PBS for 16 hours at 4°C. As described in Section 2.1.8.2 the disulphide bond in DTSSP is thought to strongly chemisorb to the gold, and the NHS esters are available to react with primary amines in avidin to form a peptide bond.

2.10.1.4 Treatment with blocking agent and testing for non-specific binding.

Subsequent to the immobilisation of the avidin, the gold coated SiO₂ wafers were thoroughly washed in PBS and exposed to additional blocking proteins (BSA, casein and Superblock TM) by incubation in a 100 μ g ml⁻¹ solution in PBS for 24 hours at 4°C. The gold coated SiO₂ wafers were then thoroughly washed in TBS (10 mM Tris, 140 mM NaCl, pH 7.4) and incubated in 10 μ g ml⁻¹ rabbit IgG - ALP in TBS for 15 minutes at room temperature. The samples were washed extensively in TBS and ALP activity was determined colourimetrically using pNPP as the substrate as described in Section 2.3.2.1. The absorbance was measured at 404 nm, and results were normalised against the response obtained for bare gold, the higher the response, the greater the degree of non-specific binding (note that the results were corrected to take account of the relatively small amount of non-specific binding occurring on the reverse side of the SiO₂ wafer by running controls of SiO₂ wafer that did not have gold attached). The results of these experiments are summarised in Section 2.20.

2.11 Methods to Investigate Regeneration of Avidinated Surfaces

2.11.1 Methods to investigate regeneration of avidinated silicon dioxide surfaces

Rabbit IgG and BSA were patterned (4μ m equal mark:space ratio lines) onto SiO₂ surfaces as described in Section 2.3.3, and rabbit IgG was visualised by the addition of goat anti-rabbit IgG - TRITC conjugate. The surfaces were examined using fluorescence microscopy, and then underwent a number of treatments to ascertain if the avidinated surface could be regenerated.

The treatments were;

- a, Incubation in NaOH solution pH 12 for 16 hours.
- b, Incubation in HCl solution pH 1.5 for 16 hours.
- c, Incubation in 6M guanidine in HCl solution pH 1.5 for 16 hours.
- d, Incubation in 10M urea for 16 hours.
- e, Incubation in PBS for 16 hours (control).

The surfaces were subsequently washed in distilled water, dried under a gentle stream of nitrogen and examined using fluorescence microscopy. The treated SiO_2 was then re-incubated with photobiotin and was subjected to repatterning with rat IgG and BSA as using the mask with 4µm equal mark:space ratio lines rotated through 90°. The presence of rat IgG was visualised by the addition of rabbit anti-rat IgG - FITC conjugate, and the samples were examined using fluorescence microscopy. The results are summarised in Section 2.21.1.

2.11.2. Method to investigate regeneration of avidinated gold surfaces

Four gold electrodes were incubated in 2 mM NAC in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature, followed by 120 minutes incubation in 1% (w/v) EDC in 10 mM phosphate buffer (pH 7.0). The electrodes were then incubated in 100 μ g ml⁻¹ NeutravidinTM in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C. All subsequent stages of the immobilisation procedure were performed at room temperature. After washing in PBS, the electrodes were first incubated in 10 mg ml⁻¹ casein in PBS for 60 minutes and then in 10 μ g ml⁻¹ long arm photobiotin in PBS, for 20 minutes in the dark. All subsequent stages of the immobilisation procedure were performed in the dark. After extensive rinsing with PBS each electrode had 50 μ l of identical solutions of goat anti-rat IgG in PBS placed onto it and the electrodes were exposed to light from a high pressure mercury vapour lamp for 15 minutes.

Two electrodes (#1 and #3) were incubated in 5 ml of 10 μ g ml⁻¹ rat IgG in PBS for 60 minutes, whilst the the others (#2 and #4) were incubated in 5 ml of 10 μ g ml⁻¹ rabbit IgG in PBS for 60 minutes. Following this electrodes were extensively washed with PBS. Background currents were obtained by performing an amperometric measurement using the modified gold surface as a working electrode, with a Ag/AgCl electrode as the reference and a bare platinum flag as a counter electrode. After thorough rinsing, the electrodes were placed in 15ml of 25 mM KCl, 10mM hydrogen peroxide, 1 mM ferrocenemonocarboxylic acid and a potential 150 mV was applied for 120 seconds during which time the current was monitored. After extensive washing with PBS the electrodes were incubated in 5 ml of a solution of 20 μ g ml⁻¹ HRP labelled goat anti-rat IgG at room temperature for 60 minutes, and following thorough rinsing, amperometric measurements were made as described

above. Currents obtained 30 seconds after the application of the 150 mV potential are shown in Section 2.21.2 (Table 2.6). Two (#1 and #2) electrodes were then incubated for 16 hours in 6M guanidine pH 1.5, whilst two (#3 and #4) were incubated for 16 hours in PBS (controls). After washing in PBS, the electrodes were incubated in 10 μ g ml⁻¹ long arm photobiotin in PBS, for 20 minutes in the dark. After extensive rinsing with PBS each electrode had 50 μ l of identical solutions of goat anti-rabbit IgG in PBS placed onto it and the electrodes were exposed to light from a high pressure mercury vapour lamp for 15 minutes. The electrodes were then incubated with rat IgG or rabbit IgG, and HRP labelled goat anti-rabbit IgG and amperometric measurements were taken as previously described, the resultant currents are presented in Section 2.21.2.

2.12 Method for Characterisation of Biomolecular Patterns using Atomic Force Microscopy and Fluorescence Microscopy.

Neutravidin and photobiotin were immobilised on a SiO₂ wafer using the silanisation method described in Section 2.3.1. The wafer was covered with 10 μ g ml⁻¹ goat anti-rabbit IgG in PBS and exposed to light, as described in Section 2.3.1, for 15 minutes through a mask with 3 μ m interval equal mark:space ratio lines. The wafer was covered with 10 mg ml⁻¹ casein in PBS and exposed to light for 15 minutes in the absence of a mask. The wafer was then incubated in 10 μ g ml⁻¹ TRITC labelled rabbit anti-rat IgG in PBS, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy. A photomicrograph of this surface is shown in Section 2.22.

The pattern of antibodies was imaged using AFM, images were obtained using a Burleigh Personal AFM (Burleigh, UK) with Si tips (spring constant 0.045 nM m⁻¹, with a tip radius of 10 nm and an aspect ratio of approximately 4:1, as quoted by Burleigh and verified using SEM (Hitachi S-900, Hitachi, Cambridge)). Imaging was carried out in air at ambient temperature and humidity with an imaging force maintained at approximately 5 nN. The resultant images are shown in Section 2.22.

2.13 Results and Discussion for Patterning of Proteins using Avidin and Photobiotin.

2.13.1 Results and discussion for patterning method using an avidin-protein conjugate by depositing a protein layer on the surface, adding photobiotin followed by the addition of the avidinated protein

The photomicrographs in Figures 2.14 and 2.16 (along with plots of relative fluorescent intensity in Figures 2.15 and 2.17) demonstrate that patterning of proteins was possible using this approach. However, the immobilisation efficiency (as evident by the intensity of the fluorescence and the lack of contrast between areas corresponding to mark and space on the photolithographic mask) was very low compared with other approaches (see Section 2.13.3). This may be due to steric hindrance of one or more of the coupling steps (i.e. the majority of photobiotin is bound to the immobilised protein layer, in such a way that the biotinyl groups are unavailable for subsequent binding to avidin). The data from the addition of the fluorescently labelled avidin suggested that the poor efficiency was due to the stage where avidin bound to the immobilised photobiotin, as again there was relatively little differentiation in fluorescent intensity in areas that had been exposed to light, and those that had not. A further disadvantage of this technique was that for a species to be immobilised it must have undergone prior avidination or biotinylation.



10 µm

Figure 2.14 Photomicrograph of silicon dioxide patterned using a BSA / photobiotin / avidin / biotin-immunoglobulin construct visualised by a TRITC labelled anti-immunoglobulin antibody. BSA was immobilised on a SiO₂ wafer as described in Section 2.3.1. The BSA-modified wafer was incubated in a solution of long arm photobiotin in PBS whilst exposed to light as described in Section 2.3.1. The wafer was then incubated in avidin in PBS and after washing in PBS, the wafer was incubated with biotinylated rabbit IgG in PBS. The wafer was then incubated in TRITC labelled goat anti-rabbit IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.



Figure 2.15 Plot of relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.14. The relative fluorescent intensity was determined by taking the brightest point (i.e. greatest fluorescence) on a line across the pattern as 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). The fluorescent intensity of each point on the line was determined by interpolation between these 2 extremes.



10 µm

Figure 2.16. Photomicrograph of silicon dioxide patterned using a BSA / photobiotin / TRITC labelled avidin construct. BSA and photobiotin were immobilised as described in Section 2.3.1.1. The wafer was then incubated in TRITC labelled avidin in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.



Figure 2.17 Plot of relative fluorescent intensity against distance for cross sectional profile through a section of the pattern in Figure 2.16. The relative fluorescent intensity was determined by taking the brightest point (i.e. greatest fluorescence) on a line across the pattern as 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). The fluorescent intensity of each point on the line was determined by interpolation between these 2 extremes.

2.13.2 Results and Discussion for patterning a protein by depositing a protein layer on the surface, and add avidinated protein with photobiotin attached

The photomicrograph in Figures 2.18 (along with plot of relative fluorescent intensity in Figures 2.19) demonstrate that patterning of proteins was possible using this approach. The immobilisation efficiency (as evident by the intensity of the fluorescence apparent when the wafer was viewed by eye through a fluorescence microscope, and the much longer exposure time necessary to obtain a photomicrograph) was low compared with other approaches (see 2.13.3). This appeared to be at least partially due to considerable crosslinking occurring between the avidinated species-photobiotin complexes in solution. Due to the generally faster reaction kinetics in solutions, it is likely that these complexes bound preferentially to each other rather than to the surface protein layer. Figure 2.17 shows the elution profile from size-exclusion chromatography of the material pre and post exposure to light. The increased A_{280} in the earlier fractions demonstrates that the post-exposure solution contained very high molecular weight material which was not present in the pre-exposure solution, this indicates that reactions between the species in solution had occurred upon exposure to light. An additional disadvantage of this immobilisation method (as described in Section 2.13.1), is that it would only be possible to immobilise species that had undergone avidination prior to the start of the immobilisation process.



10 µm

Figure 2.18 Photomicrograph of silicon dioxide patterned using a BSA / photobiotin-avidin - immunoglobulin construct. BSA was immobilised as described in scetion 2.3.2. Avidinated rabbit IgG was pre-incubated with long arm photobiotin in PBS, and was then placed on the protein coated SiO₂ wafer and exposed to light through a mask with 10µm interval equal mark:space ratio lines. The wafer was then incubated in TRITC labelled goat anti-rabbit IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.



Figure 2.19 Plot of relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.18. The relative fluorescent intensity was determined by taking the brightest point (i.e. greatest fluorescence) on a line across the pattern as 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). The fluorescent intensity of each point on the line was determined by interpolation between these 2 extremes.



Figure 2.20 Normalised A_{280} from fractions from chromatographic separation of a photobiotin / avidin / protein solution, pre and post exposure to light. 2.5 ml of 20 µg ml⁻¹ avidinated rabbit IgG was incubated with 2.5 ml of 20 µg ml⁻¹ long arm photobiotin in PBS for 20 minutes. This solution was then split into two equal portions, and half was exposed to light for 15 minutes from a 100 W high pressure mercury vapour lamp sited 185 mm from the solution, whilst the other portion was kept in the dark. Both portions were then analysed using size exclusion chomatography using a 10×1 cm Sephadex G200 column, eluting using 10 mM phosphate buffer pH 7.4. 1 ml fractions were collected and their protein content was determined by measurement of absorbance at 280 nm.

2.13.3 Results and Discussion for light activated patterning of a protein on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct

The photomicrograph in Figures 2.21 (along with plot of relative fluorescent intensity in Figures 2.22) demonstrate that patterning of proteins was effective using this approach. This immobilisation method appeared to give much better results than the alternative procedures. The immobilisation efficiency (as evident by the intensity of the fluorescence apparent when the wafer was viewed by eye through a fluorescence microscope, and the much shorter exposure time necessary to obtain a photomicrograph) was high compared with other approaches (see sections 2.13.1 and 2.13.2). The photomicrograph of the surface shown in Figure 2.21, and the plot (Figure 2.22) of relative fluorescent intensity against distance for cross sectional profile through the pattern showed that this immobilisation technique was capable of high resolution (<10 μ m) patterning of proteins onto a surface with low non-specific binding.



10 µm

Figure 2.21 Photomicrograph of silicon dioxide patterned with TRITC labelled IgG, using an avidin-photobiotin IgG construct as described in Section 2.3.3. The wafer was then incubated in TRITC labelled goat anti-rabbit IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.



Figure 2.22 Plot of relative fluorescent intensity against distance for cross sectional profile through the pattern in Figure 2.21. The relative fluorescent intensity was determined by taking the brightest point (i.e. greatest fluorescence) on a line across the pattern as 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). The fluorescent intensity of each point on the line was determined by interpolation between these 2 extremes.

2.14 Results and Discussion for Investigation into Achievable Resolution of Protein Patterning using Avidin-Photobiotin Technique.

A photomicrograph of a surface patterned with protein using a mask with lines 1.5µm in width is shown in Figure 2.23, demonstrating that patterns can be formed with features of this size. The plots shown in Figure 2.24 show that baseline resolution of fluoresence (which is related to quantity of immobilised protein) is achieved for equal mark:space ratio lines as small as 1.5µm in width.

Although a pattern of fluorescent intensity relating to that on the mask is evident, baseline resolution is not achieved for lines 1 μ m in width. The most likely explanation of the lack of resolution with 1 μ m lines was that the mask was not in intimate enough contact to the substrate being patterned to overcome dispersion of the light through refraction etc. Attempts to achieve sub-micron resolution were not successful. Briefly, these experiments involved applying pressure to the mask to bring it in closer contact with the substrate to be patterned, however it was observed that by doing so the reactant solutions were forced out of the gap between substrate and mask, and the immobilisation process did not work (even

when concentrations of reactants in the solutions were greatly increased).


н. 1 µm

Figure 2.23 Photomicrograph of SiO₂ patterned with TRITC labelled IgG, using an avidin-photobiotin IgG construct. The photograph shows that 1.5 µm lines of immobilised protein can be patterned on a SiO₂ surface. Avidin was immobilised on a SiO₂ wafer as described in Section 2.3.3. The avidin-modified wafer was incubated with photobiotin in PBS, and was then covered with rabbit IgG in PBS and exposed to light through a mask with 1.5µm interval equal mark:space ratio lines. The wafer was covered with casein in PBS and exposed to light in the absence of a mask. The wafer was then incubated in TRITC labelled goat anti-rabbit IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.

which pleasibilities any large print as photointerpreside analy with equal touch space track lines of deferences of The patiented protects since vanished by the addition of the occurdly ideallar and resolutes and commond by measurements of the tracking information (pressing across the patient. The resolute flatter case of the pair of the resolution of the tracking information (pressing across the patient. The resolute flatter case of the pair of the resolution to the brightest point (i.e. gravitest the patient. The resolute flatter case for the resolution of the tracking test point (i.e. gravitest the patient. The resolute flatter case for the resolution of the tracking test point (i.e. gravitest the patient. The functions of the track of the four error of the resolution of the apoint met would show no functions are the function of the tracking of each point on the lasts was demonstrated by the point of the resonance (present).



Figure 2.24 Plot of relative fluorescent intensity against distance for cross sectional profile of different width lines formed using the patterning of proteins as previously described. Proteins were patterned using avidin-photobiotin employing a photolithographic mask with equal mark:space ratio lines of different widths. The patterned proteins were visualised by the addition of fluorescently labelled antibodies, and resolution was examined by measurement of the relative fluorescent intensity across the pattern. The relative fluorescent intensity was determined relative to the brightest point (i.e. greatest fluorescence) assigned a value 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence The fluorescent intensity of each point on the lines was determined by interpolation between these 2 extremes.

2.15 Results and Discussion for Investigation into Timescale of Avidin-Photobiotin Immobilisation Procedure.

2.15.1. Results and discussion for monitoring effect of exposure time on patterning process using fluorescence microscopy.

Upon fluorescence microscopy examination of surfaces treated as described in Section 2.5.1, patterned IgG was clearly visible when the substrate (with immobilised avidin and photobiotin) had been exposed to light in the presence of casein for up to 250 seconds prior to exposing the substrate to light in the presence of the IgG. When the substrate had been pre-exposed to casein and light for 500 seconds, only a very faint pattern corresponding to IgG could be observed, and no pattern could be discerned for pre-exposure times longer than 500 seconds. This data indicates that under the conditions described in Section 2.5.1, some of the photobiotin could still be activated following pre-exposure to light for up to 500 seconds.

2.15.2. Results and discussion for monitoring effect of exposure time on patterning process using colourimetric enzyme assay.

The results shown in Table 2.1 and Figure 2.25 indicate that under these experimental conditions (described in Section 2.5.2), the photobiotin mediated immobilisation process proceeds rapidly, and that it is close to completion (>90%) after 500 seconds. No further immobilisation occurs after 750 seconds. These results also demonstrate that the small amount of non-specific binding present also occurs rapidly, with approximately 60% of the material that is non-specifically bound being so in the first minute of exposure.

| Time exposed to ALP (seconds) | ALP activity sample exposed to light (absorbance units / g SiO ₂) | ALP activity sample kept in dark (absorbance units / g SiO ₂) |
|----------------------------------|-------------------------------------------------------------------------------------|---------------------------------------------------------------------------------|
| 0 | 0 | 0 |
| 1 | 1.1 | 0.8 |
| 10 | 8.2 | 1.2 |
| 25 | 17.1 | 1.5 |
| 50 | 31.4 | 1.8 |
| 100 | 51.6 | 2.2 |
| 250 | 66.9 | 2.4 |
| 500 | 500 71.4 2.8 | |
| 750 | 75.3 | 3.0 |
| 1,000 | 74.9 | 2.9 |

<u>Table 2.1 Enzyme activity versus time of exposure of avidin - photobiotin coated SiO₂ to an ALP solution in the presence or absence of light</u>. Avidin D and photobiotin were immobilised onto wafers of SiO₂ as described previously. Wafers were then flood exposed to light in the presence of 100 μ g ml⁻¹ ALP in PBS for varying periods of time. Other "control" samples were exposed to the enzyme for the same periods of time but were not exposed to light. After washing in TBS, the ALP activity of all samples was assayed colourimetrically using pNPP as the substrate. Enzyme activity related per gram of SiO₂ was calculated.



Figure 2.25 Plot of enzyme activity versus time of exposure of avidin - photobiotin coated SiO₂ to an ALP solution in the presence or absence of light. Avidin D and photobiotin were immobilised onto wafers of SiO₂ as described previously. Wafers were then flood exposed to light in the presence of 100 μ g ml⁻¹ ALP in PBS for varying periods of time. Other "control" samples were exposed to the enzyme for the same periods of time but were not exposed to light. After washing in TBS, the ALP activity of all samples was assayed colourimetrically using pNPP as the substrate. Enzyme activity related per gram of SiO₂ was calculated.

2.16 Results and Discussion for Light Activated Patterning of Multiple Proteins onto Silicon Dioxide surfaces

2.16.1 Results and discussion for light activated patterning of two immunoglobulins on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct Figure 2.26 demonstrates that the avidin / photobiotin technique can be used to pattern multiple proteins site specifically on to a silicon dioxide surface with minimal non-specific binding. It should be noted that in Figure 2.26 the TRITC lines are less red than those in for example Figure 2.24 owing to the fact that a double exposure (TRITC & FITC filters) was needed to obtain the former. This colour change was not due to non-specific binding of the FITC probe, as it occurs in the absence of the FITC label.



10 µm

<u>Figure 2.26 Photomicrograph of Silicon dioxide patterned with two fluorescently labelled antibodies.</u> <u>Avidin and photobiotin were immobilised as described in Section 2.3.3.</u> The wafer was then sequentially incubated with rabbit IgG and rat IgG whilst exposed to light through a mask with 10µm interval equal mark:space ratio lines. In between the 2 incubations the mask was rotated through 90°. The wafer was then blooked with casein and incubated in TRITC labelled goat anti-rabbit IgG and FITC labelled rabbit anti-rat IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.

2.16.2 Results and discussion for light activated patterning of five antibodies on a silicon dioxide surface using an avidin / photobiotin / immunoglobulin construct

Figure 2.27 shows a series of photomicrographs demonstrating the sequential patterning of different antibodies on a silicon dioxide surface. These photographs demonstrate the micrometre scale patterning of 5 functional antibodies (along with a sixth protein, casein) on to the surface (within an area of $200 \times 50 \ \mu$ m). This work also demonstrates the construction of a qualitative multianalyte immunosensor, which is capable of specifically detecting 5 different antigens.

It should be noted that individual numbers are less intense in the later photomicrographs, as the overall amount of fluorescent light has increased and therefore the exposure time has decreased. The increased non-specific binding present in the later photomicrographs is due to the repeated drying of the substrate followed by addition of fluorescently labelled proteins. This was shown by repeating the experiment but not drying the sample in-between incubations with anti-species IgG - TRITC conjugates, where it was found that the intensity of the background was less than that for a sample that had been dried (note that as the sample had not been dried, it was not possible to take a series of sequential photomicrographs of pattern development).



Figure 2.27 A series of fluorescence photomicrographs demonstrating the sequential patterning of different antibodies on a silicon dioxide surface. Polyclonal rabbit antibodies raised against chicken, guinea pig, goat, mouse and rat were patterned using the avidin-photobiotin immobilisation method, in each case realigning the patterned substrate by means of a registration mark and a CCD camera attached to the microscope. See Section 2.6.2 for experimental procedures.

2.17 Results and discussion for Demonstration of Light Dependent Coupling of Proteins to a Gold surface

2.17.1. Results and discussion for demonstration of light dependent coupling of an enzyme to a gold surface

The results shown in Table 2.2 and Figure 2.28 demonstrate that an enzyme (in this case, GOD) can be patterned on to a gold surface using the avidin photobiotin method, whilst retaining its biological activity. Electrodes that had not been exposed to light in the presence of GOD showed negligible non-specific binding of the enzyme, whilst electrodes that had been incubated with GOD in the presence of light showed a significant response to glucose.

| | Current 30 seconds after application of 650 mV potenti | | | |
|------------------------------|--------------------------------------------------------|----------------|--|--|
| | μA (± 1SD), n=5 | | | |
| Glucose Concentration | oncentration Electrode exposed to light Electro | | | |
| (mM) | | light | | |
| 0 | 0.086 (±0.009) | 0.089 (±0.010) | | |
| 100 | 0.358 (±0.027) | 0.084 (±0.006) | | |

Table 2.2 Currents obtained from avidin - photobiotin coated gold electrodes which have been incubated with GOD in the presence or absence of light. Experimental details are in the text (Section 2.7)



Figure 2.28 Chronoamperometric responses to glucose of avidin - photobiotin coated gold electrodes which have been incubated with GOD in the presence or absence of light. Experimental details are in the text (Section 2.7)

2.17.2 Results and discussion for demonstration of light dependent coupling of an antibody to a gold surface

Data demonstrated that antibody molecules could successfully be immobilised on to gold surfaces using the avidin-photobiotin technique, whilst retaining their functional activity. The background currents (± 1 SD, n=5) for electrodes 1 and 2 (which both had rabbit anti-rat IgG immobilised on them) were $0.079\mu A \pm 0.004$ and $0.083\mu A \pm 0.005$ respectively. The current for electrode 1 which was subsequently incubated with rabbit IgG and HRP labelled anti rat IgG (± 1 SD, n=5) was $0.084\mu A \pm 0.005$ whilst the current for electrode 2 which was incubated with rat IgG and HRP labelled anti rat IgG (± 1 SD, n=5) was $0.243\mu A \pm 0.021$.

The electrode (with immobilised anti rat IgG) that had been incubated with rabbit IgG showed negligible activity as demonstrated by the lack of response using the HRP-anti-rat IgG, whilst electrode that had been incubated with rat IgG showed a significant response. This work shows the application of the avidin - photobiotin immobilisation technique to the construction of a qualitative electrochemical immunosensor for the detection of rat IgG. This work was further developed later, as a quantitative immunosensor for FSH and LH (see Chapter 3).

2.18. Results and Discussion for Light Dependent Coupling of a Protein to a Glass Surface

For the patterning of proteins on a glass substrate, areas of fluorescence were observed which matched the mask that had been used (Figure 2.29), however this fluorescent image was not as well defined as when using the immobilisation procedure on SiO₂ surfaces, and the same degree of resolution could not be obtained. Individual lines could not be observed when using a photolithographic mask with equal mark:space ratio lines 10μ m in width. A pattern of well resolved lines was clearly visible when using this mask and immobilisation procedure on SiO₂ substrates (Figure 2.21). The difference in resolution achieved using SiO₂ and glass substrates may be due to reflection of the incident light within the glass slide causing activation of the photobiotin in areas normally masked from light.



Figure 2.29 Photomicrograph of glass patterned with a fluorescently labelled antibody. Avidin was immobilised as decribed in Section 2.8. The avidin-modified slide was incubated with photobiotin in PBS, and was then covered with rat IgG in PBS and exposed to light through a mask with a triangular pattern. The wafer was covered with casein in PBS and exposed to light in the absence of a mask. The wafer was then incubated in TRITC labelled goat anti-rat IgG in PBS, and after washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.

2.19. Results and Discussion for Light Dependent Coupling of a Protein to a Carbon Surface

The results shown in Table 2.3 demonstrate that an enzyme (in this case, GOD) can be patterned on to a glassy carbon surface using the avidin photobiotin method, whilst retaining its biological activity. Electrodes that had not been exposed to light in the presence of GOD showed negligible non-specific binding of the enzyme, whilst electrodes that had been incubated with GOD in the presence of light showed a significant response to glucose. The currents for both of the electrodes in the 0 mM glucose, and that in 100 mM glucose for the electrode that had been incubated with GOD in the presence with GOD in the dark were very similar, whilst the current for 100 mM glucose with the electrode that had been incubated with GOD in the light was significantly increased.

| | Current 30 seconds afte potential μA | Current 30 seconds after application of 650 mV potential µA (± 1sd), n=5 | | |
|-------------------------------|-----------------------------------------|-----------------------------------------------------------------------------|--|--|
| Glucose Concentration (mM) | Electrode exposed to light | Electrode kept in dark | | |
| 0 | 0.092 (± 0.006) | 0.102 (± 0.011) | | |
| 100 | 0.275 (± 0.0232) | 0.094 (± 0.010) | | |

Table 2.3 Currents obtained from avidin - photobiotin coated glassy carbon electrodes which have been incubated with GOD in the presence or absence of light 30 seconds after application of 650 mV potential in glucose solutions. Experimental details are provided in the text (Section 2.9)

2.20 Results and Discussion for Investigation into Reducing Non-Specific Binding on Gold Surfaces.

It can be seen from Table 2.4 that the results obtained for Neutravidin and streptavidin are very similar, and non-specific binding tends to be lower with these preparations than with avidin D. The method of immobilisation of the avidin has a significant effect on the non-specific binding, with thiol mediated linkage (NAC & EDC, DTSSP) providing better results than that involving simple adsorption. A significant reduction in non-specific binding is seen with the use of all of the blocking agents, but the reduction is greater for BSA and casein than for Superblock. A low degree of non-specific binding is obtained when immobilising either Neutravidin or streptavidin to gold via a thiol, and using either BSA or casein as a subsequent blocking reagent

As has already been discussed, native avidin tends to show a high degree of non-specific binding of substances which is usually attributed to two inherent characteristics of the molecule, namely the high pI and the presence of carbohydrate residues.^{185,191} Streptavidin which has a lower carbohydrate content, and deglycosylated avidin both exhibit reduced non-specific binding as compared to intact avidin.^{185,191} The high isoelectric point (pI 10.0 - 10.5) of avidin, as well as the presence of carbohydrate (which accounts for about 10% of avidin's total molecular weight) on the protein contribute to non-specific binding in applications involving avidin. Due to its more favourable isoelectric point and lack of carbohydrate, streptavidin has historically been the accepted substitute for applications where the non-specific binding characteristics of avidin are unacceptable. More recently, modified forms of avidin such as Neutravidin have become available, with pI values closer to 7.0 and reduced carbohydrate contents leading to improved non-specific binding characteristics.^{192,193}

| | Normalised NSB % for blocking agent (± 1 sd), n=3 | | | |
|------------------------------------------|---------------------------------------------------|----------|----------|-------------|
| Preparation of gold surface | No blocking reagent | BSA | Casein | Superblock™ |
| Bare Gold | 100 (±3.7) | 5 (±0.3) | 6 (±0.4) | 23 (±2.7) |
| Gold + adsorbed avidin D | 17 (±1.0) | 3 (±0.3) | 4 (±0.3) | 5 (±0.4) |
| Gold + adsorbed Streptavidin | 14 (±0.9) | 4 (±0.4) | 5 (±0.3) | 7 (±0.5) |
| Gold + adsorbed Neutravidin | 15 (±0.7) | 3 (±0.3) | 6 (±0.4) | 8 (±0.6) |
| Au + avidin D linked by NAC & EDC | 15 (±0.8) | 6 (±0.3) | 7 (±0.5) | 9 (±0.5) |
| Au + Streptavidin linked by NAC & EDC | 8 (±0.3) | 3 (±0.3) | 3 (±0.2) | 5 (±0.3) |
| Au + Neutravidin linked by NAC & EDC | 10 (± 0.4) | 2 (±0.2) | 3 (±0.3) | 5 (±0.4) |
| Au + avidin D linked by DTSSP | 13 (±1.0) | 5 (±0.3) | 4 (±0.2) | 4 (±0.3) |
| Au + Streptavidin linked by DTSSP | 9 (±0.7) | 1 (±0.0) | 3 (±0.2) | 3 (±0.3) |
| Au + Neutravidin linked by DTSSP | 7 (±0.5) | 1 (±0.1) | 2 (±0.2) | 4 (±0.4) |

Table 2.4 Non-specific binding to treated gold electrodes after 15 minutes exposure to IgG - ALP conjugate. Different avidin preparations and blocking agents were immobilised as summarised in the table. The gold electrodes were then washed in TBS and incubated in 10 μ g ml⁻¹ rabbit IgG - ALP in TBS for 15 minutes at room temperature. The samples were washed extensively in TBS and ALP activity was determined colourimetrically using pNPP as the substrate.

2.21 Results and Discussion to Investigate Regeneration of Avidinated Surfaces

2.21.1 Results and discussion to investigate regeneration of avidinated silicon dioxide surfaces

As can be seen from the results in Figure 2.30 and Table 2.5, the patterned protein is resistant to treatment with HCl solution (pH 1.5) and 10 M urea solution (i.e. the pattern is still visible after treatment with these agents), although it should be noted that treatment with these solutions may denature the protein and render it non-functional. No change in the patterned protein is observed for the control treatment (incubation with PBS). Patterned protein is not resistant to treatment (i.e. the pattern is no longer visible after treatment) with NaOH solution (pH 12), but the inability to repattern the surface would indicate that a specific cleavage of the avidin photobiotin bond is not responsible. The high degree of non-specific binding observed when repatterning is attempted, suggests that this protocol either removes avidin and blocking proteins from the surface, or that it denatures them such that they no longer form an effective barrier to non-specific binding. Patterned protein is not resistant to treatment with 6M guanidine at low pH, and the fact that it is possible to repattern the surface demonstrates that it is the avidin photobiotin bond that is being broken. Treatment with guanidine has previously been shown to be able to break the avidin biotin bond.²¹⁸ The ability to break the avidin - photobiotin bond with the use of guanidine, means that it is possible to regenerate the avidinated surface and to repattern using photobiotin, which would be advantageous if one was using expensive sensor elements that one wished to reuse.



2.21 Results and Discussion to Investigate Regeneration of Avidinated Surfaces

Figure 2.30 Fluorescent photomicrographs of attempts to regenerate avidinated SiO₂ surfaces. Patterns of rabbit IgG were generated on the surface using the avidin-photobiotin technique (detailed in Section2.11.1) The wafer was incubated in TRITC labelled goat anti-rabbit IgG in PBS, and examined using fluorescent microscopy. The surface then underwent a number of treatments; a) Incubation in NaOH solution pH 12. b) Incubation in HCl solution pH 1.5. c) Incubation in 6M guanidine in HCl solution pH 1.5. d) Incubation in 10M urea. e) Incubation in PBS (all for 16 hours). The surfaces were washed in water, dried under nitrogen and examined using fluorescence microscopy. The treated SiO₂ was re-incubated with photobiotin and subjected to repatterning with rat IgG and BSA as described previously. The presence of rat IgG was visualised by the addition of rabbit anti-rat IgG - FITC conjugate, and the samples were examined using fluorescence microscopy.

| Treatment | Description of surfaces seen by fluorescence microscopy | | | |
|----------------------------------------------|---------------------------------------------------------|--------------------|----------------------------------------------------------------------|--|
| (surface incubated in solution for 16 hours) | Pre Treatment | Post Treatment | Post repatterning | |
| NaOH pH 12 | Rabbit IgG pattern | No pattern | No pattern. Bright background indicating high degree of NSB | |
| HCl pH 1.5 | Rabbit IgG pattern | Rabbit IgG pattern | Rabbit IgG pattern | |
| 6M guanidine, pH 1.5 | Rabbit IgG pattern | No pattern | Rat IgG pattern. Indicates regeneration of avidinated surface. | |
| 10M urea | Rabbit IgG pattern | Rabbit IgG pattern | Rabbit IgG pattern | |
| PBS | Rabbit IgG pattern | Rabbit IgG pattern | Rabbit IgG pattern | |

Table 2.5 Summary of results from procedures to attempt regeneration of avidinated surfaces. Patterns of rabbit IgG were generated on the surface using the avidin-photobiotin technique (detailed in Section 2.11.1) The wafer was incubated in TRITC labelled goat anti-rabbit IgG in PBS, and examined using fluorescent microscopy. The surface then underwent a number of treatments summarised in the table. The surfaces were washed in water, dried under nitrogen and examined using fluorescence microscopy. The treated SiO₂ was re-incubated with photobiotin and subjected to repatterning with rat IgG and BSA as described in Section 2.11.1. The presence of rat IgG was visualised by the addition of rabbit anti-rat IgG - FITC conjugate, and the samples were examined using fluorescence microscopy.

2.21.2 Results and discussion to investigate regeneration of avidinated gold surfaces

The data shown in Table 2.6 demonstrates that guanidine can be used to repattern proteins immobilised on gold surfaces using the avidin-photobiotin technique. The selectivity of the guanidine treated electrodes has changed, whilst that of the controls that were incubated in PBS has not.

| Electrode Treatment | Current (µA) 30 seconds after the application of 150 mV potential, for electrodes exposed to different antigens | | | |
|--------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|--------------------------------------|--------------------------------|--------------------------------------|
| | Electrode 1 (Rat IgG as Ag) | Electrode 2 (Rabbit IgG as Ag) | Electrode 3 (Rat IgG as Ag) | Electrode 4 (Rabbit IgG as Ag) |
| None (Background current) | 0.08 | 0.09 | 0.08 | 0.08 |
| Incubation with antigen and HRP - anti rat IgG | 0.31 | 0.10 | 0.31 | 0.09 |
| Treatment with guanidine and repatterning | 0.09 | 0.08 | - | - |
| Treatment with PBS and repatterning (control) | - | - | 0.30 | 0.09 |
| Treatment with guanidine, repatterning, incubation with antigen and HRP - anti rabbit IgG | 0.09 | 0.35 | - | - |
| Treatment with PBS, repatterning, incubation with antigen and HRP - anti rabbit IgG (control) | - | - | 0.32 | 0.11 |

Table 2.6 Currents obtained from gold electrodes pre and post repatterning antibodies using guanidine. Experimental details are provided in the text (Section 2.11). This data indicates that avidin -photobiotin modified gold surfaces can be regenerated using guanidine, thus allowing repatterning to take place.

2.22 Results and Discussion for Characterisation of Biomolecular Patterns using Atomic Force Microscopy and Fluorescence Microscopy.

Figures 2.33 and 2.35 demonstrate that protein patterns formed by use of the avidin-photobiotin immobilisation technique can be imaged using atomic-force microscopy. In Figure 2.34, the difference in the height between a peak (consisting of two IgGs assembled onto an avidin layer) and a trough (consisting of casein on avidin) can be estimated as approximately 7nm, which is less than would be expected on the basis of crystallographic data.²¹⁹ This may be explained by the random orientation the molecules are likely to take, alternatively the relatively large tip force necessary to obtain these images which is likely to distort the molecules significantly during imaging.^{216,217} It should be noted that in Figure 2.32 (a cross-sectional profile through the fluorescent photomicrograph in Figure 2.31) no significance can be attributed to the arbitrary units for relative fluorescent intensity with respect to the height of the pattern in the z direction.





<u>Figure 2.31, 2 - dimensional fluorescence micrograph of TRITC labelled IgG bound by a patterned</u> <u>complementary antibody.</u> The patterning procedure is detailed in the text (Section 2.12). Following the protein immobilisation procedure, the wafer was incubated in TRITC labelled rabbit anti-rat IgG. After washing the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.



Figure 2.32 Cross sectional profile through the pattern between points A and B marked in Figure 2.31. Patterned proteins were visualised by the addition of fluorescently labelled antibodies, and resolution was examined by measurement of the relative fluorescent intensity across the pattern. The relative fluorescent intensity was determined by taking the brightest point (i.e. greatest fluorescence) on a line across the pattern as 100. A fluorescent intensity of 0 was set as a point that would show no fluorescence (i.e. totally black). The fluorescent intensity of each point on the lines was determined by interpolation between these 2 extremes.





Figure 2.33. 2 - dimensional AFM image of the same pattern as in Figure 2.31. The patterning procedure is detailed in the text (Section 2.12). The pattern of antibodies was imaged using AFM, images were obtained using a Burleigh Personal AFM with Si tips (spring constant 0.045 nM m⁻¹, with a tip radius of 10 nm and an aspect ratio of approximately 4:1, Imaging was carried out in air at ambient temperature and humidity with an imaging force maintained at approximately 5 nN.







5 µm

Figure 2.35 3-dimensional reconstruction from AFM data of an area of a protein patterned surface. The patterning procedure is detailed in the text (Section 2.12). The pattern of antibodies was imaged using AFM, as detailed in the text (Section 2.12).

CHAPTER 3

Question your desires,

Know of your youth,

Examine well your blood

William Shakespeare

A Midsummer Nights Dream

3 Simultaneous Determination of FSH and LH using a Multianalyte Immunosensor

The aim of the work described in this chapter of the thesis was to demonstrate the application of avidin-photobiotin technology by constructing a multianalyte immunosensor for the simultaneous assay for follicle stimulating hormone (FSH) and luteinising hormone (LH). In this respect the measurement of FSH and LH was an attractive model system, as in clinical analysis, the determination of these hormones is usually requested simultaneously. Additionally, both FSH and LH are normally assayed by heterogeneous immunoassay, thereby providing techniques which could be readily adapted to use with devices fabricated in the course of this work.

3.1 Follicle Stimulating Hormone and Luteinising Hormone

Follicle stimulating hormone (FSH) and luteinising hormone (LH) are both glycoprotein hormones, with relative molecular masses of approximately 34,000 and 28,500 respectively.²²⁰ Both hormones contain α and β subunits, the α subunits are identical in the two hormones and consist of a single polypeptide chain containing 89 amino acids and two carbohydrate side chains. The amino acid sequences of the α subunits are also common with those of the other glycoprotein hormones, thyroid stimulating hormone (TSH) and human chorionic gonadotrophin (hCG).²²¹ The β chains of FSH and LH contain polypeptide chains of 118 and 115 amino acids, respectively, but with little homology, and the associated carbohydrate chains also differ,²²² the carbohydrate content of FSH is approximately 30% whilst that of LH is about 15%. The β side chain found in LH is very similar to that found in hCG.²²²

3.1.1 Physiological function of FSH and LH.

FSH and LH are present in both males and females, and are secreted by the anterior pituitary under the control of the hypothalamus.²²² Secretion is subject to negative and positive feedback at both the hypothalamus and the pituitary by the secretory products of the gonads. The hypothalamic control of FSH and LH release (Figure 3.1) is by means of a decapeptide, gonadotrophin releasing hormone (GRH).



Figure 3.1 Schematic diagram illustrating production of gonadotrophin releasing hormone (GRH) by the hypothalamus which stimulates production of FSH and LH by the pituitary, which in turn stimulates the production of hormones by the gonadal tissue. Androgens and inhibin exert a negative feedback influence on the release of the gonadotrophins (LH and FSH), whereas oestradiol and progesterone can exert either a positive or negative feedback influence in females, dependent upon the time in the menstrual cycle.

Before puberty, FSH and LH production is low, but an early sign of puberty is fluctuating but sub-threshold secretion of both of these hormones. Once sustained secretion of supra-threshold hormonal levels occurs, gonadal growth in both sexes occurs. FSH and LH are responsible for the development of secondary sexual characteristics in both sexes, and for the onset of menstruation in females, these hormones are therefore referred to as gonadotrophins. In females of reproductive age FSH and LH levels change during the course of the menstrual cycle as illustrated in Figure 3.2.



Figure 3.2 Typical plasma FSH and LH concentrations during the menstrual cycle.²²³ The average length of the menstrual cycle is 28 days, the first day being the onset of menstrual flow, with ovulation occurring around day 14. At the start of the cycle the anterior pituitary releases increased quantities of FSH and LH in response to low circulating blood concentrations of oestradiol and progesterone. FSH in particular stimulates the development of ovarian follicle, as ovulation approaches the follicle produces high concentrations of oestradiol which, via positive feedback, cause a surge in FSH and LH release. This increased FSH and LH concentration results in rupture of the follicle, and the release of the ovum. Following ovulation oestradiol levels decrease, whilst progesterone levels increase reaching a maximum around day 21 of the cycle, at which time a second peak in oestradiol secretion occurs. The high circulating levels of progesterone and oestradiol inhibits the release of FSH and LH, the levels of which fall.

The average length of the menstrual cycle is 28 days, the first day being the onset of menstrual flow with ovulation occurring around day 14. At the start of the cycle the anterior pituitary releases increased quantities of FSH and LH in response to low circulating blood concentrations of the gonadal steroid hormones, oestradiol and progesterone. FSH in particular stimulates the development of ovarian follicle. As ovulation approaches the follicle produces high concentrations of oestradiol which, via positive feedback, causes a surge in FSH and LH release. This increased FSH and LH concentration results in a growth spurt and rupture of the follicle, and the release of the ovum. Under the influence of LH (and to a lesser extent FSH), the ruptured follicle develops into the corpus luteum. The corpus luteum is the major source of the gonadal steroids during the post-ovulatory (luteal) phase of the cycle, which causes development and thickening of the endometrium in readiness for pregnancy.

Following ovulation oestradiol levels start to decrease, whilst progesterone levels increase reaching a maximum at around day 21 of the cycle, at which time a second peak in oestradiol secretion occurs. The high circulating levels of progesterone and oestradiol inhibits the release of FSH and LH, and the corpus luteum which is reliant on LH for support decays. As the corpus luteum decays there is a marked decrease in oestradiol and progesterone secretion, and there is an associated sloughing and haemorrhage of the superficial layers of the endometrium and the cycle ends. If fertilisation does occur, the corpus luteum persists maintained by chorionic gonadotrophin (as already mentioned LH and hCG are very similar).

In males FSH stimulates the growth of the seminiferous tubules and maintains spermatogenesis, whilst LH promotes the secretion of androgens by the interstitial (Leydig) cells, hence LH's alternative name, interstitial cell stimulating hormone (ICSH).

Plasma FSH and LH levels increase with age in both sexes. In females this increase is associated with dwindling ovarian function, and the decreased effectiveness of negative feedback control by the gonadal steroid hormones. Upon the onset of the menopause atypical values are frequently measured, often with peak concentrations at the beginning and end of each cycle. Post menopausal plasma FSH and LH concentrations are increased and remain elevated.^{221,222} Males do not show such an abrupt increase in levels, but older men tend to have a higher output of FSH.²²⁴

3.1.2 Diagnostic utility of FSH and LH

The determination of LH and FSH is of use in investigating male infertility. The majority of male infertility cases are eugonadotrophic with normal or slightly decreased LH and FSH levels, the infertility being a result of oligospermia due to failure of the seminiferous tubules. In hypergonadotrophic (primary) hypogonadism, the primary abnormality is in the testes. An increased LH indicates that there is failure of function of the Leydig cells, whereas an increased FSH indicates that there is a defect in the function of the Sertoli cells of the seminiferous tubules. In hypogonadotrophic (secondary) hypogonadism the primary abnormality is in the primary abnormality is in the testes. An increased FSH indicates that there is a defect in the function of the Sertoli cells of the seminiferous tubules. In hypogonadotrophic (secondary) hypogonadism the primary abnormality is in the hypothalamus or pituitary, LH levels are low, and usually FSH levels are also decreased. The measurement of the gonadotrophins is important in being able to distinguish between hypergonadotrophic and hypogonadotrophic hypogonadism, as in both cases testosterone levels are low.^{222,225}

The measurement of FSH and LH is most frequently used as a "fertility test" in women, or to differentiate between primary and secondary hypogonadism. Again in primary hypogonadism, the gonadotrophins will be increased due to negative feedback. Primary hypogonadism with a raised FSH and LH and low oestrogens is indicative of primary failure (which may be due to a chromosomal abnormality, autoimmune disease or idiopathic). An elevated LH with associated increased oestrogen levels, but a decreased or "low normal" FSH is indicative of polycystic ovary disease. In secondary hypogonadism low levels of FSH, LH and oestrogens are found, and in this case the abnormality is in the hypothalamus or pituitary or secondary to another endocrine disorder.^{222,225}

The determination of LH and FSH is a "first line" test in the investigation of infertility in both males and females. Stimulation tests, where a stimulus is applied to the system and the resultant change in gonadotrophin levels is monitored, can be of use in the investigation of infertility for both sexes.²²⁵

LH levels, in particular, are frequently needed for immediate decision making for assisted conception techniques, and there is a need for reliable assays where results can be made available within two to three hours of sampling.²²⁶ These tests are to determine when ovulation has occurred and to increase the chance of conception by monitoring the surge in gonadotrophin levels.

3.2 Review of Assay Methods Available for the Assay of FSH and LH

3.2.1 In vivo bioassays

A number of *in vivo* bioassays have been used for the determination of LH and FSH. LH has been measured by monitoring the weight increase of the ventral prostate of hypophysectomised immature male rats after administration of the sample preparation.²²⁷ Similarly, the weight increase of the seminal vesicles in intact or hypophysectomised immature rats has also been used,^{228,229} as the increase in weight is dependent upon the LH concentration in the sample. An alternative *in vivo* bioassay for LH was the ovarian ascorbic acid depletion assay. In this procedure immature female rats were pre-treated with pregnant mare's serum gonadotrophin and hCG. After the injection of the LH containing sample, the rat's ovaries were dissected and ascorbic acid, the concentration of which was related to the LH concentration, was determined.²³⁰ FSH has been determined by monitoring the weight increase of rodent ovaries, when excess LH or hCG is administered simultaneously with the FSH containing test material.^{231,232}

3.2.2 In vitro bioassays

LH has been assayed by monitoring the steroidogenic response of gonadal tissue. Either the production of progesterone by ovarian tissue ^{233,234} or testosterone by testicular interstitial cells²³⁵⁻²³⁹ can be used, the progesterone or testosterone produced being assayed by radioimmunoassay. LH has also been determined by combining the ovarian ascorbic acid depletion test with histochemical techniques, the reducing activity of ovarian slices was assessed by staining with Prussian blue and performing microdensitometry.²⁴⁰ The measurement of oestradiol production by cultured Sertoli cells provides an assay for FSH.²⁴¹

In vitro radioreceptor assays exist for the assay of both LH²⁴²⁻²⁴⁴ and FSH.^{245,246} Radioreceptor assays depend on the competition for binding sites on gonadal tissue preparations, between the analyte in the test material and a tracer amount of radiolabelled hormone. The bound analyte and tracer are separated from the unbound fraction by centrifugation. The radioactivity in the bound fraction is inversely proportional to the concentration of the analyte in the test material.

3.2.3 Haemagglutination assays

The haemagglutination assay is a semi-quantitative test that is capable of distinguishing samples with very elevated analyte levels, from those normally encountered. The principle of the assay is that red blood cells are coated with antibodies raised against the analyte. Upon addition of the sample, the analyte is bound to the antibodies and therefore to the red blood cells. Each analyte molecule, has multiple antigenic sites, therefore when the substance being measured is in high concentration, "cross-linking" of the blood cells occurs and an agglutination pattern is observed which can be visually differentiated from normals.²⁴⁷ The main advantages of haemagglutination assays are that they do not require specialist equipment, are relatively rapid to perform and are cheap. The major disadvantage is that they provide only very limited information, and will miss many cases which a fully quantitative assay would identify. However, haemagglutination assays for gonadotrophins have been developed, and are commercially available.²⁴⁸

3.2.4 Radioimmunoassay

FSH and LH have been measured by radioimmunoassay (RIA). This procedure uses radioisotopes as tracers to monitor the distribution of free and bound antigen in the presence of a known quantity of antibody. RIA depends on the competition for antibody binding sites between unlabelled antigen (Ag) in the sample and a known amount of added labelled antigen (Ag^*). As long as the antibody is not in excess, there is competition between the reactions in equations 3.1 and 3.2. The greater the amount of unlabelled antigen in the sample, the lower the amount of labelled antigen (and therefore, radioactivity) that is bound by the antibody (Figure 3.3). Separation of bound and free tracer has been achieved by centrifugation,²⁴⁹⁻²⁵² ethanol precipitation,²⁵³ filtration²⁵⁴ and electrophoresis.²⁵⁵

> $Ab+Ag \Leftrightarrow AbAg$ (3.1) $Ab+Ag^* \Leftrightarrow AbAg^*$ (3.2)



Log Antigen Concentration

Figure 3.3 shows the semi-logarithmic dose response curve obtained in radioimmunoassay. This procedure uses radioisotopes as tracers to monitor the distribution of free and bound antigen in the presence of a limited quantity of antibody. RIA depends on the competition for antibody binding sites between unlabelled antigen in the sample and a known amount of added labelled antigen. The greater the amount of unlabelled antigen in the sample, the lower the amount of labelled antigen (and therefore, radioactivity) that is bound by the antibody.

Early RIA methods^{249-252,254,255} used ¹³¹iodine as the radiolabel, which decays by β and γ emission. The half life (t₃) of this isotope is 8.05 days and consequently it has a higher activity than ¹²⁵ iodine, which decays with the production of γ radiation and has a half life of 60.2 days. The longer shelf life of ¹²⁵I, due to it's slower rate of decay, has meant that this is the most frequently used radiolabel in immunoassay.^{256,257} Occasionally, the incorporation of iodine (a relatively large atom) into the structure of the antigen results in steric hindrance of the antibody-antigen reaction,²⁵⁸ and in these cases ³H is frequently used as the radiolabel (β emitter, t₄ = 12.3 years).

Prior to the advent of monoclonal antibodies, great problems were encountered in obtaining specific antibodies that did not cross react with other glycoprotein hormones.²⁵³ In most immunoassays for LH there was considerable cross-reactivity with hCG,²²² due to their structural similarities. This cross reactivity was exploited in early RIA methods for LH where antibodies against hCG were used as the basis of the assay, or more commonly radiolabelled hCG was used for the tracer and unlabelled hCG was used for standards.^{249,254} Numerous RIA methods have been developed for the determination of LH and FSH levels and are commercially available.^{257,260-262}

Radioimmunoassay is generally labour intensive and time consuming, although in recent years automated sample processing systems have reduced the labour required, and the time taken to obtain a result,²⁶³⁻²⁶⁵ however they still require operation by highly trained staff.

3.2.5 Immunoradiometric assay

Immunoradiometric assay (IRMA)²⁶⁶ is a non-competitive technique where an excess of radiolabelled antibody against the analyte is added to the sample. Separation of free and antigen-bound radiolabelled antibody is performed by the addition of a second antibody (immobilised on a solid phase) which is also directed against the antigen. After removal of the radiolabelled antibody by washing, the quantity of radiolabelled antibody bound to the solid phase is proportional to the antigen concentration in the sample (Figure 3.4). Due to the low background activity, IRMA techniques tend to have lower detection limits than their RIA counterparts. Numerous IRMA methods have been developed for the determination of LH and FSH levels and are commercially available.^{261,262, 267,271}





Figure 3.4 Semi-logarithmic dose response curve of an immunoradiometric assay. Immunoradiometric assay is a non-competitive technique where an excess of radiolabelled antibody against the analyte is added to the sample. Separation of free and antigen-bound radiolabelled antibody is performed by the addition of a second antibody (immobilised on a solid phase) which is also directed against the antigen. After removal of excess radiolabelled antibody by washing, the quantity of radiolabelled antibody bound to the solid phase is proportional to the antigen concentration in the sample.

3.2.6 Dual Radioisotopic assay

 γ Radiation from different radioisotopes display different energy spectra, therefore it is possible to measure 2 analytes simultaneously by differentiating radioactivity arising from two different labels.²⁷² These types of assay are most appropriate for the measurement of those analytes that are measured in the same sample such as FSH / LH, TSH / thyroxine and Vitamin B₁₂ / folate.²⁵⁷

3.2.7 Non-Isotopic immunoassays

As has already been discussed, in recent years there has been a move away from the use of radioactive labels, and they have largely been replaced by amongst others luminescent,²⁷³ fluorescent,⁹³ and enzymatic labels.^{94,95} These non-isotopic labels possess a number of advantages such as safety, longer shelf life, shorter assay time, easier automation, improved detection limits and they introduce the possibility of developing disposable devices that can be used to perform near-patient testing.

3.2.7.1 Enzymatic immunoassay

Enzymes are the most widely used labels in immunoassays, the reactions can be followed colourimetrically, fluorimetrically, electrochemically, calorimetrically, or by using luminescence. Like other immunoassays, those using enzymes can be classified into two types, heterogeneous and homogeneous. In heterogeneous enzyme immunoassay, after the antibody and antigen have been incubated together, a step is required to separate the free antigen and antibody from the antibody-antigen complexes. The enzymic activity of one or both of these fractions is then assessed. The most common heterogeneous enzyme

immunoassay is enzyme-linked immunosorbent assay (ELISA). In this technique either the antigen or the antibody can be labelled with the enzyme. In the so called 'Sandwich' ELISA assay, an excess of immobilised antibody is incubated with the sample and the antigen binds with the immobilised antibody. After washing away the unbound sample constituents the immobilised antibody-antigen complex is incubated with excess enzyme-labelled antibody. This enzyme labelled antibody binds to a second antigenic site and excess enzyme-labelled antibody is washed off, therefore the remaining enzyme activity is directly proportional to the antigen concentration. A typical ELISA dose response curve is shown in Figure 3.5



Log Antigen Concentration

Figure 3.5 Typical semi-logarithmic dose response curve for a "sandwich" ELISA. In this technique an excess of immobilised antibody is incubated with the sample and the antigen binds with the immobilised antibody. After washing away the unbound sample constituents the immobilised antibody-antigen complex is incubated with excess enzyme-labelled antibody. This enzyme labelled antibody binds to a second antigenic site and excess enzyme-labelled antibody is washed off, therefore the remaining enzyme activity is directly proportional to the antigen concentration.
In homogeneous enzyme immunoassays (enzyme multiplied immunoassay technique (EMIT)), a separation process is not required, as the antigen-coupled enzyme activity changes when bound to an antibody. When the sample is added to this system, less of the antigen-enzyme conjugate will bind to the antibody, as it is in competition for the antibody binding sites with free antigen from the sample, therefore the change in enzymic activity is inhibited. A typical EMIT dose response curve is shown in Figure 3.6. ELISA is generally applicable to the measurement of almost any antigen, and can detect substances at much lower concentrations than can EMIT largely due to the lower inherent background signal for ELISA.



Figure 3.6 Typical semi-logarithmic dose response curve for an EMIT assay. In this technique a known amount of an antigen-enzyme conjugate is added to the sample in the presence of a limited amount of antibody, therefore there is competition for the antibody between the antigen present in the sample and the antigen-enzyme conjugate. If the antigen-enzyme conjugate is bound by the antibody then the enzymic activity is inhibited. The greater the amount of antigen in the sample, the less antigen-enzyme conjugate is bound and therefore there is less inhibition. The result is that the greater the antigen concentration in the sample the greater the enzymic activity.

In assays using colourimetric measurement of enzyme labels, the enzyme action results in a change in absorbance of one of the components of the reaction, similarly fluorimetric assays detect a change in the concentration of a fluorescent component of the reaction. Colourimetric^{94,95,274-276} and fluorimetric^{277,278} enzyme immunoassays have been developed for the determination of FSH and LH. Likewise, electrochemical enzyme immunoassays have been developed which measure a change in electroactive species within the reaction mixture as a result of enzyme activity. For example, both amperometric²⁹ and potentiometric³⁵ enzyme immunoassays have been developed for hCG, which as has already been mentioned, is similarly very similar to LH. Finally, calorimetric assays, that use a thermistor which detects the heat generated by enzymatic reactions,^{74,75} have been used for immunological analysis by labelling antibodies with enzymes.^{77,78} Luminescence detection of enzyme labels is discussed in section 3.2.7.3

3.2.7.2 Fluorescence Immunoassay

Fluorescent labels can be used for immunoassay, again classification into homogeneous and heterogeneous assays can be made. In homogeneous fluorimmunoassays the properties of the label are modified upon the antibody-antigen reaction taking place. This modification to the signal can be in the form of enhancement, quenching or a change in polarisation.²⁷⁹ In classical heterogeneous fluorimmunoassays, either antigen or antibody is labelled and assays are performed according to the principles of heterogeneous immunoassays that have already been described. These assays have tended to suffer from problems such as light scattering, quenching and background fluorescence, which has limited their use. However, the use of time-resolved fluorescence techniques has provided methods which are viable alternatives to other detection techniques, and are now commonly used.

Time-resolved fluorescence techniques (e.g. dissociation enhanced lanthanide fluorimetric immunoassay (DELFIA))²⁸⁰ are based upon the long fluorescent half life of europium chelates. The excitation light is in the form of a flash lasting less than a microsecond, the background fluorescence due to endogenous material in the sample decays rapidly with a half life of less than 50 nanoseconds. Measurement of the fluorescence arising from the europium ion can be measured after hundreds of microseconds. Time-resolved fluorescence techniques for the determination of FSH and LH are commercially available and widely used.^{248,260,261,281}

3.2.7.3 Luminescence immunoassay

Luminescence is a term that encompasses a range of processes which produce light. Luminescence, in the forms of chemiluminscence and bioluminescence are increasingly being used to monitor immunological reactions. Chemiluminescence is the emission of light as a result of a chemical reaction, bioluminescence is where the catalytic activity of a protein increases the efficiency of a chemiluminscent process. Luminescence has been used both as a label in its own right, and in reactions involving an enzymatic label. Luminescent reactions are very sensitive and have very low detection limits, as no external light source is required, all the light falling on the detector originates from the luminescent reaction. This means that the background signal is essentially zero, and that detection of a single photon produced in a luminescence reaction is possible.²⁷³ In assays where a luminescent molecule itself is used to label an antibody or antigen, light is generated by the provision of a stimulus to the luminescent molecule, for example in Figure 3.7, the addition of an oxidising agent to luminol results in its conversion to aminopthalate with the production of light. In assays where an enzyme is used as the label, there is a catalytic conversion of the solution phase luminescent molecule For example in Figure 3.7 horseradish peroxidase can

catalyse the conversion of luminol to aminopthalate. Commercial assays based on the measurement of luminescence are available for both FSH and LH.^{257,282-285}



Catalysed by horseradish peroxidase in biolumiscence

Figure 3.7 Scheme for the chemiluminescent oxidation of luminol.

3.2.8 Near-patient tests

Rapid near-patient tests for the assay of LH have been developed, and are commercially available.²⁴⁸ These tests are designed to determine when ovulation has occurred by monitoring the LH surge. This information can be used to increase the chance of pregnancy by predicting when a woman is most likely to conceive, but perhaps the most useful application of rapid gonadotrophin measurements is in assisted conception. The availability of a simple and rapid assay makes possible successful and precise timing of ovulation during artificial insemination or ovum retrieval for *in-vitro* fertilisation (IVF) procedures in treatment for infertility.²⁸⁶ One such near-patient device is the OvukitTM (formerly known as the OvutestTM), which is a semi-quantitative assay system based upon immunochromatography (the basis of the majority of pregnancy tests). In the OvukitTM an antibody specific for LH is immobilised as a line on a membrane. The sample (urine) is applied to one end of the membrane, and "wicks" along the length of the membrane and any LH in the sample is "captured" by the immobilised antibody. A similar device is the Unipath ClearPlan One Step TM device, this easy to use device can provide a result in just 3 minutes. Visualisation is by means of a conjugate which produces a coloured line visible to the naked eye, if the levels of LH are above a threshold level.^{287,288} The assay is optimised so that a positive result is only obtained during the LH surge, and therefore indicates that ovulation is occurring.

3.3 Amperometry

The immunosensor constructed in this work was an amperometric sensor, in which an electrical potential is applied to a sensor electrode, which facilitates reduction or oxidation of electroactive species at the electrode-solution interface. This "redox" reaction involves the transfer of electrons between the electrode and the electroactive species, with a current being produced which is proportional to the concentration of the electrochemical species at the electrode surface. As has already been described, amperometric transducers have been very widely used in biosensing, particularly systems based upon oxygen or hydrogen peroxide electrodes, where the current produced is directly proportional to the amount of oxygen or hydrogen peroxide reduced or oxidised at the electrode. This approach has high sensitivity and a linear concentration dependence over a given concentration range, and improved selectivity over interferents can be obtained by careful choice of the electrode potential. Many molecules (such as proteins) are not intrinsically electroactive, and therefore it is frequently necessary to introduce enzymic labels which can catalyse redox reactions and produce electroactive species. Amperometric sensing of immunological

reactions can be performed by using an immobilised antibody to bind the analyte at the electrode surface, and then adding a second enzyme labelled antibody which is also directed against the analyte.

In amperometric devices the basic requirement is to be able to maintain a working electrode at a constant potential and measure any resultant current. This can best be achieved using a three electrode system, with a working electrode being maintained at a constant potential with respect to a reference electrode, whilst allowing current to flow between the working electrode and a counter electrode in solution. When the size of the current is small (< μ A cm⁻²), the configuration can be simplified by removing the counter electrode, and allow the (measured) current to return via a combined reference/counter electrode.

The observed current is a combination of capacitive or "non-Faradaic" currents (which are due to the physical rearrangements of ions in the double layer) and the Faradaic current (which is due to redox reactions). The Faradaic current is a direct measure of the rate of the electrochemical reaction taking place at the electrode. In practice this will depend upon two dominating factors; the rate at which the material arrives at the electrode surface from the bulk solution (mass transport) and the rate at which electrons transfer between the electrode and the electroactive species (charge transfer), which is described by the Butler Volmer equation,²⁸⁹ and relates to the extent of electron transfer to the applied overpotential and the transfer coefficient.

Mass transport comprises of three components; diffusion, migration and convection. Diffusion is the dispersion of a species within another species, tending to a homogeneous mixture without chemical combination (i.e. movement "down" a concentration gradient).

Migration is the movement of charged species in an electric field, and is therefore dependent upon the magnitude of the field and the charge on the ion. Convection is the movement of analyte molecules due to inducing the physical movement of the bulk solution (e.g. stirring). These three components of mass transport to the electrode are described (in one dimension) by the Nernst-Planck equation.²⁸⁹ If the migrational and convectional contributions to the mass transport flux can be eliminated then the steady state diffusion limited current can be determined as in equation 3.1. For the elimination of convective transport, all that is required is to keep the solution quiescent, whilst the addition of a supporting electrolyte at a concentration 100 times that of the electroactive species, essentially eliminates migration by dissipating the electric field over all of the ions in solution, thus "shielding" those ions of interest from electro-migrational forces).

$$i(t) = nFAD \frac{\delta C(x,t)}{\delta x} \int_{x=0} (3.1)$$

where *n* is the number of electrons involved in the reaction (Faraday equivalents mol⁻¹), *A* is the electrode area, (cm²), *D* is the diffusion coefficient (cm² s⁻¹), *C* is the concentration (mol cm⁻³) and F = 96,484.6 C Faraday⁻¹.

3.3.1 Chronoamperometry

Chronoamperometry is a technique where an applied potential is stepped from an initial value to a different value at which the potential is held for a period of time (i.e. a square wave). The resultant current is recorded as a function of time. A typical chronoamperometric response curve displays a sharp current spike (at the time at which the potential step occurs and consisting of both Faradaic and non-Faradaic components) which

decays with time. In chronoamperometry, the applied potential is sufficient to reduce the electrode surface concentration of the electroactive species essentially to zero.²⁸⁹ The effect of the capacitive current becomes less evident with time and the Faradaic current (which is a direct measure of the rate of the electrochemical reaction taking place at the electrode) dominates. Typical chronoamperometric input and output waveforms are shown in Figure

3.8



Figure 3.8 Typical input and ouput chronoamaperometry waveforms. The input waveform comprises of an applied potential which is stepped from an initial value to a different value at which the potential is held for a period of time (i.e. a square wave). The resultant current is recorded as a function of time (output waveform). A typical chronoamperometric response curve displays a sharp current spike (at the time at which the potential step occurs and consisting of both Faradaic and non-Faradaic components) which decays with time. In chronoamperometry the applied potential is sufficient to reduce the electrode surface concentration of the electroactive species essentially to zero. The effect of the capacitive current becomes less evident with time and the Faradaic current (which is a direct measure of the rate of the electrochemical reaction taking place at the electrode) dominates.

3.4 Materials

Positive S1400-31 photoresist and Microposit developer were obtained from Shipley Europe, Coventry, England. Silicon wafers were obtained from Shin Etsu, Livingston, Scotland; NiChrome (80:20 Ni:Cr) and gold were obtained from Goodfellow Cambridge Ltd, Cambridge, England.

Ferrocenemonocarboxylic acid, anti-mouse IgG -ALP conjugate, N-acetyl-l-cysteine, and 1-ethyl-3-(3-dimethylaminopropyl)carbodiimide were obtained from Sigma, Poole, England. Purified FSH, purified LH, anti-FSH (clone BIO-FSHB-003), anti-LH (clone BIO-LH-007), HRP labelled anti-FSH (clone BIO-FSHB-002), and HRP labelled anti-LH (clone BIO-LH-005) were obtained from Biogenesis Ltd, Bournemouth, England. Neutravidin[™] and casein were obtained from Pierce Warriner, Chester, England. Photobiotin was obtained from Vector Laboratories, Peterborough, England. A Bioanalytical Systems (BAS) gold electrode and a BAS RE4 Ag/AgCl reference electrode were obtained from Biotech Instruments Ltd, Luton, England.

Instrumentation used comprised a EG&G 273A potentiostat (EG & G, Sunninghill, England), Bioanalytical System CV-37 potentiostats (Biotech Instruments Ltd, Luton, England), a Goerz SE120 dual channel chart recorder (Belmont Instruments, Glasgow, UK), a 100W HB-10101AF super high pressure mercury vapour lamp (Nikon, Tokyo, Japan) and a photolithographic mask aligner (HTG San Jose, California, USA).

3.5 Method

3.5.1 Photolithographic Lift Off patterning techniques for the fabrication of electrodes

Positive organic resists such as Shipley S1400-31 photoresist are composed of long molecular chains with high molecular weight. Upon exposure to ultraviolet (UV) light, extensive chain scission takes place resulting in fragmented molecules with low molecular weight. A developer which dissolves the molecules at a rate dependent upon their molecular weight is then employed. Those regions that have been exposed to ultraviolet light, with molecules of relatively low molecular weight dissolve rapidly. Unexposed high molecular weight regions dissolve relatively slowly. Therefore, after development, the resist image has "windows" opened where it has been exposed to UV light. This process is shown schematically in Figure 3.9a-d.

In the "lift-off" technique, metal is evaporated from a point source and is deposited both onto the resist and onto the substrate through the resist windows. The resist is then dissolved by immersion in a resist solvent, causing the overlying metallisation to lift off, therefore only the metal that has been deposited directly onto the substrate through the resist window remains. This process is shown schematically in Figure 3.9e-f.

Areas of the deposited metal can subsequently be electrically insulated from the environment by patterning a second layer of photoresist on to the surface. On this occasion only the areas of the metallised surface that one wishes exposed to the environment are irradiated, after removing the resist from these areas the remaining resist is hardened in an oven and can then act as an electrical insulator This process is shown schematically in Figure 3.9g-h.





Figure 3.9 Fabrication of electrodes using photolithographic lift-off techniques. A substrate (typically a silicon wafer) is cleaned (a) and coated with positive photoresist (b). The surface is exposed to UV light though a mask (c). A developer dissolves the resist that has been exposed to ultraviolet light (d). Metal is evaporated from a point source and is deposited both onto the entire surface (e). The resist is then dissolved by immersion in a resist solvent, causing the overlying metallisation to lift off, therefore only the metal that has been deposited directly onto the substrate through the resist window remains (f). Areas of the deposited metal can be electrically insulated by patterning a second layer of photoresist on to the surface. Only the areas of the metallised surface that one wishes exposed to the environment are irradiated (f), after removing the resist from these areas the remaining resist is hardened in an oven and can then act as an electrical insulator (h)

3.5.2 Electrode fabrication

Positive S1400-31 photoresist was spun onto 10 cm diameter silicon wafers at 4000 rpm for 30 s and then baked at 90°C for 30 minutes. The resist was then illuminated by light for 12 s from a mask aligner through a photolithographic mask containing dark field relief images of the electrode arrays (Figure 3.10). Exposed resist was removed by immersion in Microposit developer for 75 s followed by a rinse in reverse osmosis (RO) water. NiChrome and gold were deposited on the surface to a thickness of 10nm and 100nm respectively, using a modified Edwards 12E High Vacuum evaporation unit. The remaining photoresist was removed by immersion in acetone for 20 minutes, "lifting off" the metal deposited on it, and leaving the silicon wafer with the metallic electrode arrays on its surface. Again, S1400-31 photoresist was spun onto the surface and baked at 90°C as detailed above. The electrodes and bonding pads were exposed to UV light through a second mask. Exposed resist was removed as detailed above, and the wafer was then baked at 120°C for 120 minutes, to harden the remaining photoresist which acted as an electrical insulator.

Ag/AgCl reference electrodes were prepared by electrosorbtion of silver onto specified gold electrodes from a solution of 0.1M AgNO₃ in 0.1M sulphuric acid with a silver anode (at a constant current of 0.4 mA cm⁻² for 6 hours), followed by chloridisation in 0.1M HCl (0.4 mA cm⁻² for 30 minutes).



Figure 3.10 Diagrammatic representation of the fabricated electrode array for use in the construction of a multianalyte immunosensor.

The electrochemical behaviour of the fabricated electrodes was verified by comparison of cyclic voltammograms (-0.2 to +0.75 V, 20 mV s⁻¹) of 0.2 mM ferrocenemonocarboxylic acid in 50 mM Tris 50 mM KCl, pH 7.4 with those obtained using a BAS gold working electrode and a BAS RE4 Ag/AgCl reference electrode. Reproducibility of the fabricated electrode arrays, was assessed by measuring the chronoamperometric response (10 s at 0V, 30 s at +650 mv)) to 0.5 mM H₂O₂ in 50 mM sodium phosphate buffer containing 50 mM KCl, pH 7.4. These studies were performed using an EG&G 273A potentiostat.

3.5.3 Antibody immobilisation

Electrode arrays were incubated in 2 mM NAC in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature, followed by 120 minutes incubation in 1% (w/v) EDC in 10 mM phosphate buffer (pH 7.0). Note that the use of NAC provides an "open" self assembled monolayer (in contrast to highly organised SAMs), and this does not "block" the gold electrochemical surface. The electrode arrays were then incubated in $100 \mu g \text{ ml}^{-1}$ Neutravidin in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C. All subsequent stages of the immobilisation procedure were performed at room temperature. After washing in PBS, the electrodes were first incubated in 10 mg ml⁻¹ casein in PBS for 60 minutes and then in 10 µg ml⁻¹ long arm photobiotin in PBS, for 20 minutes in the dark. All subsequent stages of the immobilisation procedure were performed in the dark. After washing in PBS the wafer was covered with 10 µg ml⁻¹ monoclonal anti-FSH (clone BIO-FSHB-003), and selected electrodes were exposed to light from a mercury vapour lamp, using a suitable mask as described in section 2.3.1. After washing in PBS, the wafer was covered with 10 µg ml⁻¹ monoclonal anti-LH ((Biogenesis clone LH-007), and selected electrodes were exposed to light from the lamp for 15 minutes. After washing in PBS, the entire wafer was exposed to light from the lamp for 15 minutes in the presence of 10 mg ml⁻¹ casein in PBS, and again washed in PBS.

3.5.4 Immunoassay procedure

The immunoassay procedure used was a "sandwich" ELISA technique, with the immobilised "capture" antibodies on the electrode surfaces, and the subsequent addition of enzyme labelled antibodies directed against second epitopic sites on the antigens (Figure 3.11). Sensors were incubated with 250 μ l of sample for 60 minutes, washed thoroughly with PBS, and incubated in a mixture of 10 μ g ml⁻¹ horseradish peroxidase (HRP) labelled anti-FSH ((Biogenesis clone BIO-FSHB-002), and 10 μ g ml⁻¹HRP labelled anti-LH((Biogenesis clone LH-005) in PBS, for 60 minutes at room temperature, followed by further washing in PBS.



Figure 3.11 Principle of the "sandwich" ELISA used in the determination of LH and FSH. The specific capture antibody (BIO-FSHB-003 for FSH and BIO-LH-007 for LH) was immobilised onto the electrode surface using the avidin - photobiotin patterning technique. Upon exposure to the sample the antibody bound its complimentary antigen, and unbound material was washed away. A second specific antibody (BIO-FSHB-002 for FSH and BIO-LH-005 for LH) conjugated to HRP was added, and this bound to a second discrete epitope on the antigen. Therefore, the amount of HRP attached to the electrode surface was proportional to the concentration of the antigen in the sample.

Simultaneous assessment of HRP activity at the two sensors was performed chronoamperometrically using two Bioanalytical System CV-37 potentiostats and a Goerz SE120 dual channel chart recorder. Activity was determined by measuring the current produced at 20 s in the presence of 10 mM hydrogen peroxide, and 0.2 mM ferrocenemonocarboxylic acid in 50 mM phosphate buffer containing 50 mM KCl, pH 7.4, with a potential of +150 mV applied to the working electrode versus the Ag/AgCl reference electrode.²⁹⁰ The principle of using ferrocenemonocarboxylic acid as an electron mediator is illustrated schematically in Figure 3.12



Figure 3.12 Schematic representation of electrochemical detection of HRP using ferrocene derivatives. As HRP catalyses the conversion of H_2O_2 to H_2O , it is oxidised in the process. HRP is reconverted into the reduced form by the ferrocene (bis(h⁵-cyclopentadienyl)iron) derivative (in this case ferrocenemonocarboxylic acid), which is itself oxidised in the process and is subsequently reduced by gaining electrons from the electrode, resulting in current flow.

The response of the immunosensors to hormone concentration in a buffered aqueous solution was determined by preparation of a series of standards ($0 - 100 \text{ U} \text{ I}^{-1}$) of FSH and LH (Biogenesis) which covered the concentration range of clinical interest. The results obtained (Section 3.6.2) were subsequently used as calibration curves for further studies. The multianalyte immunosensor was used to determine gonadotrophin concentrations in 10 serum samples from hospital outpatients (with a representative range of values typically seen in clinical laboratories), and the results were compared with those obtained using an established DELFIA method.²⁹¹ Assays using the immunosensor were performed in triplicate, and the results obtained are shown in Section 3.6.2.

3.5.5 Behaviour of calibration curve over time

Sensors which had been fabricated were tested as described previously 42 days and 60 days after the initial (day 0) experiments (storage was at 4°C, in a solution of PBS containing 1% BSA). The response of the sensor to aqueous gonadotrophin concentration was assessed in triplicate at each timepoint. The responses for the FSH and LH sensors are shown in Section 3.6.3. In order to check that the immobilised antibody was not desorbing from the surface upon storage, the sensors were also incubated with an anti-mouse IgG - ALP conjugate at each time point. Following washing in TBS and incubation in the conjugate for 60 minutes, the sensors were washed extensively in TBS and their ALP activity was determined colourimetrically using pNPP as the substrate as described in Section 2.5.2.1. The absorbance was measured at 404 nm, and compared with a that of a freshly prepared reference solution (1:60000 dilution of stock conjugate), to take account of inter-assay variation.

3.6 Results

3.6.1 Electrochemical Characterisation of Electrodes

The cyclic voltammograms of 0.2 mM ferrocenemonocarboxylic acid in 50 mM Tris 50 mM KCl, pH 7.4 (-0.2 to +0.75, 20 mV s⁻¹) for the fabricated electrodes and for commercial (BAS) electrodes are shown in Figure 3.13. The potentials at which oxidation and reduction peaks were evident upon cyclic voltammetry for the fabricated electrodes were within 5 mV of those obtained when using the BAS electrodes ($E_{pa} = 0.355V$, $E_{pc} = 0.296V$), demonstrating that the fabricated electrode arrays behaved similarly to a combination of commercially obtained gold and Ag/AgCl electrodes.

The intra-batch coefficient of variation for the responses of the electrode arrays to 0.5 mMH₂O₂ in 50 mM sodium phosphate buffer containing 50 mM KCl, pH 7.4 (assessed by measuring the chronoamperometric response (10 s at 0V, 30s at +650 mV vs. Ag/AgCl, current measured 30 s after application of +650mV potential)) was 1.86% (n=20), whilst the inter-batch coefficient of variation was 2.43% (n=5), thus demonstrating that the response (and therefore the manufacture) of the fabricated electrode arrays were both reproducible.



Figure 3.13 Cyclic voltammograms of ferrocenemonocarboxylic acid obtained using fabricated electrode pair and Commercial (BAS) electrodes The electrochemical behaviour of the fabricated electrodes (manufacture described in 3.5.1) was verified by comparison of cyclic voltammograms (-0.2 to +0.75 V vs. Ag/AgCl, 20 mV s⁻¹) of 0.2 mM ferrocenemonocarboxylic acid in 50 mM Tris 50 mM KCl, pH 7.4 with those obtained using a BAS gold working electrode and a BAS RE4 Ag/AgCl reference electrode. The potentials at which oxidation and reduction peaks were evident upon cyclic voltammetry of ferrocene- monocarboxylic acid for the fabricated electrodes were within 5 mV of those obtained when using the BAS electrodes ($E_{pa} = 0.355V$, $E_{pc} = 0.296V$)

3.6.2 Response of immunosensors to aqueous and clinical samples

The response of the sensor to FSH and LH in buffer is shown in Figures 3.14 and 3.15, the current (a result of the enzymic activity) is clearly linearly related to the antigen concentration over the range 0 to 100 U l⁻¹. When corrected for the specific activities of the hormone preparations these ranges equate to 0 to 26 ng l⁻¹ and 0 to 18 ng l⁻¹ for FSH and LH respectively. Figure 3.14 demonstrates that the current produced by the FSH sensor is proportional to the FSH concentration, and that cross reactivity with LH is negligible. The response to FSH is 2.11 nA / U l⁻¹ (8.018 nA / ng l⁻¹) whilst the response to LH is $0.07 \text{ nA} / \text{U} \text{ l}^{-1}$ (0.385 nA / ng l⁻¹).

Data shown in Figure 3.15 demonstrates that the current produced by the LH sensor is proportional to the LH concentration, and that cross reactivity with FSH is negligible. The response to LH is 2.48 nA / U l⁻¹ (13.64 nA / ng l⁻¹) whilst the response to FSH is $0.11 \text{ nA} / \text{U} \text{ l}^{-1}$ (0.418 nA / ng l⁻¹).

Figures 3.16 and 3.17 show comparisons of results obtained for human serum samples from the multianalyte sensor, with those from an established DELFIA technique. There is a very good correlation between methods, and close agreement between results at all concentrations for both FSH (r = 0.9852) and, LH (r = 0.9783).



<u>Figure 3.14 Response of multianalyte sensor to FSH concentration in buffer</u> The response of the immunosensors to FSH concentration in a buffered aqueous solution was determined by preparation of a series of standards ($0 - 100 \text{ U} \text{ I}^{-1}$) of FSH (LH concentration = 0). Monoclonal anti-FSH and monoclonal anti-LH antibodies were immobilised on selected sensors using the avidin-photobiotin patterning technique.



<u>Figure 3.15 Response of multianalyte sensor to LH concentration in buffer</u>. The response of the immunosensors to LH concentration in a buffered aqueous solution was determined by preparation of a series of standards (0 - 100 U l⁻¹) of LH (FSH concentration = 0). Monoclonal anti-FSH and monoclonal anti-LH antibodies were immobilised on selected sensors using the avidin-photobiotin patterning technique.



Figure 3.16 Comparison of sensor results for FSH in clinical samples (serum) with an established DELFIA technique. Monoclonal anti-FSH and monoclonal anti-LH antibodies were immobilised on selected sensors using the avidin-photobiotin patterning technique. Sensors were incubated with 250 μ l of sample for 60 minutes, washed thoroughly with PBS, and incubated in a mixture of 10 μ g ml⁻¹ HRP labelled anti-FSH and 10 μ g ml⁻¹ HRP labelled anti-LH in PBS, for 60 minutes at room temperature, followed by further washing in PBS. HRP activity was assayed chronoamperometrically measuring the current produced at 20 s in the presence of 10 mM hydrogen peroxide, and 0.2 mM ferrocenemonocarboxylic acid in 50 mM phosphate buffer containing 50 mM KCl, pH 7.4, with a potential of +150 mV applied to the working electrode versus the Ag/AgCl reference electrode. Calibration of sensor responses was versus aqueous solutions of FSH. Error bars represent mean of triplicate assays ± 1 standard deviations



Figure 3.17 Comparison of sensor results for LH in clinical samples (serum) with an established DELFIA technique. Monoclonal anti-FSH and monoclonal anti-LH antibodies were immobilised on selected sensors using the avidin-photobiotin patterning technique. Sensors were incubated with 250 μ l of sample for 60 minutes, washed thoroughly with PBS, and incubated in a mixture of 10 μ g ml⁻¹ HRP labelled anti-FSH and 10 μ g ml⁻¹ HRP labelled anti-LH in PBS, for 60 minutes at room temperature, followed by further washing in PBS. HRP activity was assayed chronoamperometrically measuring the current produced at 20 s in the presence of 10 mM hydrogen peroxide, and 0.2 mM ferrocenemonocarboxylic acid in 50 mM phosphate buffer containing 50 mM KCl, pH 7.4, with a potential of +150 mV applied to the working electrode versus the Ag/AgCl reference electrode. Calibration of sensor responses was versus aqueous solutions of LH. Error bars represent mean of triplicate assays ± 1 standard deviations.

3.6.3 Behaviour of calibration curve over time

The responses for sensors stored at 4°C for 0, 42 and 60 days to aqueous FSH and LH are shown in Figures 3.18 and 3.19 respectively. It can be seen from these figures that there was a significant decrease in the response (in terms of current) of both sensors upon storage. After 42 days storage at 4°C, the FSH sensor's response was 79% of that after preparation (Day 0) (coefficient of variation (CV)=8.2%, n=3), whilst after 60 days the response was 70% (CV=11.1%, n=3) of the initial value. After 42 days storage at 4°C, the LH sensor's response was 69% of that at Day 0 (CV=9.7%, n=3), whilst after 60 days the response was 62% (CV=6.6%, n=3) of the initial value.

The amount of antibody immobilised on the sensor surfaces as demonstrated by testing with an anti-mouse IgG-ALP conjugate did not show any variation with time (Day 0 sample:reference ratio = 2.17 (CV=7.8, n=3), Day 42 sample:reference ratio = 2.04(CV=9.2, n=3), Day 60 sample:reference ratio = 2.22 (CV=6.2, n=3)).





3.7 Discussion



Figure 3.19 Response of multianalyte sensor to LH concentration in buffer over a 60 day period.

The response of the immunosensors to LH concentration in a buffered aqueous solution was determined by preparation of a series of standards ($0 - 100 \text{ U} \text{ I}^{-1}$) of LH (FSH concentration = 0). Monoclonal anti-FSH and monoclonal anti-LH antibodies were immobilised on selected sensors using the avidin-photobiotin patterning technique. Immunoassay with chronoamperometric detection was performed as described in Section 3.5.4. The response of the sensors was measured on the day of manufacture and after storage at 4°C for 42 and 60 days. Each point on the plot is the mean response obtained from performing the assay using three sensors.

The variation in the current recorded for a given concentration of bosisses on deficite the dates could be due to a combin of factors. Minut variations is achieved contracted for the hole boop a factor (e.g. chaoses in room tracterowner), but as there are not to be a progressive decrease in content chisined for a given concents then with these is is block up that there was a loss of biolo deci activity of the ambedies and , or the maxime root. The lack

3.7 Discussion

The applicability of the avidin-photobiotin immobilisation procedure to multianalyte immunoassays has been demonstrated using the gonadotrophins FSH and LH as a model system.²⁹² This is the first report of an electrochemical multianalyte immunosensor for high molecular weight analytes.

The detection limit (defined as the analyte concentration read off the standard curve, for a current equivalent that of the zero standard plus 3 standard deviations (n=10)) was 2.1 U Γ^{1} for FSH and 1.8 U Γ^{1} for LH. The detection limits were below the bottom of the reference range for both analytes and are comparable to many existing methods,²⁶¹ although lower detection limits have been reported in some other methods.²⁷⁸ It is likely that further optimisation of the reagents and assay protocol used in the immunosensor assay would result in a lower detection limit. Over the working range of the assays (0 to 100 U Γ^{1}), the concentration:current relationship is linear for both analytes, and the immunosensor showed excellent correlation with an established analytical technique upon testing clinical samples. The replicate (×3) analysis of the 10 patient samples shows acceptable imprecision (R.M.S. CV% = 5.2% for FSH and 6.7% for LH).

The variation in the current recorded for a given concentration of hormone on different dates could be due to a number of factors. Minor variations in analytical conditions may have been a factor (e.g. changes in room temperature), but as there appeared to be a progressive decrease in current obtained for a given concentration with time, it is likely that there was a loss of biological activity of the antibodies and / or the enzyme label. The lack of variation in the quantity of antibody on the sensor with time (as demonstrated with the use of anti mouse IgG-ALP conjugate) indicates that the decreased response was not due to desorption of the capture antibody from the sensor surface.

The current that flowed when no antigen was present was due to a number of factors, chief amongst these was the current resulting from electrochemical processes unrelated to the immunoassay (i.e. the background current obtained when there was no enzymic activity). The remainder of the current when the antigen concentration was zero was due to either to non-specific binding or to diffusion of electroactive species between electrodes.

As discussed in Chapter 2 there are a number of types of non-specific binding in solid-phase enzyme immunoassays. One of these occurs where an inappropriate antibody binds at a sensor site (e.g. anti-LH on a sensor for FSH), a circumstance which can come about for a number of reasons, such as binding through non-specific protein-protein interactions, hydrophobic interactions with non-polar surfaces, or electrostatic interactions between the protein and the surface.¹²⁴⁻¹²⁶ This inappropriate antibody is subsequently able to bind it's complimentary antigen and the enzyme labelled second antibody. Sample components themselves can bind non-specifically to sensors, by the same processes as described above (e.g. LH reacting non-specifically with anti-FSH). Subsequently, if enzyme labelled-antibody to this sample component is added a "false" signal can be produced. The labelled antibodies themselves can also bind non-specifically to the sensors with the same effect. Antibody "cross-reactivity" can also cause spurious results, this can be minimised by the use, as in this study, of highly specific monoclonal antibodies. As demonstrated in Chapter 2, the use of deglycosylated avidin (e.g. Neutravidin or streptavidin) to precoat the electrodes, decreases the amount of non-specific binding, particularly that due to protein

interactions with the surface. This enabled the multianalyte immunosensor to be fabricated, with serial immobilisation of sensor antibodies. Casein was used to block unreacted photobiotin groups and to further decrease the effect of non-specific binding.

It is likely that the non-specific binding observed in this study comprised of several or all of the above types. The binding of a labelled antibody to the sensor that is not mediated by an antigen, is independent of sample composition, whereas "antigen mediated" non-specific binding will increase with the increasing concentration of the antigen in the sample. In this study there was a slight increase in the non-specific signal with increasing antigen concentration for both sensors, possibly indicating that some LH was being bound at the FSH sensor and *vice versa*. Alternatively this increase in non-specific signal could be due to the diffusion of increased concentrations of electroactive species between electrodes, although work presented in Chapter 4 would suggest that this is unlikely.

Simultaneous multianalyte immunoassays, where several analytes are measured by a simple procedure, are required in a number of clinical situations, such as the measurement of hormones related with thyroid function and the measurement of gonadotrophins for the investigation of infertility etc. Particular examples where the use of immunosensors for the measurement of gonadotrophins are likely to be beneficial, are as an aid to conception either "naturally" by determining when a woman is most likely to conceive, or for *in vitro* fertilisation procedures. In these cases the rapid measurement of gonadotrophins is extremely important as the rapid transient elevation of gonadotrophins (particularly LH) is an indication of ovulation.^{287,288,293} Existing multianalyte immunosensors tend to be qualitative assays, examples of such systems are the Triage™ system for qualitative assays of drugs of abuse and the Spectral Diagnostics assay for markers of myocardial

infarction.^{120,121} These systems are based upon immunochromatography assays, which are observed visually and are generally unsuitable for quantitative assay.

Although initially a sensor for measuring two analytes has been constructed, the technology that has been developed is compatible with the fabrication of a sensor for as many analytes as is required.²⁹⁴ The fabrication and immobilisation procedures used in this work would be suitable for larger scale production using technology commonplace in the microelectronics industry. Additionally, there is no waste of expensive proteins such as monoclonal antibodies as non-immobilised excess protein can easily be recovered, and be reused.

CHAPTER 4

Let us not take it for granted that life exists more fully in what is commonly thought big than in what is commonly thought small.

Virginia Woolf

The Common Reader

<u>4 Introduction</u>

Another objective of this work was to investigate the possibility of miniaturising the immunosensor electrodes, so as to develop microspot immunoassays as described by Ekins and co-workers.²⁹⁵⁻²⁹⁷ High sensitivity assays based upon the microspot had previously been developed using fluorescence detection, but no there were no reports of the attempted extension of this technology to systems using electrochemical detection. Miniaturisation of electrodes, along with the protein patterning technique already described, would allow the fabrication of arrays of small sensors capable of simultaneously measuring numerous analytes, the so called "immunological compact disc".

Section 4.1 provides a background to antibody antigen reactions at surfaces, Section 4.2 provides a theoretical expansion of previously published work on microspot immunoassays with particular reference to microelectrode arrays. Practical work performed to ascertain the suitability of microspot electrochemical assays is presented in the remaining sections of the chapter.

Practical work was performed to:

1) Ascertain if the formation of hemispherical diffusion fields associated with ultramicroelectrodes caused problems in the use of multianalyte sensors.

2) Ascertain if the diffusion of electrochemical species generated at one electrode to other electrodes caused problems in the use of multianalyte sensors, in particular those employing ultramicroelectrodes.

3) Construct an enzymic biosensor using ultramicroelectrodes capable of simultaneously measuring two analytes.

4) Construct a multianalyte immunosensor using ultramicroelectrodes and capable of simultaneously measuring two analytes.

4.1 Antibody antigen reactions at surfaces

In order to investigate the feasibility of developing an amperometric dual label microspot immunoassay, a degree of theoretical analysis of the principles of immunoassay is required. The equilibrium constant (K) is given by equation 4.1 where [Ag] = concentration of analyte and [Ab] = concentration of the antibody (in M⁻¹).

$$K = \frac{[AbAg]}{[Ab][Ag]} (4.1)$$

All immunoassays are based upon the measurement of the proportion of available antibody binding sites that are bound to the antigen. It can be shown that the fractional occupancy (F) of antibody binding sites by antigen at equilibrium is given by equation 4.2.²⁹⁷

$$F^{2} - F(\frac{1}{[Ab]} + \frac{[Ag]}{[Ab]} + 1) + \frac{[Ag]}{[Ab]} = 0 \quad (4.2)$$

where [Ag] = concentration of analyte and [Ab] = concentration of the immobilised sensor antibody (both in units of the reciprocal of the equilibrium constant (1/K). From this equation, it can be shown that as [Ab] tends to 0,

$$\mathbf{F} \approx \frac{[Ag]}{[1+Ag]} (4.3)$$

Equation 4.3 implies that the fractional occupancy is dependent on [Ag] and independent of the total amount of immobilised sensor antibody. The effect of this is that when very small amounts of immobilised sensor antibody are used, the assay becomes independent of sample volume. In practice this applies when analyte depletion is less than 1%.²⁹⁴ Assays

that are independent of sample volume have a number of advantages, particularly in their ease of use if they are to be used in non-laboratory environments.

4.2 microspot immunoassay

If it is assumed that antibody molecules exist with a uniform surface density on a support, then a change in surface area over which the antibody is distributed implies a change in antibody concentration. For example, as shown below if an antibody has an affinity constant (K) of 10^{11} M⁻¹, the total incubation volume is 1 ml and the antibody surface density is 6000 molecules/ μ m², a surface area of $10^{5} \mu$ m² (0.1mm²) corresponds to an antibody concentration of 0.1K.

6000 molecules/ μ m² ×10⁵ μ m² = 6 ×10⁸ molecules

6 ×10⁸ molecules ÷ 6.02 ×10²³ molecules × 1 ×10³ ≈ 1× 10⁻¹² M

 6.02×10^{23} molecules is Avagadros number (number of molecules in a mole), the factor of 10^3 is to convert 1 ml to 1 litre (volume unit used for expressing molarity).

 1×10^{-12} M (number of molecules) $\div 1 \times 10^{-11}$ M (reciprocal of affinity constant) = 0.1K Similarly, if the area is $10^4 \ \mu m^2$ (0.01mm²), and the other parameters remain unchanged then the antibody concentration is calculated as 0.01K

The solution for F in equation 4.2 is given by;

$$F = \frac{1 + \frac{1}{[Ab]} + \frac{[Ag]}{[Ab]} - \sqrt{\left(1 + \frac{1}{[Ab]} + \frac{[Ag]}{[Ab]}\right)^2 - 4x \frac{[Ag]}{[Ab]}}}{2} \quad (4.4)$$

Plots showing values of F at antibody concentrations between 0.0001K and 10K for several different antigen concentrations are shown in Figure 4.1. From these plots it can be seen that at antibody concentrations below 0.01K that for all practical purposes, the fractional occupancy F is dependent solely upon the antigen concentration.



Figure 4.1 Fractional antibody binding-site occupancy (F) related to antibody concentration for several different antigen concentrations. All concentrations are in unit of 1/K. For antibody concentrations [Ab] less than approximately 0.01/K, F is governed solely by [Ag] and is essentially unaffected by variations in [Ab].

The percentage of antigen bound by the antibody (i.e.depletion of antigen from solution) is given by Equation 4.5 (from rearrangement of Equation 4.4) and plots showing the percentage of antigen bound at antibody concentrations between 0.0001K and 10K for several different antigen concentrations are shown in Figure 4.2. Note that for antibody concentrations of less than approximately 0.01/K, the percentage of antigen bound is less than 1%.

% Antigen bound =

$$\left(\left(\left(1+\frac{1}{[Ab]}+\frac{[Ag]}{[Ab]}-\sqrt{\left(1+\frac{1}{[Ab]}+\frac{[Ag]}{[Ab]}\right)^2-4\times\frac{[Ag]}{[Ab]}}\right)\times[Ab]\right)/2/[Ag]\right)\times100$$
(4.5)



Figure 4.2 Percentage of antigen bound to antibody related to antibody concentration for several different antigen concentrations. All concentrations are in unit of 1/K. For antibody concentrations [Ab] less than approximately 0.01/K the percentage binding of the analyte is less than 1%.

It has previously been shown that a decrease in the sensor area and therefore antibody concentration (Figure 4.3a), can result in an increase in the specific/non-specific signal ratio, assuming that the "field of view" of the detecting instrument is restricted to the area on which the sensor antibody is deposited.²⁹⁴ However, if the amount of capture antibody is reduced either by decreasing the antibody coated area (Figure 4.3b), or the density of antibody coating (Figure 4.3c) whilst the field of view remains constant then the signal to noise ratio (S_{sb} / S_{nsb}) ratio falls.


Figure 4.3 Diagram demonstrating changes in field of view, antibody-coated area and coated antibody density. Shading indicates antibody-coated area (intensity of shading represents density of antibody coated onto surface). In (a) and (c) the antibody-coated area is equivalent to the field of view, in (b) the unshaded outer circle represents the field of view. In (a), both the antibody-coated area and the field of view are reduced which leads to increased signal / noise ratio. In (b) the antibody coated area and the field of view remains constant, in this case the signal / noise ratio is decreased. In (c) the antibody coated area and the field of view remains constant, but the coated antibody density decreases, again this results in a decreased signal / noise ratio.

The following calculations and tables demonstrate the effect of the changes in antibody coated area, antibody density and field of view upon the specific/ non-specific signal ratio.

The number of molecules specifically bound is given by;

 $F \times$ number of antibody molecules

The specific/non-specific signal ratio is given by

Fractional occupancy (F)× coated antibody density (molecules/ μ m²) × area (μ m²) non-specifically bound labelled antibody density (molecules/ μ m²) × area (μ m²)

(4.6)

The fractional occupancy (F) is obtained from equation 4.4.

For example if one assumes that as before, an antibody has an affinity constant (K) of 10^{11} M⁻¹, the total incubation volume is 1 ml and the antibody surface density is 6000 molecules / μ m², but one also assumes that the analyte concentration is 0.01K (6 × 10⁷ molecules / ml) and that the labelled second antibody binds non-specifically to the surface at 1 molecule / μ m².

For an area of $1 \times 10^6 \,\mu\text{m}^2$, F = 4.99×10^{-3} (from equation 4.4).

 $4.99 \times 10^{-3} \times 6 \times 10^9$ molecules = 2.994×10^7 molecules contributing to the specific signal. Non-specific binding of the labelled second antibody is 1 molecule / μ m², therefore there are 1×10^6 molecules contributing to the non-specific signal.

Therefore the specific / non-specific signal ratio = $\frac{2.994 \times 10^7}{1 \times 10^6} = 29.94$

Similarly for an area of $1 \times 10^5 \ \mu\text{m}^2$, F =9.02 × 10⁻³, resulting in a specific/non-specific signal ratio of 54.1, and for an area of $1 \times 10^5 \ \mu\text{m}^2$, F =9.80 × 10⁻³, resulting in a specific/non-specific signal ratio of 58.8

If the reduction in the antibody-coated area were not accompanied by a corresponding reduction in the detecting instrument's field of view, there would not be a decrease in the background produced by non-specifically bound labelled antibody, and the specific / non-specific signal ratio would be adversely affected. This is demonstrated by calculated examples shown in Table 4.1. In this table it can be seen that if the field of view is equivalent to the area of antibody deposition, that as the size of this area decreases, the specific / non-specific signal ratio increases. If the area of antibody deposition is increased so that it is larger than the field of view, the specific / non-specific signal ratio remains unchanged, but if the area of antibody deposition is decreased so that it is smaller than the

field of view, the specific / non-specific signal ratio decreases. The implication of this is that if one is working with instrumentation whose field of view can not be decreased below a certain area, it may be beneficial to ensure that the antibody-coated surface covers this entire area, the larger area's advantage of a proportionally smaller signal contribution by non-specifcally bound molecules outweighing the improved fractional occupancy of smaller areas.

| | Specific/Non-specific signal ratio | | | |
|---------------------------------------|------------------------------------|-------------------|-------------------|--|
| Area of antibody deposition μm^2 | 1×10^{6} | 1×10^{5} | 1×10^{4} | |
| Field of view Area μm^2 | | | | |
| 1 × 10 ⁶ | 29.94 | 5.41 | 0.59 | |
| 1 × 10 ⁵ | 54.1 | 54.1 | 5.88 | |
| 1 × 10 ⁴ | 58.8 | 58.8 | 58.8 | |

Table 4.1 Examples of specific / non-specific signal ratios for differently antibody deposition and field of view areas.

In table 4.2 it can be seen that for a given field of view (assumed equivalent to the area of antibody deposition), that as the antibody surface density decreases, so does the specific / non-specific signal ratio. It should also be noted that as the antibody surface density decreases the advantages of using a smaller field of view are greatly diminished.

| | Specific/Non-specific signal ratio | | | |
|---------------------------------------------------------------------|------------------------------------|---------------------|---------------------|--|
| Field of View μm^2 (equivalent to area of antibody deposition) | 1 × 10 ⁶ | 1 × 10 ⁵ | 1 × 10 ⁴ | |
| Antibody surface density molecules/µm ² | | | | |
| 6,000 | 29.94 | 54.1 | 58.8 | |
| 600 | 5.41 | 5.88 | 5.93 | |
| 60 | 0.59 | 0.59 | 0.59 | |

Table 4.2 Examples of specific / non-specific signal ratios for different antibody deposition and antibody surface densities.

4.2.1. Influence of working electrode area as proportion of total area on the specific /non-specific signal ratio.

When measuring analytes in solution amperometrically, the sensor's "field of view" is not restricted to the area on which the sensor antibody is immobilised. The non-specific signal can have a number of constituents in addition to the non-specific binding of the second antibody-enzyme conjugate to the sensing area itself. Signal generating molecules that are bound to the surrounding area also need to be considered, and mass transport of electroactive species between electrodes can occur.

If one was to consider a case where the field of view encompasses the total surface and where diffusion was instantaneous, then the effect (for a 2 electrode system) of an electrode's size (expressed as a fraction of the entire surface) upon the specific signal $(S_{sb})/non-specific signal (S_{nsb})$ ratio is given by equation 4.7. The specific/non signal ratio is dependent upon the electrode area (A_e) , the area of other electrodes within the system (A_o) , the total area (A_t) , the density of signal producing molecules specifically bound to electrode (D_{es}) and the density of signal producing molecules non-specifically bound to the electrode (D_{en}) , other electrodes in the system (D_{on}) and to areas that are not electrodes (D_{nn}) .

$$\frac{S_{sb}}{S_{nsb}} = \frac{(A_e.D_{es})}{(A_e.D_{en} + (A_o.D_{on}) + (A_f - A_e - A_o).D_{nn})}$$
(4.7)

For equation 4.7 to apply for a system for any number (n) of electrodes the term $A_o.D_{on}$ is replaced with the term $\Sigma(A_o.D_{on})_{1.n-1}$. Similarly the term $(A_rA_e-A_o)$ is replaced by $(A_rA_e-\Sigma(A_o)_{1.n-1})$, resulting in equation 4.8.

$$\frac{S_{sb}}{S_{nsb}} = \frac{(A_e.D_{es})}{(A_e.D_{en} + \Sigma(A_o.D_{on})_{1..n-1} + (A_t - A_e - \Sigma(A_o)_{1..n-1}).D_{nn})}$$
(4.8)

The effect of Equation 4.8 is that as the electrode size (expressed as a fraction of total area) decreases there is a corresponding reduction in the specific/non-specific signal ratio. This is shown graphically in Figure 4.4, for a number of simulated examples.





If a stationary electrode is immersed in an unstirred electrolytic solution and a potential is applied, provided that electroactive substances that can undergo oxidation or reduction are present in the electrolytic solution, charge transfer will occur at the electrode/liquid interface. Reduction involves electron transfer from the electrode to the electroactive substance, whilst in oxidation electrons are transferred from the electroactive substance to the electrode. As this occurs, concentrations of substances at the electrode surface start to change, and diffusive mass transport commences. Dependent upon the size of the electrode and the volume of the solution three limiting cases of diffusion can be described.²⁹⁸ The first case is an electrode in a thin-layer cell where the cell volume/ electrode surface area ratio is very low. In this case mass transport within the cell is negligible, and there is no diffusion gradient. In the second case, that of a macroelectrode (not in a thin layer cell), the cell volume/electrode surface area ratio is much larger and one observes planar diffusion. In the final case when one reduces the size of the electrode so that it has an extremely small surface area (diameter < 20 μ m), the conditions change and the diffusion process becomes dependent upon the size and geometry of the electrode. For disk ultramicroelectrodes the spatial diffusion field that develops is hemispherical. These different types of diffusion are illustrated in Figure 4.5.



Lateral Diffusion Insignificant

Figure 4.5 Transverse sections of a thin layer cell, microelectrode and a macroelectrode demonstrating differences in diffusion. In the thin layer cell the available transport paths are short and no diffusion gradient develops. In the microelectrode a spatial diffusion field develops (in the case of a disk electrode a hemispherical field). For the macroelectrode, planar diffusion occurs.²⁹⁸

Predominant

As mentioned in Chapter 3, mass transport of electroactive species consists of three components, diffusion, migration and convection. Mass transport can be limited to just the diffusion term if the solution is quiescent (eliminating convection) and has a relatively high concentration of supporting electrolyte (eliminating migration).

As one decreases the size of an electrode, to the point of it becoming an

ultramicroelectrode, then time-dependent changes in mass transport become evident. These changes involve the transformation of one-dimensional planar diffusion fields into spatial fields, with a radial diffusive component (parallel to the electrode surface) being added to the usual axial diffusion. In mathematical terms diffusion is described by Fick's Laws, which vary according to electrode geometry.²⁹⁸ The equation for planar diffusion at a macroelectrode is shown in Equation 4.9, whilst that for hemispherical diffusion at a disk electrode is shown in Equation 4.10.

$$\frac{\delta c}{\delta t} = D \frac{\delta^2 c}{\delta x^2} (4.9)$$

$$\frac{\delta c}{\delta t} = D(\frac{\delta^2 c}{\delta r^2} + \frac{1}{r} \frac{\delta c}{\delta r} + \frac{\delta^2 c}{\delta z^2}) \quad (4.10)$$

The increased diffusion rate of species to ultramicroelectrodes leads to a number of analytical advantages such as very high current densities (in spite of small currents), rapidly achieving a time independent stationary state, and a large drop in ohmic resistance due to the resistance of the solution.²⁹⁸⁻³⁰⁰ However, in constructing arrays of ultramicroelectrodes for multianalyte analysis, the hemispherical diffusion field can be problematical. The contribution to the observed current due to signal producing molecules that are non-specifically bound to areas other than the electrode (terms D_{on} and D_{nn} in equations 4.7 and 4.8) are greatly increased, resulting in a poorer specific/non-specific signal ratio. This in turn will lead to poorer sensitivity and reduced discrimination.

The formation of the hemispherical diffusion field effectively means that the field of view of the detecting instrument is not limited to the area on which the sensor antibody is deposited and the specific / non-specific signal ratio (from equation 4.6) would be adversely affected, as shown in Table 4.1.

In Table 4.1 it can be seen that if the field of view is equivalent to the area of antibody deposition, that as the size of this area decreases, the specific / non-specific signal ratio increases. If the area of antibody deposition is increased so that it is larger than the field of view, the specific / non-specific signal ratio remains unchanged, but if the area of antibody deposition is decreased so that it is smaller than the field of view, the specific / non-specific signal ratio decreases. As stated earlier, the implication of this is that if one is working with instrumentation whose field of view can not be decreased below a certain area (such as hemispherical diffusion at ultramicroelectrodes), it may be beneficial to ensure that the antibody-coated surface covers this entire area, the larger area's advantage of a proportionally smaller signal contribution by non-specifically bound molecules outweighing the improved fractional occupancy of smaller areas.

4.3 Materials

Positive S1400-31 photoresist and Microposit developer were obtained from Shipley Europe, Coventry, England. Glucose oxidase (EC 1.1.3.4, 20 U mg⁻¹), sodium glutamate, ferrocenemonocarboxylic acid, N-acetyl-l-cysteine, and1-ethyl-3-(3-dimethylaminopropyl) - carbodiimide were obtained from Sigma, Poole, England.

Glutamate oxidase (EC 1.4.3.11.5, 16 U mg⁻¹) was kindly provided by Yamasa Shoyu, Chiba, Japan.

Silicon wafers were obtained from Shin Etsu, Livingston, Scotland. NiChrome (80:20 Ni:Cr) and gold were obtained from Goodfellow Cambridge Ltd, Cambridge, England. Neutravidin[™] and casein were obtained from Pierce Warriner, Chester, England. Photobiotin was obtained from Vector Laboratories, Peterborough, England.

A Bioanalytical Systems (BAS) RE4 Ag/AgCl reference electrode was obtained from Biotech Instruments Ltd, Luton, England.

Instrumentation used comprised a EG&G 273A potentiostat (EG & G, Sunninghill, England), Bioanalytical System CV-37 potentiostats (Biotech Instruments Ltd, Luton, England), a Goerz SE120 dual channel chart recorder (Belmont Instruments, Glasgow, UK), a 100W HB-10101AF super high pressure mercury vapour lamp (Nikon, Tokyo, Japan) and a photolithographic mask aligner (HTG San Jose, California, USA).

4.4. Methods

4.4.1 Method for fabrication of electrodes and electrode arrays using photolithographic lift-off techniques.

Positive S1400-31 photoresist was spun onto 10 cm diameter silicon wafers at 4000 rpm for 30 s and then baked at 90°C for 30 minutes. The resist was then illuminated by light for 12 s from a mask aligner through a photolithographic mask containing relief images of the electrode arrays (Figures 4.6 to 4.8). Exposed resist was removed by immersion in Microposit developer for 75 s followed by a rinse in reverse osmosis (RO) water. NiChrome and gold were deposited on the surface to a thickness of 10nm and 100nm respectively, using a modified Edwards 12E High Vacuum evaporation unit. The remaining photoresist was removed by immersion in acetone for 20 minutes, lifting off the metal deposited on it, leaving the silicon wafer with the metallic electrode arrays on its surface. S1400-31 photoresist was spun onto the surface again, and baked at 90°C as detailed above. For the electrodes shown in Figures 4.6 and 4.7, all areas of the surface apart from the connecting tracks were exposed to light through a second mask (this was so that the connecting track did not form part of the electrode, and to allow investigations into the effect of specifically and non-specifically bound enzymes to the area surrounding the electrode). For the electrodes shown in Figures 4.8, electrodes and bonding pads (but not connecting tracks) were exposed to light through a second mask. Exposed resist was removed as detailed above, and the wafer was then baked at 120°C for 120 minutes, to harden the remaining photoresist that acted as an electrical insulator.

169



Figure 4.6 Diagrammatic representation of a photolithographically fabricated gold electrode with an electrode 2 mm in diameter and surrounded by a gold non-electrode surface. Details of the fabrication procedure are provided in the text (Section 4.4.1)



Figure 4.7 Diagrammatic representation of an photolithographically fabricated gold electrode with an electrode 10 µm in diameter and surrounded by a gold non-electrode surface. Details of the fabrication procedure are provided in the text (Section 4.4.1)



Figure 4.8 Diagrammatic representation of a photolithographically fabricated electrode array with electrodes either 1 mm or 10 µm in diameter. Details of the fabrication procedure are provided in the text (Section 4.4.1)

4.4.2. Avidin-photobiotin immobilisation procedure.

Electrodes (fabricated as shown in Figures 4.6 to 4.8) were incubated in 2 mM NAC in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature, followed by 120 minutes incubation in 1% (w/v) EDC in 10 mM phosphate buffer (pH 7.0). The electrodes were then incubated in 100 μ g ml⁻¹ Neutravidin in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C. All subsequent stages of the immobilisation procedure were performed at room temperature. After washing in phosphate buffered saline, pH 7.4 (PBS), the electrodes were first incubated in 10 mg ml⁻¹ casein in PBS for 60 minutes and then in 10 μ g ml⁻¹ long arm photobiotin in PBS, for 20 minutes in the dark.

4.4.3. Amperometric response of electrodes.

Amperometric enzyme based assays were performed using the gold electrode as a working electrode, with an Ag/AgCl electrode as the reference and a bare platinum flag as a counter electrode. Chronoamperometry was performed in a working solution containing 10 mM sodium phosphate, 140 mM NaCl, 25 mM KCl, pH 7.0. Dependent upon the experiment, glucose and / or glutamate could be added to this working solution to provide predetermined concentrations. Initially, the working electrode was poised at a potential of 0V for 10 seconds, after which the potential was "stepped" to 650mV for 120 seconds at which time the current was recorded.

4.4.4 Investigation into effect of hemispherical diffusion field upon response of ultramicroelectrodes.

4.4.4.1 Enzyme immobilisation procedure

Avidin and photobiotin were immobilised on electrodes (as shown in Figures 4.6 and 4.7) as described in Section 4.4.2. After washing in PBS the wafer was covered with $10 \ \mu g \ ml^{-1}$

glucose oxidase (GOD) and the desired areas (as shown in Table 4.3) were exposed to light as described in Section 2.3.1. After washing in PBS, the entire wafer was exposed to light for 15 minutes in the presence of 10 mg ml⁻¹ casein in PBS, and again washed in PBS. Areas of electrodes (as shown in Table 4.3) were then exposed to deep UV radiation ($\lambda <$ 200 nm, irradiance = 200 mW cm⁻²) in order to inactivate any GOD that had been non-specifically bound. The areas containing specifically and non-specifically bound GOD are shown in Table 4.4. After the enzyme immobilisation procedure was complete a plastic ring was attached to the substrate to allow solutions to be placed in the cell. The prepared electrodes were stored in PBS until use.

| Electrode | Area exposed to light in presence of GOD during patterning process | Area exposed to DUV following patterning process. |
|-----------|-----------------------------------------------------------------------|---------------------------------------------------|
| A | None | Entire surface |
| В | None | Entire surface apart from electrode |
| С | None | Electrode only |
| D | None | None |
| E | Electrode only | Entire surface apart from electrode |
| F | Electrode only | None |
| G | Entire surface apart from electrode | Electrode only |
| Н | Entire surface apart from electrode | None |
| Ι | Entire surface | None |

<u>Table 4.3 Areas of sensor surfaces exposed to light in presence of GOD during patterning process and those</u> <u>areas exposed to DUV radiation following the patterning process</u>. Exposure to light during the patterning process resulted in specific immobilisation of GOD, whilst in areas irradiated with DUV light the GOD was rendered non-functional.

| | | Area with non-specifically bound GOD | | | | |
|----------------------------------------|----------------------|--------------------------------------|-----------|----------------------|-----|--|
| | | None | Electrode | Other than electrode | All | |
| Area with specifically bound GOD | None | A | В | С | D | |
| | Electrode | - | E | - | F | |
| | Other than electrode | - | - | G | Н | |
| | All | - | - | - | I | |

Table 4.4 Areas of sensor surfaces containing specifically and non-specifically bound GOD. Note that it was not possible to have an area with specifically bound GOD in the absence of non-specifically bound GOD.

4.4.4.2. Amperometric Analysis

Chronoamperometry was performed as described in Section 4.4.3 using solutions containing 100 mM glucose. Amperometric analysis was performed using 2mm and 10 μ m diameter electrodes (Figures 4.6 & 4.7) with a +650 mV potential (versus Ag/AgCl) applied to the electrode alone, and with a potential of +650 mV also applied to the surrounding gold area from a second independent potentiostat. Results are shown in Section 4.5.1.

<u>4.4.5 Investigation into analytical "cross-talk" caused by mass transport of electrochemical</u> <u>species generated at one electrode to other electrodes in an array.</u>

4.4.5.1 Enzyme immobilisation procedure

Avidin and photobiotin were immobilised on electrodes (as shown in Figure 4.8) as described in Section 4.4.2. After washing in PBS the wafer was covered with $10 \ \mu g \ ml^{-1}$ GOD and the generator electrode (electrode 1 in Figure 4.8) was exposed to light and then blocked with casein as described in Section 4.4.4.1. Areas other than the generator electrode were exposed to DUV radiation, fitted with a plastic ring and stored as described in Section 4.4.4.1.

4.4.5.2 Amperometric analysis

Chronoamperometry was performed as described in Section 4.4.3. A series of experiments were performed, applying a potential of +650 mV versus the Ag/AgCl reference electrode to each of the electrodes in turn. Upon obtaining a stable baseline reading, glucose was added to give a final concentration of 100 mM. The current was recorded 120 seconds after the addition of the glucose. This series of experiments were then repeated, applying a potential of 650mV to the generator electrode as well as the monitoring electrode. Plots of current versus distance of measuring electrode from generator electrode are shown in Section 4.5.2.

4.4.6 Construction of multianalyte enzymic biosensor using ultramicroelectrodes.

The analytes were chosen were glucose and glutamate as oxidase enzymes (whose action resulted in the formation of H_2O_2) were available for these 2 substances.

glutamate oxidase Glutamate + $O_2 \longrightarrow 2$ -Oxoglutarate + $NH_3 + H_2O_2$

4.4.6.1 Electrode fabrication.

Electrodes were photolithographically fabricated as described in Section 4.4.1. The fabricated electrodes were 10 μ m in diameter and located 150 μ m apart. Tracks 3 μ m wide (which were electrically insulated by the use of photoresist) connected the electrodes to bonding pads, where connections were made to the potentiostats.

4.4.6.2 Enzyme immobilisation procedure.

Avidin and photobiotin were immobilised on electrodes as described in Section 4.4.2. After washing in PBS the wafer was covered with $10 \ \mu g \ ml^{-1}$ glutamate oxidase (GLOD) for 15 minutes and one of the electrodes was exposed to light as described in Section 4.4.4.1. After washing in PBS the wafer was covered with $10 \ \mu g \ ml^{-1}$ GOD and the other electrode was exposed to light 15 minutes. The surface was then blocked with casein as described in Section 4.4.4.1. After section 4.4.4.1. Areas of the surface other than the electrodes were exposed to DUV radiation, fitted with a plastic ring and stored as described in Section 4.4.4.1.

4.4.6.3 Amperometric analysis.

Chronoamperometry was performed as described in Section 4.4.3. with solutions containing varying concentrations of glucose or glutamate. Results are shown in Section 4.5.3.

4.4.7 Construction of multianalyte immunosensor using ultramicroelectrodes.

4.4.7.1 Electrode Fabrication

Electrodes were photolithographically fabricated as described in Section 4.4.1. The fabricated electrodes were 10 μ m in diameter and were fabricated in pairs situated 1mm apart (centre to centre), individual electrodes were as shown in Figure 4.7. Tracks 3 μ m wide(which were electrically insulated by the use of photoresist) connected the electrodes to bonding pads where attachments were made to the potentiostats.

4.4.7.2 Antibody immobilisation

Avidin and photobiotin were immobilised on electrodes as described in Section 4.4.2. After washing in PBS the wafer was covered with $10 \ \mu g \ ml^{-1}$ goat anti-rabbit IgG for 15 minutes and one of the electrodes was exposed to light as described in Section 4.4.4.1. After

washing in PBS the wafer was covered with 10 μ g ml⁻¹ goat anti-rat IgG and the other electrode was exposed to light 15 minutes The surface was then blocked with casein as described in Section 4.4.4.1. Areas of the surface other than the electrodes were exposed to DUV radiation, fitted with a plastic ring and stored as described in Section 4.4.4.1.

Antibodies were also immobilised to electrodes in the way described above, but instead of just the electrodes being exposed to light during the patterning process, the area of patterned gold surrounding the electrodes (Figure 4.7) was also exposed, thus increasing the area of specifically bound antibody.

4.4.7.3 Immunoassay procedure.

Electrodes were incubated in solutions of PBS or PBS containing 10 μ g ml⁻¹ rat IgG and / or 10 μ g ml⁻¹ rabbit IgG for 60 minutes at room temperature. Following washing with PBS, electrodes were incubated in 10 μ g ml⁻¹ goat anti-rat IgG-HRP conjugate and 10 μ g ml⁻¹ goat anti-rabbit IgG-HRP conjugate for 60 minutes at room temperature

Simultaneous assessment of HRP activity was performed chronoamperometrically using two Bioanalytical System CV-37 potentiostats and a Goerz SE120 dual channel chart recorder. Activity was determined by measuring the current produced at 120 s in the presence of 10 mM hydrogen peroxide, and 0.2 mM ferrocenemonocarboxylic acid in 50 mM phosphate buffer containing 50 mM KCl, pH 7.4, with a potential of +150 mV applied to the working electrode versus the Ag/AgCl reference electrode.²⁹⁰

176

4.5 Results and Discussion

4.5.1 Investigation into effect of hemispherical diffusion field upon response of ultramicroelectrodes.

The results shown in Table 4.5 demonstrate that the observed current for 2 mm diameter electrodes was predominantly a result of electroactive species being generated at the electrode surface. The currents observed when GOD has been specifically immobilised on the electrode were significantly higher than those where GOD had been specifically immobilised on the surrounding gold but not on the electrode itself, and for those where only non-specific binding was present. The currents observed where GOD had been specifically immobilised on the surrounding gold but not on the electrode itself, were slightly higher than those where only non-specific binding was present. The currents observed where GOD had been specifically immobilised on the surrounding gold but not on the electrode itself, were slightly higher than those where only non-specific binding was present, indicating that some diffusion to the electrode of H_2O_2 generated in the surrounding area was occurring. These results also indicate that the effect of non-specifically bound GOD on observed currents is small.

The results shown in Table 4.6 demonstrate that for 10 μ m diameter electrodes non-specific binding of GOD to the gold surface surrounding the electrode can produce a very significant current at the working electrode. This is as the H₂O₂ produced in this surrounding area diffuses to the working electrode, and this non-specific signal is similar in magnitude to that obtained when GOD is specifically immobilised at the working electrode. These results indicate that when using ultramicroelectrode arrays for multianalyte assays the effect of non-specific binding (in particular to the surrounding surface) can be critical.

The results in Tables 4.5 and 4.7 indicate that if in the presence of glucose, a potential of +650 mV vs. Ag/AgCl was applied to the area of gold surrounding the 2 mm diameter

electrode as well as to the electrode itself, only a small decrease in current from the working electrode was observed compared with when the potential was applied to the electrode alone. This was as the diffusion at the 2 mm diameter electrode is predominantly planar with only a relatively small lateral component, which is further demonstrated by the relatively small current observed when there is no functional GOD deposited on the electrode itself, but functional GOD is present on the surrounding gold surface.

The results in Tables 4.6 and 4.8 indicate that if in the presence of glucose, a potential of +650 mV vs. Ag/AgCl was applied to the area of gold surrounding the 10 μ m diameter electrode as well as to the electrode itself, the resultant current from the working electrode was, in most cases, indistinguishable from the signal when no glucose was present. This was because the surrounding gold was acting as a macroelectrode and eliminated the hemispherical diffusion field surrounding the ultramicroelectrode. The only exception was that where GOD was specifically immobilised on the electrode, and any non-specifically bound GOD in the surrounding area was rendered non-functional by DUV irradiation. In this case a hemispherical diffusion of glucose to the electrode can occur.

These results suggest that for multianalyte sensors employing ultramicroelectrodes one should specifically immobilise the biological element over the surface surrounding the electrode as well as on the electrode itself. This is so that the specific signal is being generated over the entire "field of view" (i.e. the hemispherical diffusion field), thus improving the specific/nonspecific signal ratio as discussed earlier in this chapter by increasing S_{sb} in Equations 4.7 and 4.8. An alternative is to render any molecules that are non-specifically bound to the area surrounding the electrode non-functional by techniques such as DUV irradiation, and thus decreasing S_{nsb} in Equations 4.7 and 4.8.

| - | | Current nA | | | |
|--------------|----------------------|--------------------------------------|-----------|----------------------|-----|
| | | Area with non-specifically bound GOD | | | GOD |
| | | None | Electrode | Other than electrode | All |
| Area with | None | 63 | 76 | 78 | 84 |
| specifically | Electrode | - | 358 | - | 346 |
| GOD | Other than electrode | _ | - | 118 | 122 |
| | All | - | - | - | 347 |

Table 4.5 Currents obtained from 2 mm diameter electrodes in the presence of glucose when no potential was applied to the surrounding gold surface. Chronoamperometry was performed as described in Section 4.4.4.2.

| | | Current pA Area with non-specifically bound GOD | | | |
|--------------|----------------------|----------------------------------------------------|-----------|----------------------|-----|
| | | | | | GOD |
| | | None | Electrode | Other than electrode | All |
| Area with | None | 23 | 47 | 102 | 99 |
| specifically | Electrode | - | 243 | - | 265 |
| GOD | Other than electrode | - | - | 233 | 238 |
| | All | * | - | - | 258 |

Table 4.6 Currents obtained from 10 µm diameter electrodes in the presence of glucose when no potential was applied to the surrounding gold surface. Chronoamperometry was performed as described in Section 4.4.4.2.

| | | Current nA | | | |
|--------------|----------------------|--------------------------------------|-----------|----------------------|-----|
| | | Area with non-specifically bound GOD | | | GOD |
| | | None | Electrode | Other than electrode | All |
| Area with | None | 54 | 69 | 57 | 67 |
| specifically | Electrode | - | 338 | - | 341 |
| GOD | Other than electrode | | - | 71 | 77 |
| | All | - | - | - | 329 |

Table 4.7 Currents obtained from the 2 mm diameter electrodes in the presence of glucose when a potential of +650 mV was applied to the surrounding gold surface as well as to the working electrode.

Chronoamperometry was performed as described in Section 4.4.4.2.

| | | Current pA Area with non-specifically bound GOD | | | |
|--------------|----------------------|----------------------------------------------------|-----------|----------------------|-----|
| | | | | | GOD |
| | | None | Electrode | Other than electrode | All |
| Area with | None | 19 | 23 | 18 | 22 |
| specifically | Electrode | - | 198 | - | 23 |
| GOD | Other than electrode | | - | 30 | 22 |
| | All | | - | - | 31 |

Table 4.8 Currents obtained from the 10 μ m diameter electrodes in the presence of glucose when a potential of +650 mV was applied to the surrounding gold surface as well as to the working electrode. Chronoamperometry was performed as described in Section 4.4.4.2.

4.5.2 Investigation into analytical "cross-talk" caused by mass transport of electrochemical species generated at one electrode to other electrodes in an array.

The data presented in Figures 4.9 and 4.10 indicates that unless electrodes are in very close proximity the contribution of electroactive species generated at one electrode to the observed response at another is minimal. It should be noted that distances in these Figures are expressed from the centre of one electrode to the centre of another. The contribution of electroactive species generated at one electrode to the observed response at another is more pronounced for the 10 μ m electrodes than for the 1 mm electrodes, which reflects the increased importance of lateral diffusion for ultramicroelectrodes. That lateral diffusion is less significant for larger electrodes, is also shown by the greater proportional change in current for ultramicroelectrodes, when the generator as well as the measuring electrode is held at a potential of +650 mV vs. Ag/AgCl.



Figure 4.9 Current (background subtracted) as a function of electrode distance from the generator electrode for 1 mm diameter electrodes. Experimental details are provided in the text (Section 4.4.5)



Figure 4.10 Current (background subtracted) as a function of electrode distance from the generator electrode for 10 µm diameter electrodes. Experimental details are provided in the text (Section 4.4.5)

Equations 4.7 and 4.8 show that the contribution to the non-specific signal (S_{nsb}), is dependent upon the relative area and number of other electrodes within an array. The effects are cumulative and care should be taken in the design of multianalyte electrode arrays to locate electrodes so that lateral diffusion of electrochemical species between them is negligible, particularly when using ultramicroelectrodes, where lateral diffusion predominates. The hemispherical diffusion field associated with ultramicroelectrodes is a limiting factor for how closely electrodes can be placed together in a multianalyte system.

4.5.3 Construction of multianalyte enzymic biosensor using ultramicroelectrodes.

The results shown in Figures 4.11 and 4.12 demonstrate that a combination of the avidin-photobiotin immobilisation technique and DUV irradiation of non-electrode areas can be used to construct a multianalyte biosensor using ultramicroelectrodes. The avidin-photobiotin immobilisation procedure specifically immobilised enzymes on the electrodes, and any molecules that had non-specifically bound to non-electrode surfaces were inactivated by the use of DUV irradiation.

There was minimal response of the glucose sensor to the prescence of glutamate in solution and *vice versa*. Based on the size of electrodes and distances them used in this work it would be possible to construct a sensor 1mm^2 capable of measuring at least 25 discrete analytes. If the distances between the electrodes were reduced this would allow significantly more sensor elements to be fitted into the same area, for example data shown in Figure 4.11 shows minimal"cross-talk" between electrodes 50 µm apart. If electrodes were located 50 µm apart, then theoretically one could fit over 350 sensors in a 1mm^2 area, each sensor responding to a different analyte.



Figure 4.11 Response of the glucose oxidase electrode to solutions containing glucose and glutamate. GOD and GLOD were specifically immobilised on discrete 10 µm diameter ultramicroelectrodes using the avidin-photobiotin method. An amperometric enzyme based assay was performed using the modified gold surface as a working electrode, with an Ag/AgCl electrode as the reference. Chronoamperometry was performed as described in the text.



Figure 4.12 Response of the glutamate oxidase electrode to solutions containing glucose and glutamate. GOD and GLOD were specifically immobilised on discrete 10 µm diameter ultramicroelectrodes using the avidin-photobiotin method. An amperometric enzyme based assay was performed using the modified gold surface as a working electrode, with an Ag/AgCl electrode as the reference. Chronoamperometry was performed as described in the text.

4.5.4 Construction of multianalyte immunosensor using ultramicroelectrodes.

The results in Table 4.9 show that when antibodies were specifically immobilised on the ultramicroelectrode alone, no significant difference in current could be observed when antigen was or was not present. This reflects insufficient signal generating molecules (i.e. conjugate) present in the small area of the electrode surface to produce a current differentiable from that of the background.

The results in Table 4.10 show that when antibodies were specifically immobilised on the area surrounding the electrode as well as the ultramicroelectrode itself, significant differences in current could be observed when antigen was or was not present. The difference between this case and that when antibody was immobilised on the electrode alone, is due to a greatly increased amount of conjugate being bound, with a related increase in signal generation.

| | Current pA | | | |
|-----------------------------------------------------------------------|-----------------------------------------|--------------------------------------------|--|--|
| Sample | Electrode with anti-rat IgG immobilised | Electrode with anti-rabbit IgG immobilised | | |
| PBS | 25 | 22 | | |
| 10 μg ml ⁻¹ rat IgG | 31 | 25 | | |
| 10 µg ml ⁻¹ rabbit IgG | 26 | 28 | | |
| 10 μg ml ⁻¹ rat IgG + 10 μg ml ⁻¹ rabbit IgG | 29 | 28 | | |

Table 4.9 Currents obtained in presence of antigens for ultramicroelectrode immunoassay when antibody was immobilised only on electrode. Experimental details are provided in the text (Section 4.4.7)

| | Current pA | | | | |
|-----------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------|--|--|--|
| Sample | Electrode with anti-rat IgG immobilised | Electrode with anti-rabbit IgG immobilised | | | |
| PBS | 23 | 24 | | | |
| 10 μg ml ⁻¹ rat IgG | 107 | 31 | | | |
| 10 μg ml ⁻¹ rabbit IgG | 28 | 94 | | | |
| 10 μg ml ⁻¹ rat IgG + 10 μg ml ⁻¹ rabbit IgG | 124 | 115 | | | |

Table 4.10 Currents obtained in presence of antigens for ultramicroelectrode immunoassay when antibody was immobilised on surrounding area as well as on electrode. Experimental details are provided in the text (Section 4.4.7)

The qualitative immunosensor fabricated in this section of work demonstrates the feasibility of constructing multianalyte amperometric immunosensors using ultramicroelectrodes. The work presented in this chapter shows that when using ultramicroelectrodes in multianalyte assays, it is necessary take into account the field of view of the system (defined by the area of hemispherical diffusion). The field of view effects can be limited by destroying the activity of any biological molecules non-specifically bound to the surrounding area (e.g. by DUV irradiation). This provides increased sensitivity and specificity by decreasing the non-specific signal (S_{nsb} in Equations 4.7 and 4.8). Alternatively one can increase the specific signal (S_{sb} in Equations 4.7 and 4.8) by extending the area of specifically immobilised molecules to that of the area surrounding the electrode.

CHAPTER 5

And in such indexes, although small pricks

To their subsequent volumes, there is seen

The baby Figure of the giant mass

of things to come at large

William Shakespeare

Troilus and Cressida

5 Further applications of avidin-photobiotin immobilisation procedure

This chapter examines the extension of the avidin-photobiotin immobilisation technique to applications other than protein based biosensors. Initially, work was performed to demonstrate the patterning of DNA on silicon dioxide, following on from this, the avidin-photobiotin patterning procedure was used to immobilise oligonucleotides as the basis of an assay to detect a genetic polymorphism. Additionally, since the publication of the avidin-photobiotin immobilisation technique,¹⁹⁷ it (or variants thereof) has been employed by a number of other workers, and this work, along with some future prospects, are briefly reviewed in this chapter.

5.1 Patterning of nucleic acids

Patterning of different nucleic acids or oligonucleotides on to sensor surfaces could be useful in the construction of biosensors for the detection of a number of genetic diseases or polymorphisms. An example where this application may prove useful is in the assessment of thrombotic risk. There are several polymorphisms, associated with an increased thrombotic risk, of particular interest are those for Factor V_{Leiden} , Factor II_{Leiden} and methylene- tetrahydrofolate reductase (MTHFR).

Factor V is one of the components of the coagulation cascade (Figure 5.1) and acts as a cofactor for Factor X. Factor V is converted to its active form by the action of thrombin, and subsequently inactivated by the action of activated protein C (APC). It has been noted that in some people who are prone to thrombosis, that their Factor V is resistant to cleavage by APC.³⁰¹ It has subsequently been demonstrated that the abnormality is caused by the

replacement of arginine by glutamine at residue 506,³⁰² which is a major cleavage site for APC, and thus the mutation is associated with a prolonged plasma half-life of activated Factor V. The possession of this variant commonly referred to as Factor V_{Leiden}, is associated with an increased risk of thrombosis and is an important risk factor in populations of European descent, where it has an incidence of approximately 7%.³⁰² In heterozygotes the risk of thrombosis is increased approximately 7-fold, whereas homozygotes have about 80 times more chance of having a thrombotic episode as compared to someone with the normal Factor V gene.³⁰²

Similarly a mutation with the Factor II (prothrombin) gene is associated with increased risk of thrombosis.³⁰³ Like Factor V, Factor II is a component of the coagulation cascade (Figure 5.1), and the variant protein is frequently referred to as Factor II_{Leiden} This mutation is present in about 1% of the European population and is associated with elevated prothrombin levels. It is thought that these elevated levels disturb the balance of the coagulation system and result in a tendency for inappropriate clot formation to occur.³⁰³

Homocysteine is a sulphur containing amino acid, that although itself not a building block for proteins, lies at an important metabolic branch point in the metabolism of methionine and cysteine as shown in Figure 5.2.³⁰⁴ A mutation in MTHFR (the enzyme responsible for converting homocysteine back to methionine) is thought to be a frequent and significant cause for mild hyperhomocysteinaemia.³⁰⁵ A number of large patient-control studies have shown that even mild hyperhomocysteinamia is a risk factor for thrombosis.^{306,307}



Figure 5.1 Simplified version of the coagulation cascade. Arrows in green are those catalysing conversion of clotting factors and promoting procoagulant activity. Arrows in red are inhibitory, converting an activated factor to an inactive form and thus promoting anticoagulant activity.



Figure 5.2 Biochemical pathways of the conversion of homocysteine to methione and cysteine.

What is particularly important about the above mentioned mutations predisposing to thrombosis, is that the risk increases cumulatively according to the number of the mutations that an individual has. For example, if an individual is heterozygotic for Factor V_{Leiden} then the risk of thrombosis is increased approximately 7-fold; if the individual is also heterozygotic for Factor II_{Leiden}, then the thrombotic risk is increased approximately 30-fold; and if the individual also has the MTHFR variant, then they are about 100 times more likely to have a thrombotic event.³⁰⁸ This cumulative risk make it relevant to measure these analytes in a multianalyte system.

5.1.1 DNA amplification

In order to obtain sufficient quantities of DNA from tissue samples for genetic analysis, it is usually necessary to use a DNA amplification procedure, the most commonly used is the polymerase chain reaction (PCR). PCR relies on multiple amplifications between two oligonucleotide primers complementary to the anti-parallel DNA strands using a directionally specific DNA polymerase.^{309,310} The principle of the polymerase chain reaction is shown in Figure 5.3.

PCR utilises the ability of an oligonucleotide "primer" to target a specific nucleotide sequence on denatured DNA and form an initiation site for DNA polymerase to synthesize a copy of the template. Primers are selected to be complementary to opposite strands of the DNA, and so that the sequence to be detected is present within the section between them. The action of the polymerase results in the formation of two additional DNA strands containing the primers at the 5' end. After denaturation, 4 strands of DNA are available for annealing of primers and subsequent polymerisation. The three steps of the reaction;



Figure 5.3 Principle of the polymerase chain reaction method of amplifying DNA. a), Sample DNA is denatured so that the complementary strands separate. b) An oligonucleotide primer anneals to the denatured DNA and c) nucleotides are incorporated into a new DNA chain by the action the action of Taq polymerase (elongation). d) A new cycle begins with denaturation of sample and newly synthesised DNA, and primers annealing to the DNA strands followed by e) elongation. The process is repeated for further cycles, with a potential doubling of DNA each time.

denaturation, annealing and polymerisation (elongation) are generally performed at three different temperatures (denaturation ~ 94°C, annealing ~ 55°C and elongation ~ 72°C). The exact temperatures and time that each temperature is maintained are dictated by factors such as sample type and the length of DNA to be amplified. These steps are repeated with a potential doubling of the amount of DNA with each cycle.

Reactants in a typical PCR are the sample DNA, each deoxynucleotide triphosphate (A, T, C, G), oligonucleotide primers and the DNA polymerase. The DNA polymerase used in PCR is derived from the thermophilic bacteria *Thermus aquaticus*, and thus allows the enzyme to remain active at the required elevated temperatures (the enzyme usually being referred to as Taq polymerase). Theoretically, the reaction is capable of amplifying DNA by a factor of 2ⁿ where n equals the number of cycles, so 20 cycles could result in a million-fold amplification. In practice, as the efficiency of the reaction is less than 100 %, 20 cycles would typically result in an amplification by a factor of about 250,000.

5.1.2 Principle of assay for detection of Factor V_{leiden} genotype

Oligonucleotide primers were immobilised using the avidin-photobiotin technique and in-situ polymerase chain reaction amplification and detection of DNA was undertaken.

The principle of the assay is shown in Figure 5.4. Wells of a PCR microtitre plate had one of two oligonucleotide primers attached using the avidin-photobiotin immobilisation technique. Primer 1 which was complementary to the normal Factor V genotype was immobilised in 50% of the wells whilst Primer 2 whichwa complementary to the Factor V_{Leiden} genotype was immobilised in the others. Extracted human DNA was added to the wells (each sample being added to 2 wells, one containing primer 1 and the other containing

primer 2) along with the other reactants (as described in Section 5.1), and the plate was placed in a thermal cycler.

The DNA was denatured and bound to the immobilised specific Factor V genoytype primer(s) with which it had complementary sequences. The sample DNA also bound to the biotinylated primer that is common to DNA for both genotypes (Figure 5.4a). Taq polymerase acted to elongate the primer chain by incorporation of nucleotides using the sample DNA as template (Figure 5.4b). With a new cycle, primers bound to the newly synthesised DNA chains (as well as to original DNA) and amplification continued (Figure 5.4c) until biotinylated primers were attached to all of the immobilised primers via newly synthesised DNA chains (Figure 5.4d). Streptavidin-HRP conjugate was added which attached to biotin (Figure 5.4e). Substrate and chromaphore were added and colour development indicated the presence of a sample DNA sequence complementary to that of the immobilised primer.

The combination of results from wells containing the 2 different primers gave the Factor V genotype. If colour development was seen in only wells with primer 1 immobilised, then the individual had the normal Factor V. Colour development in both wells indicated that the individual was heterozygotic whereas colour development in only wells with primer 2 immobilised indicated a homozygotic Factor V_{leiden} .

193



<u>Figure 5.4 Principle of the assay for identification of Factor V genotype.</u> a) Sample DNA binds to the immobilised specific Factor V genoytype primer if its sequences are complementary. Sample DNA also bind the biotinylated primer that is common to DNA for both genotypes. b) Elongation of primers occurs using sample DNA as template. c) With a new cycle primers bind to the newly synthesised DNA chains (as well as to original DNA) and amplification continues until d) biotinylated primers are attached to all of the immobilised primers via newly synthesised DNA chains. e). Streptavidin-HRP conjugate is added which attaches to biotin and allows colour development to occur thus demonstrating presence of genotype.

5.2 Materials

Photobiotin and biotinylated DNA were obtained from Vector Laboratories, Peterborough, England. Sodium cyanoborohydride, streptavidin-FITC, Taq polymerase, dNTPs and Streptavidin-horseradish peroxidase were obtained from Sigma, Fancy Road, Poole, Dorset, England. Neutravidin and casein were obtained from Pierce Warriner, Chester, England. Oligonucelotide primers were obtained from Oswel DNA Service, Southampton, England. 1,3-trimethoxysilylpropyl- ethylene diamine, ethanol, sodium dihydrogen phosphate, disodium hydrogen phosphate, sodium chloride, glutaraldehyde, magnesium chloride and hydrogen peroxide were obtained from BDH, Poole, Dorset, England. TMB substrate was obtained from Shield Diagnostics, Dundee, Scotland

Extracted human DNA was kindly provided by Dr. David Stirling of Edinburgh Royal Infirmary.

Silicon wafers were obtained from Shin Etsu, Livingston, Scotland. COOH-ThermoFast PCR microtitre plates were obtained from Advanced Biotechnologies, Leatherhead, England.

A Touchdown[™] thermal cycler was provided by Life Sciences International, Basingstoke, England.

195
5.3 Methods

5.3.1 Patterning of nucleic acids using avidin and photobiotin.

Neutravidin was immobilised on a SiO₂ wafer using the silanisation process as described in Section 2.3.1. The avidin-modified wafer was incubated in 5 ml of 10 μ g ml⁻¹ long arm photobiotin in PBS for 20 minutes, and was then covered with 10 μ g ml⁻¹ biotinylated DNA and exposed to light as described in Section 2.3.1 through a mask with 10 μ m interval equal mark:space ratio lines. The wafer was covered with 10 mg ml⁻¹ casein in PBS and exposed to light for 15 minutes in the absence of a mask. The wafer was then incubated in 10 μ g ml⁻¹ FITC labelled streptavidin in PBS, for 60 minutes. After washing in PBS and distilled water the sample was dried in a stream of nitrogen and examined using fluorescent microscopy. A photomicrograph of this surface is shown in Section 5.4.1.

5.3.2 Utilising the avidin-photobiotin immobilisation technique in the determination of Factor V genotypes

5.3.2.1 Immobilisation of oligonucleotide primers.

Wells of a COOH-ThermoFastTM plate were incubated with 1% (w/v) EDC in 10 mM phosphate buffer (pH 7.0) for 120 minutes at room temperature. The wells were then incubated in 200 μ g ml⁻¹ Neutravidin in 10 mM phosphate buffer (pH 7.0) for 16 hours at 4°C. All subsequent stages of the immobilisation process were performed at room temperature under dark room conditions. After washing in PBS, wells were incubated with 10 μ g ml⁻¹ long-arm photobiotin in PBS for 20 minutes. Solutions containing 1 μ g ml⁻¹ of either primer 1 (normal sequence) or primer 2 (Factor V_{Leiden} sequence) were added to the wells and the plate was exposed to light from a high pressure mercury vapour lamp for 15 minutes. After washing in 10 mM phosphate buffer containing 1M NaCl and 0.1% Triton X-100 (to remove any non-covalently bound oligonucleotides) the plate was ready for use.

5.3.2.2 PCR amplification of DNA

Samples consisted of DNA extracted from the blood of individuals with known Factor V genotypes. 1 μ l of sample was added to 50 μ l of PCR reagent (20 U ml⁻¹ Taq polymerase, 0.3mM dNTPs, 1.5 mM MgCl₂, 6 μ g ml⁻¹ biotinylated common primer in 10 mM Tris-HCl pH 7.6) in wells with both types of immobilised primer. The plate was then placed in the thermal cycler and PCR was performed using the following protocol:

| Initial Denaturation step: | | 5 Minutes at 95°C |
|----------------------------|--------------|--------------------|
| 30 cycles of | Denaturation | 15 seconds at 95°C |
| | Annealing | 15 seconds at 55°C |
| | Elongation | 30 seconds at 72°C |
| Final Elongation step | | 5 minutes at 72°C |

5.3.2.3 Detection of PCR products

Following the PCR amplification, the wells were washed with PBS containing 1% BSA and 0.1% Triton X-100. A streptavidin-HRP conjugate was added to the wells and incubated at 20°C for 30 minutes, and wells were then washed with PBS containing 1% BSA and 0.1% Triton X-100 prior to the addition of the substrate solution (contained H₂O₂ and 3,3',5,5'-tetramethylbenzidine (TMB) as the chromophore). After 5 minutes incubation at 20°C, the addition of 1M H₂SO₄, stopped the action of the peroxidase and changed the colour of the chromagen from a pale blue to an intense yellow colour ($\lambda_{Max} = 450$ nm). Absorbances were recorded and are shown in Section 5.4.2 for people with different known Factor V genotypes (verified by PCR and gel electrophoresis).

5.4. Results and Discussion

5.4.1 Patterning of nucleic acids using avidin and photobiotin.

The photomicrograph of the surface shown in Figure 5.5 demonstrates that high resolution $(<10\mu m)$ patterning of nucleic acids onto a surface could be achieved using the avidin-photobiotin technique developed in this work. It should be noted that the biotinylated DNA does not bind to the avidinated surface, via its inherent biotin groups. This is because the biotin binding sites of the immobilised avidin have been saturated by pre-incubation with photobiotin.



10 µm

Figure 5.5 Photomicrograph the patterning of nucleic acids by use of the avidin-photobiotin technique. Avidin and photobiotin were immobilised on to the surface of a SiO2 wafer as described in Section 2.3.3. The surface was then covered with biotinylated DNA in PBS and exposed to light through a mask with 10 μ m interval equal mark:space ratio lines. The wafer was covered with casein and exposed to light in the absence of a mask. The wafer was then incubated in FITC labelled streptavidin in PBS. Following washing the sample was dried in a stream of nitrogen and examined using fluorescent microscopy.

5.4.2 Utilising the avidin-photobiotin immobilisation technique in the determination of

Factor V genotypes

The results shown in Table 5.1 indicates that by immobilising oligonucleotide primers with the avidin-photobiotin technique, and using this modified surface in conjunction with PCR and a suitable detection method it is possible to detect genetic polymorphisms. Negligible absorbances are obtained when the sample DNA sequence does not match that of the specific immobilised primer. Where sequences are complementary between sample and primer, high absorbances are obtained.

| Sample | A ₄₅₀ Normal genotype primer | A ₄₅₀ Variant genotype primer |
|--------------------|--------------------------------------------|---------------------------------------------|
| Homozygous normal | 2.302 | 0.067 |
| Heterozygous | 1.892 | 1.726 |
| Homozygous variant | 0.078 | 2.251 |

<u>Table 5.1 Results obtained from patients with different Factor V genotypes.</u> Individuals with a homozygous normal genotype only show a reaction with the oligonucleotide primer for the normal genotype. Individuals with a homozygous variant genotype only show a reaction with the oligonucleotide primer for the variant genotype, whilst individuals who are heterozygous show reactions with both primers.

It may be possible to construct sensor arrays for use in genetic analysis using the avidin-photobiotin immobilisation technique to selectively immobilise different oligonucleotide primers to specific parts of the array. The substrate of the array could incorporate an element allowing its temperature to be cycled, and thus an integrated device could be fabricated that would have primers capable of recognising specific nucleotide sequences and the capability of obtaining the temperature control necessary for PCR without the requirement for expensive thermal cyclers. A major problem in the construction of a device that incorporates the DNA amplification stage would be the elimination of non-specific binding. Because of the tremendous amplification achieved by PCR, even the presence of one molecule of non-specifically bound nucleotide sequence could have a seriously detrimental effect.

5.5 Review of related work from other groups since publication of the avidin-photobiotin immobilisation technique.

Since the first description of using photobiotin to selectively immobilise biological molecules¹⁹⁷, a number of other workers have used the technique or modifications thereof.³¹¹⁻³¹⁷

The avidin-photobiotin patterning method has been applied to direct sensing detection systems.^{311,312} Figure 5.6 shows a SPR microscope image of IgG patterned on to a surface using the avidin-photobiotin method.³¹¹ The pattern was created using a a mask with 50 μ m wide equal mark:space ratio lines.



500 µm

Figure 5.6 Surface plasmon resonance microscopy image of IgG patterned on to a surface using the avidin-photobiotin method. The pattern was created using a a mask with 50 μ m wide equal mark:space ratio lines.³¹¹

One group used avidin/photobiotin to attach enzymes to a carbon electrode surface such that the enzymes were spatially separated and directly adjacent to electrode transfer sites on the same electrode surface.³¹³ High resolution patterning of the enzymes was required, so that the distance between the enzyme-loaded domains and the electron transfer sites was less than 5 µm, so as to maintain the fast response time and high sensitivity required for the measurement of neurotransmitter dynamics. Patterning of enzymes was achieved by use of an interference pattern generated from a UV laser. Characterisation of the surfaces was performed using fluorescence microscopy and AFM. This group has also shown that extensive biotin/avidin derivatisation of carbon electrode surfaces results in a decrease of the electron transfer rate of the electrode, but use of photobiotin to specifically immobilise proteins directly adjacent to electrode transfer sites (where photobiotin has not been activated, and protein immobilisation has not occurred) on the same electrode surface maintained fast response times and high sensitivity.³¹⁴ The use of an interference pattern generated from a UV laser to activate photobiotin has recently been used to immobilise antibodies onto a carbon surface with with sub-micron resolution.³¹⁵

Other workers have created patterns of biotin on polymer (polystyrene and nitrocellulose) surfaces, by exposing the polymeric surface to photobiotin in the presence of light through a mask.³¹⁶ The subsequent addition of avidin resulted in the formation of surfaces to which biotinylated molecules could then be bound through a biotin-avidin-biotin bridge. The technique was used to pattern alkaline phosphatase and horseradish peroxidase whilst retaining their catalytic activity. It should be noted that this method is very similar to that described in Sections 2.3.1 and 2.13.1, the difference being the use of a polymer instead of a protein modified surface as the matrix for attachment of photobiotin.

A variation on the photobiotin based immobilisation method, is the use of a "caged" biotin analogue.³¹⁷ The biotin analogue (which contains a photoactivable nitroveratryloxycarbonyl (Nvoc) group) is covalently linked to a substrate surface. The Nvoc group blocks the avidin-biotin interaction, but upon exposure to UV light it is cleaved yielding biotin sites which can be used to immobilise biotinylated macromolecules via a biotin-avidin-biotin bridge. Patterning can be achieved by the use of a photolithographic mask.

Advances have been made in other methods of patterning biological molecules, particularly with substrates containing micron and sub-micron scale channels.^{318,319} The availability of micro machined dispensing systems capable of dispensing picolitre quantities of liquids,^{320,321} provides the opportunity of physical deposition of material with high resolution. Sub-micron resolution patterning of immunoglobulins has been achieved using microfluidic networks, using gold, glass and polystyrene as substrates.³¹⁸

5.6 Future possibilities

The work presented in this thesis has demonstrated the development of a novel method for selectively immobilising biological molecules in predefined areas with micron scale resolution, and its use in a number of applications. There are a number of areas in which the work outlined in this thesis could be taken further.

5.6.1 Multianalyte Enzymic biosensors

The successful fabrication of a multianalyte enzymic biosensor using ultramicroelectrodes was demonstrated in Chapter 4. Future work could be undertaken to construct arrays of electrodes measuring a large number of analytes simultaneously. In these arrays a number of sensor elements could measure the same analyte in parallel, which would allow more accurate and reproducible determinations (through averaging) and minimising the effect of (or even excluding) any sensor elements producing a result considered to be an outlier. The avidin-photobiotin immobilisation technique could also be applied to the manufacture of multianalyte optical biosensors. One particularly attractive feature of this application is that for sensors such as those employing fibre-optic technology the light necessary for immobilisation could be delivered by the sensors tranducing element itself, which would make fabrication very simple.

5.6.2 Multianalyte Immunosensors

The successful fabrication of a number of multianalyte immunosensors has been demonstrated in the course of this work. In Chapter 2 the fabrication of qualitative immunosensors (both optical and amperometric) for up to 5 analytes was demonstrated. In Chapter 3 a quantitative amperometric multianalyte immunosensor for the determination of gonadotrophins was constructed, and in Chapter 4 the feasibility of using very small electrodes in multianalyte amperometric immunosensors was shown.

Arrays of electrodes could be constructed, and as for enzymic sensors (Section 5.7.1) these could be used in controlling the quality of the results produced. Again as for enzymic systems the avidin-photobiotin method could be utilised in the production of optical immunosensors.

5.6.3 Integrated devices for analysis of nucleic acids.

There are numerous reports in the literature of biosensors designed for the analysis of nucleic acids.³²²⁻³²⁴ One potential application of an immobilisation technique such as the one developed in the course of this work, is in the fabrication of an integrated device for the amplification and analysis of nucleic acids. As discussed in section 5.4.4, oligonucleotide primers complementary to a variety of nucleic acid sequences could be immobilised onto elements of an array on a substrate that incorporates an element capable of controlling its temperature. This sensor could be placed in contact with the sample and reagents and its temperature cycled to perform the PCR amplification.

A multianalyte sensor constructed in this way could be very small, and thus very rapid temperature changes could take place resulting in a shortening of assay time. A number of potential detection methods exist including fluorescence, amperometry or direct methods such as SPR. As was discussed in section 5.4.4. a likely major problem in the construction and use of such a device would be the complete elimination of non-specific binding (due to the ability of the PCR reaction to amplify minute amounts of nucleic acids).

5.6.4 Direct sensing

The avidin-photobiotin immobilisation method could be used for multianalyte direct and real-time sensing of reactions involving a change of mass at the sensor surface. Such reactions would include immunological reactions and detection of nucleic acid amplification products as discussed in section 5.6.3. The combination of direct sensing techniques with the avidin-photobiotin immobilisation procedure has already been demonstrated (Section 5.5).^{311,312}

5.6.5. Cell Guidance

A further possible application of this immobilisation procedure is in the fabrication of a device for the directional propagation of an individual cell, (e.g. nerve cell processes) on a patterned substrate. In this case, the essential prerequisites are the same as for the design of a diagnostic device, in so much that it is desirable to position molecules as a pre-defined pattern on a substrate with no interference from non-specific adsorption. Patterned molecules may act as a chemotactic or topological template for the guidance of the cell, which grow preferentially in a given direction determined by the pattern.¹⁴² Particularly important applications would be in the manufacture of devices either to control endothelial cell growth for wound healing, or to control nerve cell growth to promote regeneration.

In addition to the design of devices for the guidance of individual cells, it may be desirable to use a patterning technique to alter the behaviour of many cells. For example, by coating appropriate molecules onto a surface, it may be possible to differentially promote or prevent

cell adherence and growth on the outer surface of a miniature implantable *in vivo* sensor, in order to enhance the biocompatibility properties of the device.

An initial investigation was performed to assess if the avidin-photobiotin immobilisation technique could be used for guiding cell growth. Briefly, concanavalin A and BSA were patterned on to a silicon dioxide wafer. Snail neurones were placed on the substrate and incubated in growth media for 7 days. The cells were examined under a microscope. No guidance of the cell processes was evident, whereas in a control where protein had been patterned using a photolithographic lift-off technique¹⁴² (see section 1.9.2) guidance of cell processes was evident with cellular processes running parallel to each other with distances between them reflecting the periodicity of lines on the photolithographic mask. The effectiveness of both patterning techniques was validated by patterning of biotinvlated Concanavalin A, followed by incubation with a streptavidin-FITC conjugate. The reason for the difference in cell guidance between surfaces patterned with protein by these two techniques is not known. The lift-off technique may provide a firmer topological guide to cells than does the avidin-photobiotin technique, or changes to deposited protein as a result of the exposure to organic solvents in the lift-off technique may be influential if cell guidance is chemotactic in nature.

5.6.6 Other applications of patterning technique

The immobilisation procedure developed in the course of this work could also be used to selectively deposit molecules onto a surface in ordered arrays for applications other than diagnostic sensors. Using this procedure, groups of molecules (which may possess particular properties, for example photodynamic proteins) may be positioned precisely with respect to other groups of molecules, or to other structures in order to build up functional molecular architectures. As in the case for the design of a diagnostic device, a variety of different transducer materials may be used as immobilisation substrates.

5.6.7 Photoactivable Reagents other than Aryl Azides

There are a number of reactive groups other than aryl nitrenes which may be formed photochemically. Carbenes may be formed from diazo compounds and diazirines, and are even more highly reactive than aryl nitrenes.²⁰⁴ Nitrophenyl ethers can undergo photochemical aromatic substitution selectively at amino groups, and psoralens have been used to react with pyrimidine and purine groups in nucleic acids. Other photactivable groups include non aryl azides, α , β -unsaturated ketones, aryl halides, nitroaryl compounds, purines and pyrimidines.²⁰⁴ A major advantage of using photactivable reagents other than aryl azides would be in orientation of the immobilised molecule. A particular example where this would be extremely useful would be the patterning of oligonucleotides. Although the immobilisation technique employing an aryl azide (in the photobiotin molecule) was successful in that DNA amplification and detection occurred, it is likely that a significant proportion of the immobilised oligonucleotide was unable to act as a primer due to steric hinderance. As aryl azides react with a very wide range of species, the

photobiotin may attach the oligonucleotide such that subsequent binding of nucleotides or the action of Taq polymerase are inhibited. Oligonucleotides with an amino group at one end are easily synthesised and are commercially available. Selective immobilisation using a photoactivable reagent such as nitrophenyl ethers that react selectively with amino groups would allow optimal orientation of the nucleotide for DNA amplification and detection. Photoreactive compounds containing specific peptide sequences promoting cell adhesion have been synthesised, and have been used to prepare 2-dimensional cellular patterns.³²⁵

5.6.8 Very High Resolution Patterning

The resolution of the protein pattern that can be "written" using a photolithographic mask is limited to approximately half the wavelength of light used.³²⁶ Other workers have demonstrated sub-micron resolution when employing avidin-photobiotin, by use of laser interferometric techniques (Section 5.5).³¹⁵ Another potential approach to patterning of biological molecules with improved resolution is by the use of a near-field optical scanning microscope (NFOSM).^{327,328} This instrument is capable of achieving resolution of about 20 nm. In a NFOSM a tiny aperture, typically 10 nm in diameter, is positioned at the apex of a conical metal screen. This aperture is illuminated from its reverse side and is scanned over the sample in very close proximity to it, maintaining the sample tip gap by utilisation of the tunnelling current as in STM. This technique has been applied to the generation of submicron scale patterns in photoresists,³²⁹⁻³³¹ and it is possible that it could be applied to the avidin-photobiotin patterning technique.

REFERENCES

He ranged his tropes, and preached up patience; Backed his opinion with quotations.

Matthew Prior

Paulo Purganti and his wife

You will find it very good practice always to verify your references, Sir!

Martin Joseph Routh

John William Burgon Lives of Twelve Good Men

- Boitieux, J. L., Desmet, G., Wilson, G. & Thomas, D. "The specific immobilisation of antibody fragments on membrane for the development of multifunctional biosensors.", Annals of the New York Academy of Sciences, 613, 390-395 (1990)
- Bresler, H. S., Lenkevich, M. J., Newman, A. L. & Roblin, R. O. "Multianalyte antibody biosensor for simultaneous detection of antibodies to HIV antigens-progress towards an electronic western blot.", *FASEB Journal*, 5, 1457 (1991)
- Cooper, J. M. & McNeil, C. J. "Biosensors into the 1990s.", Anal. Proc., 27, 95-96 (1990)
- Cammann, K., Lemke, U., Rohen, A., Sander, J., Wilken, H. & Winter, B.
 "Chemical sensors and biosensors", *Angew. Chemie Int. Ed. Engl.*, 30, 516-539 (1992)
- 5. Sethi, R. S. "Transducer aspects of biosensors.", GEC J. of Research, 9, 81-96 (1991)
- 6. Updike, S. J. & Hicks, G. P. "The enzyme electrode.", *Nature*, **214**, 986-988 (1967).
- 7. Guilbault, G. G. & Montalvo, J. "A urea specific electrode.", J. Am. Chem. Soc.,
 91, 2164-2165 (1969)
- 8. Janata, J. " An immunoelectrode.", J. Am. Chem. Soc., 97, 2914-2916 (1975)
- 9. Aizawa, M., Kato, S. & Suzuki, S. "Immunoresponsive membrane.", J. Membrane Sci., 2, 125-132 (1977)
- Schubert, F. & Scheller, F. W. "Organelle electrodes." in *Methods in Enzymology* Volume 137 (Ed. Mosbach, K.), Academic Press (London), 152-160 (1988)

- 11. Tedseco, J. L., Krull, U. J. & Thompson, M. "Molecular receptors and their potential for artificial transduction.", *Biosensors*, 4, 135-167 (1989)
- Deshpande, M. V. & Hall, E. A. H., "An electrochemically grown polymer as an immobilisation matrix for whole cells. Application in an amperometric dopamine sensor.", *Biosensors*, 5, 431-448 (1990)
- Rechnitz, G. A., Arnold, M. A. & Meyerhiff, M. E. "Bio-selective membrane electrode using tissue slices.", *Nature*, 278, 466-467 (1979)
- 14. Wijesurija, D. C. & Rechnitz, G. A. "Biosensors based on plant and animal tissues.", *Biosensors*, 8, 155-160 (1993)
- Lowe, C. R., Yon Hin, B. F. Y., Cullen, D. C., Evans, S. E., Stephens, L. D. G. & Maynard, P. "Biosensors.", J. Chromat., 510, 347-354 (1990)
- 16. Lowry, O. H. "Amplification by enzymatic cycling.", Molecular and Cellular Biochemistry, **32**,135-146 (1980)
- Watson, L.D., Maynard, P., Cullen, D. C., Sethi, R. S., Brettle, J. & Lowe, C. R.
 "A micro-electronic conductimetric biosensor.", *Biosensors*, 3, 101-115 (1987)
- Sutherland, R. M., Dahne, C., Place, J.F. & Ringrose, A. S. "Optical detection of antibody / antigen reactions at a glass / liquid interface.", *Clin. Chem.*, 30, 1533-1538 (1984)
- 19. Roederer, J. E. & Bastiaans, G. J. "Microgravimetric immunoassay with piezoelectric crystals.", *Anal. Chem.*, **55**, 2333-2336 (1983)
- Bataillard, P., Stefgren, E., Haemmerli, S., Manz, A. & Widner, H. M. "An integrated silicon thermopile as biosensor for the thermal monitoring of glucose, urea and penicillin.", *Biosensors*, 8, 89-98 (1993)

- Hundeck, H. G., Wei, M., Scheper, T. & Schubert, F. "Calorimetric biosensor for the detection of enantiometric excesses in aqueous and organic phases.", *Biosensors*, 8, 205-208 (1993)
- 22. Clark, L. C. & Lyons, C. "Electrode system for continuous monitoring in cardiovascular surgery.", Ann. N. Y. Acad. Sci., 102, 29-45 (1962)
- Aizawa, M. Morioka, A, Suzuki, S., & Nagamura, Y. "Enzyme immunosensor III. Amperometric determination of human chorionic gonadotrophin by membrane bound antibody", *Anal. Biochem*, 94, 22-28 (1979)
- 24. Itagaki, H., Hakoda, Y., Suzuki, Y. & Haga, M. "Drug sensor. an enzyme immunoelctrode for theophylline.", *Chem. Pharm. Bull.*, **31**, 1283-1288 (1983)
- Renneberg, R. Schlossler, W. & Scheller, F. "Amperometric enzyme sensor based enzyme immunoassay for Factor VIII-related antigen.", *Anal. Lett.*, 16, 1279-1289 (1983)
- Meusel, M., Renneberg, R., Spener, F. & Schmitz G. "Development of a heterogeneous amperometric immunosensor for the determination of apolipoprotein E in serum." *Biosensors*, 10, 577 -586 (1995)
- 27. Doyle, M. J., Halsall, H. B. & Heinemann W. R. "Enzyme-linked immunoabsorbent assay with electrochemical detection for a1-acid glycoprotein". *Anal. Chem.*, 56, 2355-2360 (1984)
- 28. Kalab, T. & Skaldal, P. "A disposable amperometric immunosensor for
 2,4-dichlorophenoxyacetic acid.", Anal. Chim. Acta, 304, 361-369 (1995)
- 29. Robinson, G. A., Cole, V. M., Rattle, S. J. & Forrest, G. C. "Bioelectrochemical immunoassay for human chorionic gonadotrophin in serum using an electrode-immobilised capture antibody.", *Biosensors*, **2**, 45-57 (1986)

- Rishpon, J. & Rosen, I. "The development of an immunosensor for the electrochemical determination of the isoenzyme LDH5.", *Biosensors*, 4, 61-74 (1989)
- Haben, F. & Klemensiewicz, Z. "Uber elektrische phasengrenzkrafte.", Z. Phys. Chem., 67, 385-431 (1909)
- Van der Schoot, B. H., Bergveld, P. "ISFET based enzyme sensors.", *Biosensors*, 3, 161-186 (1988)
- 33. Caras, J. & Janata, J. "Field effect transistor sensitive to penicillin.", Anal. Chem.,
 52, 1935-1937 (1980)
- 34. Ghindils, A., Skorobat'ko, O., Gavrilova, V. & Yarapalov, A. "A new approach to the construction of potentiometric immunosensors.", *Biosensors*, 7, 301-304 (1992)
- 35. Yamamoto, N., Nagasawa, Y., Shuto, S., Tsubomura, H., Sawai, M. & Okumura, H.
 "Antigen-antibody reaction investigated with the use of a chemically modified electrode.", *Clin. Chem.*, 26, 1569-1572 (1980)
- Boitieux, J-L., Desmet, G. & Thomas, D. "An 'Antibody electrode,' preliminary report on a new approach in enzyme immunoassay.", *Clin. Chem.*, 25, 318-321 (1979)
- Solsky, R. L. & Rechnitz, G. A. "Antibody selective membrane." Science, 204, 1308-1309 (1979)
- 38. Bush, D. L. & Rechnitz, G. A. "Antibody response of polymer membrane electrodes incorporating antigenic ionophores.", J. Memb. Sci., **30**, 313-320 (1987)
- 39. Schasfoort, R. B., Kooyman, R. P., Bergveld, B. & Greve, J. "A new approach to immunoFET operation.", *Biosensors*, 5, 103-124 (1990)

- 40. Sandberg, R. G. "A conductive polymer based immunosensor for the analysis of pesticide residues.", *Am. Chem. Soc. Symp. Ser.*, **511**, 81-88 (1992)
- 41. Hicks, G. P. & Updike, S. J. "Preparation and characterisation of lyophilised polyacrylamide gels for chemical analysis.", *Anal. Chem.*, **8**. 726-730 (1966)
- 42. Yagiuda, K., Hemmi, A., Ito, S., Asano, Y., Fushinuki, Y., Chen, C.-Y. & Karube, I.
 "Development of conductivity-based immunosensor for sensitive detection of methamphetamine in human urine.", *Abstracts of 3rd World Congress on Biosensors* (1994)
- Moss, D. W. & Henderson, R. "Enzymes" in *Tietz Textbook of Clinical Chemistry:* 2nd Edition., (Eds. Burtis, C. A. & Ashwood, E. R.), W. B. Saunders (Philadelphia), 771, (1994)
- Buck, R. H., Norman, R. J., Reddi, K. Moodley, J. & Joubert, S. M. "Various methods for the early diagnosis of ectopic pregancy.", *Clin. Chem.*, 32, 879-882 (1986)
- 45. Glad, C. & Grubb, A. O. "Immunocapillary migration-a new method for immunochemical quantitation.", *Anal. Biochem.*, **85**, 180-187 (1978)
- 46. Hopper, G. D., Wagman, B. & Tilsson, S. "On-site theophylline analysis using enzyme immunochromatography.", Am. Clin. Prods. Rev., 5, 36-41 (1986)
- 47. Yoder, J. M., "A sensitive type of immunocapillary migration assay that detects insulin.", *Clin. Chem.*, 25, 814 (1979)
- 48. Yoshida, M., Swigeromi, K., Sugimura, M. & Matano, M. "Sensitivity enhancement of evanescent wave immunoassay.", *Meas. Sci. Technol.*, **4**, 1077-1079 (1993)
- 49. Starodub, N., Arenkov, P. Starodub, A. & Berezin, V. "Construction and biomedical application of biosensors based on fiber optics and enhanced chemiluminescence.", Optical Eng., 33, 2958-2963 (1994)

- 50. Starodub, N., Arenkov, P. Starodub, A. & Berezin, V. "Fiber optic immunosensors based on chemiluminescence and their application to determine different antigens.", Sensors & Actuators B Chem., 18, 161-165 (1994)
- 51. Selifnova, O., Burlage, R. & Barkay, T. "Bioluminescent sensors for detection of bioavailable Hg(II) in the environment", *Appl, Envion. Microbiol.*, 59, 3083-3090 (1993)
- 52. Hemilla, I. "Fluoroimmunoassays and immunofluorometric assays", *Clin. Chem.*,
 31, 359-370 (1985)
- 53. Coombs, R. R. A., Scott, M. L. & Cranage, M. P. "Assays using red cell- labelled antibodies.", J. Immunol. Meth., 101, 1-14 (1987)
- 54. Urdal, P., Borch, S.M., Landaas, S., Krutnes, M.B., Gogstad, G.O. and Hjortdahl, P.
 "Rapid immunometric measurement of C-reactive protein in whole blood." *Clin. Chem.* 38, 580-584, (1992)
- 55. Sutherland, R. & Dahne, C. "IRS devices for optical immunoassays.", in *Biosensors.* Fundamentals and applications (Eds. Turner, A. P. F., Karube, I. & Wilson G. S.), Oxford University Press (Oxford), 655-679 (1987)
- 56. Vroman, L. & Adams, A. "Findings with the recording ellipsometer suggesting rapid exchange of specific plasma proteins at solid / liquid interfaces.", Surf. Sci., 16, 438-446 (1969)
- 57. Liedberg, B. Nylander, C. & Lundstrom, I. "Surface plasmon resonance for gas detection and biosensing.", *Sensors & Actuators*, 4, 299-304 (1983)
- 58. Nellen, P., Tiefenthaler, K., & Lukosz, W. "Inetgrated optical input grating couplers as biochemical sensors.", *Sensors & Actuators*, **15**, 285-297 (1988)

- Sutherland, R., Simpson, B. & Allman, B. "Surface effect immunoassay", in *Principles and Practice of Immunoassay* (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 515-542 (1991)
- Ruzgas, T. A., Razumas, V. J. & Kulys, J. J. "Ellipsometric immunosensors for the determination of γ-interferon and human serum albumin.", *Biosensors*, 7, 305-308 (1992)
- 61. Attridge, J. W., Daniels, P. B., Deacon, J. K., Robinson, G. A. & Davidson G. P.,
 "Sensitivity enhancement of optical immunosensors by the use of a surface plasmon resonance fluoroimmunoassay.", *Biosensors*, 6, 201-214 (1991)
- Ogert, R. A., Brown, J. E., Singh, R. A., Shriver-Lake, L. C. & Ligler, F. S.
 "Detection of *Clostridium botulinum* toxin A using a fiber optic-based biosensor.", *Anal. Biochem.*, 205, 306-312 (1992)
- 63. Curie, P. & Curie, J. "Cristallopysique.- Developpement, par pression, de l,electricite polaire dans les cristaux hemiedres a faces inclinees.", C. R. Acad. Sci., 91, 294-295 (1880)
- 64. Barnes, C., D'Silva, C., Jones, J. P. & Lewis, T. J. "The theory and operation of piezoelectric quartz crystal sensors for biochemical application.", Sens. and Actuators A, 31, 159-163 (1992)
- 65. Alder, J. & McCallum, J. "Piezo-electric crystals for mass and chemical measurements", *Analyst*, **108**, 1169-1189 (1983)
- 66. Guilbault, G. G. & Jordan, J. "Analytical uses of the piezoelectric crystal detector.", Crit. Rev. Anal. Chem., 19, 1-28 (1988).
- 67. Ngeh-Ngwainbi, J., Suleiman, A. & Guilbault, G. "Piezoelectric crystal biosensors.", *Biosensors*, 5, 13-26 (1990)

- 68. Guilbault, G. G., Hock, B. & Schmid, R. "A piezoelectric immunobiosensor for atrazine in drinking water.", *Biosensors*, 7, 411-419 (1992)
- 69. Yokoyama, K. "Highly sensitive quartz-crystal immunosensors for multisample detection of herbicides", *Anal. Chim. Acta.*, **304**, 139-145 (1995)
- Miura, N., Higobashi, H., Sakai, G., Takeyashu, A., Uda, T. & Yamazoe, N.
 "Piezoelectic crystal immunosensor for detection of methamphetamine (stimulant drug) in human urine.", Sensors & Actuators B Chem., 13, 188-191 (1993)
- Aberl, F., Wolf, H., Kosslinger, C., Drost, S., Woias, P., & Koch, S. "HIV serology using piezoelectric immunosensors." Sensors & Actuators B Chem., 18, 271-275 (1994)
- 72. Fox, C. & Alder, J. "Surface acoustic wave sensors for atmospheric gas monitoring: a review.", *Analyst*, **114**, 997-1004 (1989)
- 73. Gizeli, E., Goddard, N. J., Lowe, C. R. & Stevenson, A. C. "A love plate biosensor utilising a polymer layer.", Sensors and Actuators B Chem., 6, 131-137 (1992)
- 74. Mosbach, K. & Danielsson, B. "An enzyme thermistor." *Biochim. Biophys. Acta Rep.*, 364, 140-145 (1974)
- 75. Mosbach, K. "Thermal Biosensors.", Biosensors, 6, 179-182 (1991)
- Mattiasson, B. & Danielsson, B. "Calorimetric analysis of sugars and sugar derivatives with the aid of an enzyme thermistor.", *Carbohydr. Res.*, 12, 273-282 (1982)
- Birnbaum, S., Bulow, L. Hardy, K., Danielsson, B. & Mosbach, K. "Automated thermometric enzyme immunoassay of human proinsulin produced by *Escherischia coli*." *Anal. Biochem.*, 158, 12-19 (1986)

- Mattiasson, B., Borrebaeck, C., Sanfrdison, B. & Mosbach, K. "thermometric enzyme linked immunosorbent assay: TELISA.", *Biochim. Biophys. Acta*, 483, 221-227 (1977)
- 79. Wide, L. "Solid phase antigen-antibody systems.", in *Radioimmunoassay Methods* (Eds. Kirkham, K. E. & Hunter, W. M.), Churchill-Livingstone (Edinburgh), 405-412 (1971)
- Schlatter, D., Barner, R., Fattinger, C., Huber, W., Hubscher, J., Hurst, J.,
 Koller, H., Mangold, C. & Muller, F. "The difference interferometer: Application as a direct affinity sensor.", *Biosensors*, 8, 109-116 (1993)
- 81. Fert, V. & Baret, A. "Preparation and characterisation of xanthine-oxidase-antibody and -hapten conjugates for use in sensitive chemiluminescent immunoassays.",
 J. Immunol. Meth., 131, 237-247 (1990)
- 82. Chapman, R. S., Sutherland, R. M. & Ratcliffe, J. G. "Application of 1,1'-carbonyldiimidazole as a rapid practical method for the production of solid-phase reagents." in *Immunoassays for Clinical Chemistry* (Eds. Hunter, W. M. & Corrie, J. E. T.) 2nd edition, Churchill-Livingstone (Edinburgh), 178-190 (1983)
- Valkirs, G. E., & Barton, R. "Immunoconcentration[™] a new format for solid-phase immunoassays.", *Clin. Chem.*, 31, 1427-1431 (1985)
- Hersh, L. S. & Yaverbaum, S., "Magnetic Solid Phase Radioimmunoassay.", Clin. Chim. Acta, 63, 69-72 (1975)
- Forrest, G. C. & Rattle, S. J., "Magnetic Particle Radioimmunoassay.", in Immunoassays for Clinical Chemistry (Eds. Hunter, W. M. & Corrie, J. E. T.) 2nd edition, Churchill-Livingstone (Edinburgh), 147-162 (1983)

- Kemeny, D. M. & Challacombe, S. J. "Microtitre plates and other solid-phase supports." in *ELISA and other Solid Phase Immunoassays: Theoretical and Practical Aspects* (Eds Kemeny, D. J. & Challacombe, S. J), John Wiley and Sons (Chichester), 31-56 (1988)
- 87. Catt, K. & Tregear, C. W. "Solid phase radioimmunoassay in antibody coated tubes.", *Science*, **158**, 1570-1572 (1967)
- Jefferis, R. & Deverill, I. "The antigen-antibody reaction." in *Principles and* Practice of Immunoassay (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 1-18 (1991)
- 89. Alberts, B. Molecular Biology of the Cell, Garland Publishing (New York), (1983)
- Berson, S. A. & Yalow, R. S. "Isotopic tracers in the study of diabetes.", Adv. in Biol. Med. Phys., 6, 349-430 (1958)
- 91. Yalow, R. S. & Berson, S. A. "Radiobiology assay of plasma insulin in human subjects by immunological methods.", *Nature*, **184**, 1648-1649 (1959)
- 92. Kricka, L. J. "Chemiluminescent and bioluminescent techniques.", Clin. Chem., 37, 1472 1481 (1991)
- 93. Ullman, E. F. & Khanna, P. L. "Fluorescence excitation transfer immunoassay (FETI)." in *Methods in Enzymology Volume 74* (Eds. Langone, L. & Van Vunakis, H.), Academic Press (London), 28-60 (1981)
- Keller, C. H., Fries, P., Kline, E., Neider, K., Doles, P. & Ogunro, E. A. "Rapid automated enzyme immunoassay for follitropin with Abbott IMx analyzer.", *Clin. Chem.*, 36, 697-698 (1990)
- 95. Plebani, M., Masiero, M., Sciacovelli, L., Faggian, D. & Burlina, A. "A rapid specific immunoassay for follitropin and lutotropin determination.", *Clin. Chem.*, 34, 772 (1988)

- 96. Johannsson, A. "Heterogeneous enzyme immunoassay." in *Principles and Practice of Immunoassay* (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 295-325 (1991)
- 97. Morgan, C. L., Newman, D. J. & Price, C. P. "Immunosensors: technology and opportunities in laboratory medicine.", *Clin. Chem.*, **42**, 193 209 (1996)
- 98. Guilbault, G., Kramer, D. N. & Cannon, P. L., "Electrochemical determination of organophosphorous componds", *Anal. Chem.*, **34**, 1437-1439 (1962)
- 99. Behizud, M., Cumming, R.H., Rowell, F.J., Salisbury, T.T., & Stewart, J.W. "Safety in biotechnology-The use of biosensors for the detection of hazardous biochemicals in air.", *Process Biochemistry*, **24**, 126-132, (1989)
- 100. Ludi, H., Bataillard, P., Haemmerli, S., & Widmer, H.M. "Biochemical sensors in industry." Sensors and actuators B4, 207-209, (1991)
- Brooks, S.L., Ashby, R.E., Turner, A.P.F., Cadder, M.R., & Clarke, D.J.
 "Development of an on-line glucose sensor for fermentation monitoring." *Biosensors*, 3, 45-56, (1987)
- 102. Van Emon, J.M. and Lopez-Avila, V. "Immunochemical methods for environmental analysis." Anal. Chem. 64, 79A-88A, (1992)
- Dennison, M. J. & Turner, A. P. F. "Biosensors for environmental monitoring.", Biotechnol. Adv., 13, 1-12 (1995)
- 104. Watanabe, E., Toyama, K., Karube, I., Matsuoka, H. & Suzuki, S. "Multifunctional biosensor for the determination of fish meat freshness." Annals of the New York Academy of Sciences 434, 529-532 (1984)
- 105. Haga, M., Sugawara, S., and Itagaki, H. "Drug sensor-liposome immunosensor for theophylline." Anal. Biochem. 118, 286-293 (1981)

- 106. Newman, J.D. and Turner, A.P.F. "Biosensors-The analysts dream." Chemistry and Industry, 10, 374-378 (1992)
- 107. "Biosensors: Canaries Dwarfed.", Economist, 320, 96-97, 28th September 1991.
- 108. Vadgama, P., "Biosensors in clinical biochemistry.", Ann. Clin. Biochem., 30, 337-340 (1993)
- 109. Marks, V. "Essential considerations in the provision of near-patient testing facilities.", Ann. Clin. Biochem., 25, 220-225 (1988)
- 110. Crofford, O. B. "Diabetes control and complications.", Annu. Rev. Med., 46, 267-279 (1995)
- Sherwood, M., Hinnefeld, S., Stromjensen, P., Lichatowich, D., Ganser, N.,
 Warchal, M. E. & Mecklenburg, G. "Determination of glucose in whole-blood with glucostix reagent strips." *Clin. Chem.*, 32, 1119 (1986)
- 112. Smith, J. H. "Laboratory staff and nurses performance compared when using the blood-glucose reflotest-reflomat system." *Med. Lab. Sci.*, 40, 283-285, (1983)
- 113. Davies, G. "Development of a commercial multichannel clinical sensor chip.",
 Abstracts of 3rd World Congress on Biosensors (1994)
- Pfeiffer, E. F. "The Ulm-Zucker-Uhr-system and its consequences." Hormone & metabolic Research, 26, 510-514 (1994)
- 115. Kolbezensimoniti, M. & Pretnardarovec, A. "Clearblue. Test for early-pregnancy detection.", Zdravstveni Vestnik, 60, 81-82 (1991)
- Yon-Hin, B. F. Y., Sethi, R. S. & Lowe, C. R. "Multi-analyte microelectronic biosensors.", Sensors and Actuators, B1, 550-554 (1990)

- Hanazoto, Y., Nakako, M., Shiono, S., and Maeda, M. "Integrated multi-biosensors based on an ion-sensitive field-effect transistor using photolithographic techniques." *IEEE Transactions on Electron Devices*, 36, 1303-1310 (1989)
- 118. Nakamoto, S., Ito, N., Kuriyama, T. & Kimura, J. "A lift-off method for patterning enzyme -immobilised membranes in multi-biosensors.", *Sensors and Actuators*, 13, 165-172 (1988)
- 119. Cullen, D.C., Sethi, R.S., and Lowe, C.R. "Multianalyte miniature conductance biosensor." *Anal. Chim. Acta*, 231, 33-40, (1990)
- 120. Buechler, K.F., Moi, S., Noar, B., McGrath, D., Villela, J., Clancy, M., Shenav, A., Colleymore, A., Valkirs, G., Lee, T., Bruni, J.F., Walsh, M., Hoffman, R., Almaty, F., Nowakowski, M., Buechler, J., Mitchell, M., Boyd, D., Stiso, N., & Anderson, R. "Simultaneous detection of seven drugs of abuse by the Triage[™] panel for drugs of abuse.", *Clin.Chem.* 38, 1678-1684 (1992)
- 121. Scwartz, J. G., Gage, C. L., Farley, N. J. & Prihoda, T. J. "Evaluation of the cardiac STATus (TM) CK-MB/myoglobin card test to diagnose acute myocardial infarctions in the ED.", Am. J. Emerg. Med., 15, 303-307 (1997)
- 122. Zhou, Y., Magill, J.V., Cushley, W., Laybourn, P.J.R., and De la Rue, R.M.
 Multichannel evanescent fluoresence immunosensing using potassium and sodium ion-exchanged wave-guides. J.Molecular Electronics, 7, 135-149 (1991)
- 123. Glazier, S.A. and McCurley, M.F. "Dual analyte fiber optic immunosensors employing antibody fragments labeled with environment sensitive dyes." *Abstracts* of papers of the American Chemical Society, **204**, 36 (1992)
- 124. Norde, W. "Energy and entropy of protein adsorption.", J. Dispersion Science & Technology, 3, 363-377 (1992)
- 125. Sadana, A. "Protein adsorption and inactivation on surfaces. Influence of heterogeneities.", *Chem. Rev.*, **92**, 1799-1818 (1992)

- 126. Lutanie, E., Schaaf, P., Schmitt, A., Voegel, J.C., Freund, M., and Cazenave, J.P.
 "Adsorption of human IgG molecules onto glass-beads reversible & irreversible aspects." *J.Dispersion Science and Technology* 13, 379-398, (1992)
- 127. Langmuir, I. "Adsorption of gases on plane surfaces of gases, mica and platinum", J.
 Am. Chem. Soc., 40, 1361-1403 (1918)
- Miller, I.R. & Bach, D. "Biopolymers at interfaces." in Surface and Colloid Science.
 Volume 6, Ed Matijevic, E.:Wiley-Interscience (New York), 185-260 (1973)
- Ahluwalia, A., De Rossi, D., Ristori, C., Schirone, A. & Serra, G. "A comparative study of protein immobilisation techniques for optical immunosensors.", Biosensors, 7, 207-214 (1991)
- 130. Weisshar, D. E., Lamp, B. D. & Porter, M. D. "Thermodynamically controlled electrochemical formation of thiolate monolayers at gold: Characterisation and comparison to self-assembled analogs.", J. Am. Chem. Soc., 114, 5860-5862 (1992)
- Jonsson, U., Malmqvist, M. & Ronnberg, I. "Immobilisation of immunoglobulins on silica surfaces: Stability.", *Biochem. J.*, 227, 363-371 (1985)
- Jonsson, U., Malmqvist, M. & Ronnberg, I. "Immobilisation of immunoglobulins on silica surfaces: Kinetics of immobilisation and influence of ionic strength.", *Biochem. J.*, 227, 373-378 (1985)
- 133. Hart, A. L. & Turner, A. P. F. "Production of a prototype lactate sensor by screen printing.", Abstracts of 3rd World Congress on Biosensors (1994)
- 134. Künnecke, W., Mohns, J., Rohm, I. & Bilitewski, U. "Development of screen printed biosensors for process monitoring.", Abstracts of 3rd World Congress on Biosensors (1994)
- Nagata, R., Yokoyama, K. & Karube, I. "Glucose sensor manufactured by printing technique.", Abstracts of 3rd World Congress on Biosensors (1994)

- Poplawski, M. E., Midgely, A. R. & Brown, R. B. "Batch fabricated amperometric biosensors.", Abstracts of 3rd World Congress on Biosensors (1994)
- 137. Kimura, J., Kawana, Y. & Kuriyama, T. "An immobilised enzyme membrane fabrication method using an ink-jet nozzle.", *Biosensors*, **4**, 41-52 (1988)
- 138. Newman, J. D., Turner, A. P. F. & Marrazza, G. "Ink-jet printing for the fabrication of amperometric glucose biosensors.", *Anal. Chim. Acta*, **262**, 13-17 (1992)
- 139. Blanchard, A. P., Kaiser, R. J. & Hood, L. E. "High density oligonucleotide arrays.", *Biosensors*, 11, 687-690 (1996)
- 140. Newman, J. D. "Printing processes for the manufacture of biosensors.", Abstracts of 3rd World Congress on Biosensors (1994)
- 141. Kleinfeld, D., Kahler, K. H. & Hockberger, P. E. "Controlled outgrowth of dissociated neurones on patterned substrates.", J. Neuroscience, 8, 4098-4120 (1988)
- Britland, S., Perez-Arnaud, E., Clark, P., McGinn, B., Connolly, P. & Moores, G.
 "Micropatterning proteins and synthetic peptides on solid supports: A novel application for microelectronics fabrication technology.", *Biotechnol. Prog.*, 8, 155-160 (1992)
- Bhatia, S. K., Teixeira, J. L., Anderson, M., Shriver-Lake, L. C., Calvert, J. M., Georger, J. H., Hickman, J. J., Dulcey, C. S., Schoen, P. E. & Ligler, F. S.
 "Fabrication of surfaces resistant to protein adsorption and application to two-dimensional protein patterning.", Anal. Biochem., 208, 197-205 (1993)
- 144. Johnson, K. W. "Reproducible electrodeposition of biomolecules for the fabrication of miniature electroenzymatic devices.", Sens. & Actuators, B-Chemical, 5, 85-89 (1991)

- 145. Nakamoto, S., Ito, N., Kuriyama, T. & Kimura, J. "A lift-off method for patterning enzyme-immobilised membranes in multi-biosensors.", Sensors and actuators, 13, 165-172, (1988)
- 146. Cooper, J. M. & Pritchard, D. J. "Biomolecular sensors for neurotransmitter determination. Electrochemical immobilisation of glutamate oxidase at microelectrodes in a poly(o-phenylenediamine) film.", J. Materials Science: Materials in Electronics, 5, 111-116 (1994)
- 147. Umana, M. & Waller, J. "Protein-modified electrodes. Glucose oxidase / polypyrrole system.", Anal. Chem., 58, 2979-2983 (1986)
- 148. Foulds, N. C. & Lowe, C. R. "Enzyme entrapment in electrically conducting polymers. Immobilisation of glucose oxidase in polypyrrole and its application in amperometric glucose sensors.", J. Chem. Soc. Faraday Trans. 1., 82, 1259-1264 (1986)
- 149. Bartlett, P. N. & Whitaker, R. G. "Electrochemical immobilisation of enzymes part I. Theory.", J. Electroanal. Chem., 224, 27-35 (1987)
- 150. Bartlett, P. N. & Whitaker, R. G. "Electrochemical immobilisation of enzymes part II. Glucose oxidase immobilised in poly-N-methylpyrrole.", J. Electroanal. Chem., 224, 37-48 (1987)
- 151. Iwakura, C., Kajiya, Y. & Yoneyama, H. "Simultaneous immobilisation of glucose oxidase and a mediator in conducting polymer films.", J. Chem. Soc. Commun., 1019-1020 (1988)
- 152. Gregg, B. A. & Heller, A. "Cross-linked redox gels containing glucose oxidase for amperometric biosensor applications.", *Anal. Chem.*, **62**, 258-263 (1990)
- 153. Malitesta, C., Palisano, F., Torsi, L. & Zambronin, P. G. "Glucose fast-response amperometric sensor based on glucose oxidase immobilised in an electopolymerised poly(o-phenylenediamine) film.", Anal. Chem., 62, 2735-2740 (1990)

- 154. Geise, R. J., Adams, N.J. "Electropolymerised films to prevent interferences and electrode fouling in biosensors.", *Biosensors*, 6, 151-160 (1991)
- 155. Sasso, S. V., Pierce, R. J., Wall, R. & Yacynych, A. M. "Electroploymerised 1,2-diaminobenzene as a means to prevent interferences, fouling and stabilise immobilised enzyme in electrochemical biosensors.", *Anal. Chem.*, 62, 1111-1117 (1990)
- 156. Kawashima, K. "Immobilisation of enzymes and microorganisms by radiation polymerisation.", in *Methods in Enzymology Volume 139*, (Ed. Mosbach, K.), Academic Press (New York), 146-153 (1987)
- 157. Hanazoto, Y., Nakako, M., Shiono, S. & Maeda, M. "Integrated multi-biosensors based on an ion-sensitive field-effect transistor using photolithographic techniques.", *IEEE Trans. on Electron Devices*, 36, 1303-1310 (1989)
- 158. Galiatsos, C., Ikariyama, Y., Mark, J. E. & Heineman, W. R. "Immobilisation of glucose oxidase in a poly(vinyl alcohol) matrix on platinised graphite electrodes by gamma-irradiation.", *Biosensors*, 5, 47-61 (1990)
- 159. Tsuji, I., Eguchi, H., Yasukouchi, K., Unoki, M. & Taniguchi, I. "Enzyme immunosensors based on electropolymerised polytyramine modified electrodes.", *Biosensors*, 5, 87-101 (1990)
- 160. Schuhmann, W., Lammert, R., Uhe, B. & Schmidt, H-L. "Polypyrrole. A new possibility for covalent binding of oxidoreductases to electrode surfaces as a base for stable biosensors.", Sens. & Actuators, B, 537-541 (1990)
- Cooper, J. M., Morris, D. G. & Ryder K. S."A bio-electronic interface using functionalised conducting poly(pyrroles)." J. Chem. Soc. Chem. Comm., 697 - 698 (1995)
- 162. Strike, D. J., de Rooij, N. F. & Koudelka-Hep, M. "Electrodeposition of enzymes on miniature electrodes.", Abstracts of 3rd World Congress on Biosensors (1994)

- 163. Group Discussion following presentation by Strike, D. J. at 3rd World Congress on Biosensors (1994)
- 164. Turko, I. V., Yurkevich, I. S. & Chaschin, V. L. "Orientated immunoglobulin G layer onto the Langmuir-Blodgett films of Protein A.", *Thin Solid Films*, 210/211, 710-712 (1992)
- 165. Barraud, A., Perrot, H., Billard, V., Martelet, C. & Therasse, J. "Study of immunoglobulin G thin layers obtained by the Langmuir-Blodgett method. Application to immunosensors.", *Biosensors*, 8, 39-48 (1993)
- 166. Sasaki, Y. C., Suzuki, Y. & Ishibashi, T. "Fluorescent X-ray interference form a protein monolayer.", *Science*, **263**, 62-64 (1994)
- 167. Kumar, A. & Whitesides, G. M. "Patterned condensation figures as optical diffraction gratings.", Science, 263, 60-62 (1994)
- 168. Clark, S. L., Montague, M. & Hammond, P. T. "Selective deposition in multilayer assembly: SAMs as molecular templates.", *Supramol. Sci.*, 4, 141-146 (1997)
- Duschl, C., Liley, M. & Vogel, H. "Micrometer-scale lateral structuring of organic thiolate layers through self-organisation.", Angew. Chem. Int. Ed. Engl., 33, 1274-1276 (1994)
- 170. Bhatia, S. K., Hickman, J. J. & Ligler, F. S. "New approach to producing patterned biomolecular assemblies.", J. Am. Chem. Soc., 114, 4432-4433 (1992)
- 171. Fodor, S. P. A., Read, J. L., Pirrung, M. C., Stryer, L., Lu, A. T. & Solas, D.
 "Light-directed spatially addressable parallel chemical synthesis.", *Science*, 251, 767-773 (1991)
- Peas, A. C., Solas, D., Sullivan, E. J., Cronin, M. T., Holmes, C. P. & Fodor, S. P.A.
 "Light-generated oligonucleotide arrays for rapid DNA sequence analysis.", *Proc. Nat. Acad. Sci. USA*, 91, 5022-5026 (1994)

- 173. Merrifield, B. "The synthesis of a tetra-peptide.", J. Am. Chem. Soc., 85, 2149-2145 (1963)
- 174. von Kiederowski, G. "Light-directed parallel synthesis of up to 250,000 different oligopeptides and oligonucleotides." *Angew. Chem. Int. Ed. Engl.*, **30**, 822-823, (1991)
- 175. Jacobs, J. W. & Fodor, S. P. A. "Combinatorial chemistry-Applications of light-directed chemical synthesis.", *Trends in Biotech.*, 12, 19-26 (1994)
- Lowe, C. R. & Earley, F. G. P. "Diagnostic device incorporating a biochemical ligand.", UK Patent Application No. 8413167 (1984)
- Rozsnyai, L. F., Benson, D. R., Fodor, S. P. A. & Schultz, P. G. "Photolithographic immobilisation of biopolymers on solid supports.", *Angew. Chem. Int. Ed. Engl.*, 31, 759-761 (1992)
- 178. Shimura, T., Nakamura, T., Kawakami, A., Haga, M. & Kato, Y. "A new type of enzyme immunosensor using antigen-bound membrane and multivalent antibody (F_{AB}'-alpha-amylase conjugate).", *Chemical and Pharmaceutical Bulletin*, 34, 5020-5025 (1986)
- Morales, P & Sperandei, M. "New method for deposition of biomolecules for bioelectronic purposes" Applied Physics Letters, 64, 1042-1044 (1994)
- 180. Morales, P., Pavone, A., Sperandei, M., Leter, G., Mosiello, L. & Nencini, L. "A laser assisted deposition technique suitable for the fabrication of biosensors and molecular electronic devices." Materials Science & Engineering C-Biomimetic Materials Sensors and Systems, 2,173-179, (1995)
- Livnah, O., Bayer, E. A., Wilchek, M. & Sussman, J. L. "Three dimensional structures of avidin and the avidin-biotin complex.", *Proc. Natl. Acad. Sci. USA.*, 90, 5076-5080 (1993)

- Hudson, L. & Hay, F. C. Practical Immunology, 3rd Edition, Blackwell Scientific (Oxford), (1989)
- Ikariyama, Y. & Aizawa, M. "Bioaffinity sensors.", in Methods in Enzymology Volume 137 (Ed. Mosbach, K.), Academic Press (London), 111-124 (1988)
- 184. Richards, F. M. "Reflections.", in *Methods in Enzymology Volume 184*(Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 3-5 (1990)
- 185. Green, N. M. "Avidin and streptavidin.", in Methods in Enzymology Volume 184 (Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 51-67 (1990)
- 186. Wilchek, M. & Bayer, E. A. "Introduction to avidin-biotin technology.", in Methods in Enzymology Volume 184 (Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 5-13 (1990)
- 187. Ebersole, R.C., Miller, J. A., Moran, J. R. & Ward, M. D. "Spontaneously formed functionally active avidin monolayers on metal surfaces: A strategy for immobilising biological reagents and design of piezoelectric biosensors.", J. Am. Chem. Soc., 112, 3239-3241 (1990).
- 188. Achtnich, U. R., Tiefenauer, L. X. & Andres, R. Y. "Covalent immobilisation of avidin on glassy carbon electrodes as the basis for multivalent biosensors.", *Biosensors*, 7, 279-290 (1992)
- 189. Ebato, H, Herron, J. N., Muller, W., Okahata, Y., Ringsdorf, H. & Suci, P.
 "Docking of a second functional protein layer to a streptavidin matrix on a solid support: Studies with a quartz crystal microbalance.", Angew. Chem. Int. Ed. Engl. 31, 1087-1090 (1992)
- Wilchek, M. & Bayer, E. A. "Applications of avidin-biotin technology. Literature survey.", in *Methods in Enzymology Volume 184* (Eds. Wilchek, M. & Bayer, E.A.), Academic Press (London), 14-45 (1990)

- 191. Hiller, Y., Bayer, E. A. & Wilchek, M. "Nonglycosylated avidin.", in Methods in Enzymology Volume 184 (Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 68-70 (1990)
- Hiller, Y., Gershoni, J. M., Bayer, E. A. & Wilckek, M. "Biotin binding to avidin. Oligosaccharide side chain not required for ligand association", *Biochem. J.*, 248, 167-171 (1987)
- 193. Dulhamel, R. C. & Whitehead, J. S. "Prevention of non-specific binding of avidin." in Methods in Enzymology Volume 184 (Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 201-207 (1990)
- 194. Hetland, O., Goransson, L. & Nilsen, D. W. T. "Cardiac troponin T immunoassay on biotin - streptavidin coated microplates. Preliminary performance in acute myocardial infarction.", Scand. J. Clin. Lab. Invest., 55, 701-713 (1995)
- 195. Forster, A. C., McInnes, J. L., Skingle, D. C. & Symons, R. H. "Non-radioactive hybridisation probes prepared by the chemical labelling of DNA and RNA with a novel reagent, photobiotin.", *Nucleic Acids Res.*, **13**, 745-761 (1985)
- Lacey, E. & Grant, W. N. "Photobiotin as a sensitive probe for protein labelling.", Anal. Biochem., 163, 151-158 (1987)
- 197. Pritchard, D. J., Morgan, H. & Cooper, J. M. "Micron scale patterning of biological molecules.", Angew. Chemie. Int. Ed. Engl., 35, 1-93 (1995)
- Pritchard, D. J., Morgan, H. & Cooper, J. M. "Immobilisierung von biomolekülen in zweidimensionalen mustern im mikrometermaßstab.", Angew. Chemie., 107, 84-86 (1995)
- 199. Morgan, H., Pritchard, D. J. & Cooper, J. M. "Selective immobilisation and patterning of molecules on a surface.", UK Patent Application No 9325100.7 (1993)

- 200. Smith, P. A. S. "Aryl and heteroaryl azides and nitrenes.", in Azides and Nitrenes. Reactivity and Utility. (Ed. Scriven, E. F. V.), Academic Press (London), 95-204 (1984)
- 201. Breslow, D. S. "Industrial applications.", in Azides and Nitrenes. Reactivity and Utility. (Ed. Scriven, E. F. V.), Academic Press (London), 491-521 (1984)
- 202. Reiser, A. & Wagner, H.M. "Photochemistry of the azido group.", in *The Chemistry* of the Azido Group, (Ed. Patai, S.), Wiley Interscience (New York), 441-501 (1971)
- 203. McInnes, J. L., Forster, A. C., Skingle, D. C. & Symons, R. H. "Preparation and uses of photobiotin.", in *Methods in Enzymology Volume 184* (Eds. Wilchek, M. & Bayer, E. A.), Academic Press (London), 588-600 (1990)
- 204. Bayley, H. & Staros, J. V. "Photoaffinity labeling and related techniques", in Azides and Nitrenes. Reactivity and Utility. (Ed. Scriven, E. F. V.), Academic Press (London), 433- 490 (1984)
- 205. Sheehan, J. C. & Hess, J. P. "A new method of forming peptide bonds.", J. Am. Chem. Soc., 77, 1067-1068 (1955)
- 206. Lynn, M. "Inorganic support intermediates." in Immobilised Enzymes, Antigens, Antibodies and Peptides: Preparation and Characterisation Vol 1. Enzymology. Ed.Weetall H. H., Marcel Dekker (New York), pp1-48 (1975)
- 207. Bauminger, S. & Wilchek, M. "The use of carbodiimides in the preparation of immunising conjugates." in *Methods in Enzymology Volume 70* (Ed. Mosbach, K.), Academic Press (London), 151-159 (1980)
- 208. Bavik, C. O., Busch, C. & Eriksson, U. "Characterisation of a plasma retinol-binding protein membrane receptor expressed in the retinal pigment epithelium", J. Biol. Chem., 267, 23035-23042 (1992)
- 209. Mellchers, F., Messer, W. "The activation of mutant beta-galactosidase by specific antibodies. Purification of eleven antibody activable mutant proteins and their subunits on sepharose immunosorbents." *Eur. J. Biochem.*, **17**, 267-272 (1970)
- Weintraub, B. D. "Concentration and purification of human chorionic somatommamotropin (HCS) by affinity chromatography. Application to radioimmunoassay.", *Biochem. Biophys. Res. Commun.*, 39, 83-89 (1970)
- 211. Binnig, G, Rohrer, H., Gerber, Ch, & Weibel, E., Phys. Rev. Lett., 49, 57 (1982)
- 212. Lindsay, S. "Biological applications of scanning probe microscopy", in Scanning Tunnelling Microscopy and Spectroscopy, Theory, Techniques and Applications, (Ed. Bonnel, D. A.), VCH Publishers (New York), 335-408 (1993)
- 213. Haussling, L., Michel, B., Ringsdorf, H. & Rohner, H."Direct observation of streptavidin specifically adsorbed on biotin-functionalised self assembled monolayers with the scanning tunneling microscope", Angew. Chem. Int. Ed. Engl., 30, 569-572 (1991)
- 214. Cooper, J. M., Shen, J., Young, F. M., Connolly, P., Barker, J. R. & Moores, G.
 "The imaging of streptavidin and avidin using scanning tunnelling microscopy", J.
 Mat. Sci.: Materials in Electronics, 5, 106-110 (1994)
- 215. Connolly, P., Moores, G. R., Monaghan, W, Shen, J., Britland, S. & Clark, P.
 "Microelectronic and nanoelectronic interfacing techniques for biological systems.", Sensors & Actuators B, 6, 113-121 (1992)
- Weisenhorn, A. L., Schmitt, F-J., Knoll, W. & Hansma, P. K. "Streptavidin binding observed with an atomic force microscope.", *Ultramicroscopy*, 42-44, 1125-1132 (1992)
- Perssen, B. N. J. "The atomic force microscope-Can it be used to study biological molecules", Chem. Phys. Lett., 141, 366-368 (1987)

- 218. Wilchek, M. & Bayer, E. A., "The avidin-biotin complex in bioanalytical applications.", *Anal. Biochem.*, **171**, 1-32 (1988)
- Silverton, E. W., Navia, M. A. & Davies, D. R. "Three dimensional crystal structure of an intact human immunoglobulin.", *Proc. Natl. Acad. Sci. USA*, 74, 5140-5144 (1977)
- 220. Lehninger, A. L. Principles of Biochemistry, Worth Publishers (New York), 726 (1982)
- 221. Butt, W.R. "Gonadotrophins." in *Hormones in Blood. Volume 1*, (Eds. Gray, C.H. & James, V.H.T.), 411-471. (1979)
- 222. Gowenlock, A. H., Varleys Practical Biochemistry. Sixth edition, Heinemann (London) (1988)
- 223. Hall, R., Anderson, J., Smart, G. A. & Besser, M., Fundamentals of Clinical Endocrinology, Third Edition, Pitman Medical (London) (1980)
- 224. Wide, L., Nillius, S. J., Gemzell, C., & Roos, P. "Radioimmunosorbent assay of follicle stimulating hormone and luteinising hormone in serum and urine from men and women." Acta Endocr. 73 Suppl. 174, 1-58, (1973)
- 225. Whitby, L. G., Smith, A. F. and Becket, G. J. "Gonadal function." in *Lecture Notes* on Clinical Chemistry. 4th edition, Blackwell Scientific (Oxford), 354-362 (1988)
- 226. Wright, J. W. "Disorders of the reproductive system." in *Biochemistry in Clinical Practice*. 2nd edition, (Eds. Williams, D & Marks, V.), Butterworth Heinemann (Oxford), 707-721 (1994)
- 227. Greep,R.O., Van Dyke,H.B. & Chow,B.F. "Use of anterior lobe of prostate gland in the assay of metakentin.", *Proc.Soc. Exp. Biol.Med.*, 46, 644-649 (1941)

- 228. Watts, R.M. & Adair, F.L. "Excretion of estrogen and gonadotrophin in late pregnancy.", Am. J. Obst. Gynec., 46, 183-207 (1943)
- 229. Van Hell, H., Matthijsen, R. & Overbeek, G.A. "Effects of human menopausal gonadotrophin preparations in different bioassay methods.", *Acta Endocr.*, 47, 409-418 (1964)
- 230. Parlow, A.F. "Bioassay of pituitary luteinising hormone by depletion of ovarian ascorbic acid." in *Human Pituitary Gonadotrophins*, C. C. Thomas, Springfield (Illinois), 300-310 (1961)
- 231. Steelman, S. L., Pohley, F. M. "Assay of the follicle stimulating hormone based on the augmentation with human chorionic gonadotrophin.", *Endocrinology*, 53, 604-616 (1953)
- 232. Brown, P. S. "The assay of gonadotrophin from urine of non-pregnant human subjects.", J. Endocrinology, 13, 59-64 (1955)
- 233. Watson, J. "The bioassay of luteinising hormone by progesterone synthesis in ovarian tissue of prolactin-treated rats in-vivo.", J. Endocr., 50, 711-712 (1971)
- 234. Shirley, A. & Stephenson, J., "Progestin production in vitro by rabbit granulosa cells as an assay for luteinising hormone.", J. Endocr., 58, 345-346 (1973)
- 235. Moyle, W, R. & Ramachandran, J. "Effect of LH on steroidogenesis and cyclic AMP accumulation in rat Leydig cell preparations and mouse tumour Leydig cells.", *Endocrinology*, 93, 127-134 (1973)
- 236. Van Damme, M. P., Robertson, D. M. & Diczfalusy, E. "An improved in vitro bioassay method for measuring luteinising hormone (LH) activity using mouse Leydig cell preparations.", *Acta Endocr.*, 77, 655-671 (1964)

- 237. Qazi, M. H., Romany, P. & Diczfalusy, E. "Discrepancies in plasma LH activities as measured by radioimmunoassay and an *in vitro* bioassay.", *Acta Endocr.*, 77, 672-685 (1974)
- 238. Dufau, M. L., Hodgen, G. D., Goodman, A. L. & Catt, K. J. "Bioassay of circulating luteinising hormone in the rhesus monkey: Comparison with radioimmunoassay during physiological changes.", *Endocrinology*, 100, 1557-1565 (1977)
- 239. Picon, R. & Habert, R. "A sensitive bioassay for luteinising hormone-like activity applied to systemic plasmas of foetal rats.", *Acta Endocr*, **97**, 176-180 (1981)
- 240. Rees, L. H., Holdaway, I. M., Kramer, R., McNeilly, A. S. & Chard, T. "New bioassay for luteinising hormone.", *Nature*, 244, 232-234 (1973)
- 241. Van Damme, M. P., Robertson, D. M., Marana, R. Ritzen, E. M. & Diczfalusy, E.
 "A sensitive and specific *in vitro* bioassay method for the measurement of follicle stimulating hormone activity.", *Acta Endocr.*, 91, 224-237 (1979)
- 242. Catt, K. J., Dufau, M. L. & Tsuruhara, T. "Radioligand-receptor assay of luteinising hormone and chorionic gonadotrophin.", J. Clin. Endocr. Metab., 34, 123-132 (1972)
- 243. Leidenberger, F. L., Reichert, L. E. "Evaluation of a rat testis homogenate radioligand receptor assay for human pituitary LH.", *Endocrinology*, 91, 901-909 (1972)
- 244. Leidenberger, F. L., Willaschek, R., Pahnke, V. G. & Reichert, L. E. "Application of a radioligand receptor assay for determination of luteinising hormone in human serum.", *Acta Endocr.*, 81, 54-72 (1976)
- 245. Reichert, L. E. & Bhalla, V. K. "Development of a radioligand tissue receptor assay for human follicle stimulating hormone.", *Endocrinology*, 94, 483-491 (1974)

- Cheng, K. W. "A radioreceptor assay for follicle-stimulating hormone.", J. Clin. Endocr. Metab., 41, 581-589 (1975)
- 247. Miyai, K. "Classification of immunoassay", in Principles and Practice of Immunoassay (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 246-264 (1991)
- Bieglmayer, C. & Fischl, F. "Gonadotrophin measurements with fast, non-radioactive methods.", J. Clin. Chem. & Clin. Biochem, 25, 747-750 (1987)
- 249. Midgley, A. R. "Radioimmunoassay: A method for human chorionic gonadotrophin and human luteinising hormone.", *Endocrinology*, **79**, 10-18 (1966)
- 250. Midgley, A. R. "Radioimmunoassay of human follicle-stimulating hormone.", J. Clin. Endocrin & Metab., 27, 295-299 (1967)
- 251. Fairman, C. & Ryan, R. J. "Radioimmunoassay of human follicle stimulating hormone.", J. Clin. Endocrin & Metab., 27, 444-449 (1967)
- 252. Odell, W. D., Ross, G. T. & Rayford, P. L. "Radioimmunoassay for luteinising hormone in human plasma: Physiological Studies.", J. Clin. Invest., 46, 248-255 (1967)
- 253. Lynch, S. S. & Shirley, A. "Production of specific antisera to follicle-stimulating hormone and other hormones.", J. Endocr., 65 127-132 (1975)
- 254. Bagshawe, K. D., Wilde, C. E. & Orr, A. H. "Radioimmunoassay for human chorionic gonadotrophin and luteinising hormone.", *Lancet*, 1, 1118-1121 (1966)
- 255. Saxena, B. B., Demura, H., Gandy, H. M. & Peterson, R. E. "Radioimmunoassay of human follicle-stimulating and luteinising hormones in plasma.", J. Clin. Endocrin & Metab., 28, 519-534 (1968)

- Edwards, R. "Radiolabelled immunoassay." in Principles and Practice of Immunoassay (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 265-294 (1991)
- 257. Goncharov, N. P., Krylin, V. V., Vorontsov, V & I., Kolesnikova, G. S.
 "Comparative characterisation of radioimmunologic and non-isotopic methods of hormonal analysis", *Probl. Endokrinol. Mosk.*, 38, 26-27 (1992)
- 258. Edwards, R. "Radiolabelled immunoassay.", in Principles and Practice of Immunoassay 2nd Edition (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 325-348 (1997)
- 259. Prentice, L. G. & Ryan, R. J. "LH and its subunits in human pituitay, serum and urine." J.Clin.Endocr.Metab., 40, 303-312, (1975)
- 260. Menjivar, M., Ortiz, G., Cardenas, M. & Garza-Flores, J. "Comparison of the DELFIA and RIA methods for measuring luteinizing and follicle stimulating hormones in serum.", *Rev. Invest. Clin.*, 45, 579-584 (1993)
- 261. Madersbacher, S., Shu-Chen, T., Schwarz, S., Dirnhofer, S., Wick, G. & Berger, P. "Time-resolved immunofluorometry and other frequently used immunoassay types for follicle stimulating hormone compared by using identical monoclonal antibodies.", *Clin. Chem.*, **39**, 1435-1439 (1993)
- 262. Vermes, I., Bonte, H. A., van der Sluijs-Veer, G. & Schoemaker, J. "Interpretation of five monoclonal immunoassays of lutotropin and follitropin: effects of normalisation with WHO standard", *Clin. Chem.*, 37, 415-421 (1991)
- 263. Forrest, G. C. "A general review of automated RIA." in Immunoassay for Clinical Chemistry, (Eds. Hunter, W. M. & Corrie, J. E. T.) Churchill-Livingstone (Edinburgh), 211-220 (1983)

- Ismail, A. A. A., West, P. M. & Goldie, D. J. "The "Southmead System": A simple fully-automated continuous flow system for immunoassays.", *Clin. Chem.*, 24, 571-579 (1978)
- 265. Gorman, E. G., Arentzen, R., Dezyk, W. & Cassidy, L. A. "An overview of immunoassay automation.", in *Principles and Practice of Immunoassay 2nd Edition* (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 299-323 (1997)
- 266. Miles, L. E. M. & Hales, C. N. "Labelled antibodies and immunological assay systems.", *Nature*, 219, 186-189 (1968)
- 267. Benkirane, M. M., Delori, P., Lebec, S. Cordeil, M. & Delaage, M. A. "Production of monoclonal antibodies complementary to an antibody-antigen complex. Use in an immunoradiometric assay for follitropin.", J. Immunol. Meth., 111, 189-194 (1992)
- 268. Hunter, W. M., Bennie, J. G., Kellett, H. A., Micklem, L. R., Scott, A. & James, K.
 "A monoclonal antibody based immunoradiometric assay (IRMA) for h-LH", Ann. Clin. Biochem., 21, 275-283 (1984)
- 269. Berger, P., Panmoung, W, Kaschabi, D., Mayregger, B & Wick, G. "Antigenic features of human follicle stimulating hormone delineated by monoclonal antibodies and construction of an immunoradiometric assay.", *Endocrinology*, **123**, 2351-2359 (1988)
- 270. Kamariah, K., Lopez, J. B. & Satgunasingam, N. "Immunoassay of gonadotrophins using a fully-automated benchtop analyser.", Br. J. Biomed. Sci., 51, 296-298 (1994).
- 271. Kobayashi, M. Sakai, R. & Kanai, M. "A study on LH-and FSH-RIA kits by immunoradiometric assay.", *Radioisotopes*, **39**, 45-48 (1990)
- 272. Morgan, C. R. "Immunoassay of human insulin and growth hormone simultaneously using ¹³¹I and ¹²⁵I tracers.", *Proc. Soc. Exp. Biol. Med.*, **123**, 230-233 (1966)

- 273. Kricka, L., Thorpe, G. & Stott, R. "Luminescence Immunoassay." in Principles and Practice of Immunoassay (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 417-445 (1991)
- 274. Lovesey, A. C., "Solid phase immunoassay for human plasma follicle stimulating hormone.", *Ann. Clin. Biochem.*, **17**, 95-100 (1980)
- 275. Varalli, M., Bonnazi, M. T., Salcini, A., Cazzola, E., & Militano, D."Analytical performance of an heterogeneous enzyme immunoassay using the same separation system.", *Boll. Chim. Farm.*, 132, 275-289 (1993)
- 276. Dias, J. A., Driskell, W. J. & Reichert, L. E. "An enzyme immunoassay for human follicle stimulating hormone.", *Anal. Biochem.*, **114**, 268-275 (1981)
- 277. Huguet, J., Bonnin, M. R., Guillen, E. & Navarro, M. A., "A fluorimetric enzyme immunoassay for follitropin and lutotropin.", Eur. J. Clin. Chem. & Clin. Biochem., 29, 565-567 (1991)
- 278. Armbruster, D. A. & Hawes, L. C. "Assay of follitropin and lutotropin by fluorescence enzyme immunoassay.", J. Clin. Lab. Anal., 4, 170-174 (1990)
- 279. Price, C. P. & Newman, D. J. "Homogeneous fluoroimmunoassay "in Principles and Practice of Immunoassay (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 393-416 (1991)
- 280. Lövgren, T., Hemmilla, I., Petterson, K., Eskola, J. U. & Bertoft, E. "Determination of hormones by time-resolved fluoroimmunoassay.", *Talanta*, 31, 909-916 (1984)
- 281. Suonpaa, M., Markela, E., Stahlberg, T. & Hemmila, I. "Europium-labelled streptavidin as a highly sensitive universal label. Indirect time-resolved immunofluorometry of FSH and TSH.", J. Immunol. Meth., 129, 247-253 (1992)

- 282. Biro, G., Samira, A., Butz, H., Leicht, E. & Weinges, K. F. "Development of a luminescence immunoassay for follitropin suitable for clinical routine.", *Eur. J. Clin. Chem. & Clin. Biochem.*, 29, 182-192 (1991)
- Sankolli, G. M., Lynch, S. S., Rudd, B. T., Thorpe, H. H., Stott, R. A. & Kricka, L.
 J. "An enhanced chemiluminescent enzyme immunoassay for follicle stimulating hormone.", J. Immunoassay, 10, 201-219 (1989)
- 284. Brockelbank, J. L., Kim, J. B., Barnard, G. J., Collins, W. P., Gaier, B. & Kohen, F.
 "The measurement of urinary LH, by a solid phase chemiluminescence immunoassay.", Ann. Clin. Biochem., 21, 284-289 (1984)
- 285. Terouanne, B, Alameddine, S., Martin, J. L., Nicolas, J. C., Cristol, P., Sultan, C., & Crastes-de-Paulet, S. O. "Bioluminescence assay of luteinising hormone in plasma and urine.", Ann. Biol. Clin. Paris, 47, 15-21 (1989)
- 286. Chiu, T. T., Tam, P. P. & Mao, K. R. "Evaluation of a semi-quantitative urinary LH assay for ovulation detection.", *Int. J. Fertil.*, 35, 120-124 (1990)
- 287. Price, C. P., Thorpe, G. H. G., Hall, J., & Bunce R. A. "Disposable integrated Immunoassay devices.", in *Principles and Practice of Immunoassay, 2nd Edition.*", (Eds Price, C. P. & Newman, D. J.), Stockton Press. (New York), 579-603 (1997)
- 288. May, K. "Unipath ClearBlue One Step[™], ClearPlan One Step[™], and Clearview[™]." in *The Immunoassay Handbook*, (Ed Wild, D.), MacMillan Press (London), 233-235 (1994)
- 289. Bard, A. J. & Faulkner, L. R., Electrochemical Methods., Wiley (New York), (1980)
- 290. Frew, J.E., Harmer, M.A., Hill, H.A.O. & Libor, S.I. "A method for estimation of hydrogen-peroxide based on mediated electron-transfer reactions of peroxides at electrodes.", *J.Electroanal.Chem.* 201, 1-10 (1986)

- 291. Lovgren, T., Hemilla, I., Petterson, K., Eskola, J.U. & Bertoft, E. "Determination of hormones by time-resolved fluoroimmunoassay." *Talanta* 31:909-916 (1984)
- 292. Pritchard, D. J., Morgan, H. & Cooper, J. M. "Simultaneous determination of follicle stimulating hormone and luteinising hormone using a multianalyte immunosensor.", Anal. Chim. Acta., 310, 251-256 (1995)
- 293. Stenman, U.-H., Alfthan, H., Koskimies, A., Seppalla, M., Petterson, K., & Lovgren, T., "Monitoring the LH surge by ultrarapid and highly sensitive immunofluorimetric assay." Ann .NY Acad.Sci. 442, 544-550 (1985)
- 294. Pritchard, D.J., Morgan, H., and Cooper, J.M. "Patterning and regeneration of surfaces with antibodies". *Anal. Chem.* 67, 3605-3607, (1995)
- 295. Ekins, R., Chu, F. & Biggart, E. "Development of microspot multi-analyte ratiometric immunoassay using dual fluorescent-labelled antibodies.", Anal. Chim. Acta, 227, 73-96 (1989)
- 296. Ekins, R., Chu, F. & Biggart, E. "Multianalyte immunoassay: The immunological 'compact disk' of the future.", J. Clin. Immunoassay, 13, 169-181 (1990)
- 297. Ekins, R. "Immunoassay design and optimisation." in *Principles and Practice of Immunoassay* (Eds. Price, C. P. & Newman, D. J.), Stockton Press (New York), 96-153 (1991)
- Heinze, J. "Ultramicroelectrodes in Electrochemistry.", Angew. Chem. Int. Ed. Engl., 32, 1268-1288 (1993)
- 299. Penner, R. M. & Lewis, N. S. "Invisible electrodes with amazing powers", Chemistry & Industry, 788 - 791 (1991)
- 300. Aoki, K. "Theory of ultramicroelectrodes.", *Electroanalysis*, 5, 627-639 (1993)

- 301. Dahlback, B., Carlsson, M. & Svennson, P. J. "Familial thrombophilia due to a previously unrecognised mechanism characterized by poor anticoagulant response to activated Protein C: Prediction of a cofactor to activated protein C.", *Proc. Nat. Acad. Sci. U. S. A.*, 90, 1004-1008 (1993)
- 302. Bertina, R. M., Koelman, B. P. C., Koster, T., Rosendaal, F. R., Dirven, R. J., De-Ronde, H. Van-Der-Velden, P. A. & Reitsma, P. H. "Mutation in blood coagulation Factor V associated with resistance to activated protein C.", *Nature*, 369, 64-67 (1994)
- 303. Poort, S. R., Rosendaal, F. R., Reitsma, P. H. & Bertina, R. M. "A common genetic variation in the 3'-untranslated region of the prothrombin gene is associated with elevated prothrombin levels and an increase in venous thrombosis.", *Blood*, 88, 3698-3703 (1996)
- 304. Silverman, L. M. & Christenson, R. H. "Amino acids and proteins.", in *Tietz Textbook of Clinical Chemistry, 2nd Edition*, (Eds. Burtis, C. A. & Ashwood, E.R.),
 W. B. Saunders (Philadelphia), 663-667 (1994)
- 305. Frosst, P., Blom, H. J., Goyette, P, Sheppard, C. A., Mathews, R. G. & Boers, H. H. J. "A candidate genetic risk factor for cardiovascular disease: a common mutation in methylenetetrafolate reductase.", *Nature Genetics*, 10, 111-113 (1995)
- 306. den Heijer, M., Blom, H. J., Gerrits, W. B. J., Rosendaal, F. R., Haak, H. L., Wijermans, P. W. & Bos, G. M. J. "Is hyperhomocysteinaemia a risk factor for recurrent thrombosis?", *Lancet*, 345, 882-885 (1995)
- 307. den Heijer, M., Koster, T., Blom, H. J., Bos, G. M. J., Briët, E., Reitsma, P. H., Vanderbrouke, J. P. & Rosendaal, F. R. "Hyperhomocysteinaemia as a risk factor for deep vein thrombosis.", N. Engl. J. Med., 334, 759-762 (1996)

- 308. Salomon, O, Zivelin, A., Dardik, R., Inbal, A., Varon, D., Martinowitz, U., Mani, A., Lubetsky, A., Steinberg, D. M. & Seligsohn, U. "Risk assessment of single, double and triple prothombotic polymorphisms in patients with idiopathic venous thrombembolism.", *Blood, Abstracts of 39th Annual meeting of American Society of Hematology*, A1125 (1997)
- 309. Mullis, K. B. & Faloona, F. A. "Specific synthesis of DNA in vitro via a polymerase-catalysed chain reaction.", Methods Enzymol., 155. 335-350 (1986)
- Saiki, P. K., Gelfand, D. H., Stoffel, S., Scharf, S., J., Higuchi, R., Horn, G. T.,
 Mullis, K. B. & Ehelich, H. A. "Primer-directed enzymatic amplification of DNA *in vitro*: the polymerase chain reaction, *Science*, 239, 487-491 (1988)
- 311 Morgan, H. Unpublished Work (1995)
- 312. Lacy, W. B., Vittal, V. R., Turner, D. C., Conrad, D. W. & Gaber, B. P. "An LFM biosensor based on protien patterning of photoreactive PEG-silane and photobiotin surfaces", Abstracts of American Chemical Society, 216, 39 (1998)
- 313. Dontha, N., Nowall, W. B. & Kuhr, W. G. "Generation of biotin/avidin/enzyme nanostructures with maskless photolithography.", Anal. Chem., 69, 2619-2625 (1997)
- 314. Nowall, W. B., Dontha, N. & Kuhr, W. G. "Electron transfer kinetics at a biotin/avidin patterned glassy carbon electrode.", *Biosensors*, 13, 1237-1244 (1998)
- 315. Dontha, N., Nowall, W. B. & Kuhr, W. G. "Development of sub-micron patterned carbon electrodes for immunoassays.", J. Pharm. Biomed. Anal., 19, 83-91 (1999).
- 316. Hengsakul, M. & Cass, A. E. G. "Protein Patterning with a photactivable derivative of biotin.", *Bioconjugate Chemistry*, 7, 249-254 (1996)

- 317. Sunberg, S. A., Barrett, R. W., Pirrung, M., Lu, A. L., Kiangsoontra, B. & Holmes,
 C. P. "Spatially-addressable immobilisation of macromolecules on solid supports.",
 J. Am. Chem. Soc., 117, 12050-12057 (1995)
- 318. Delamarche, E., Bernard, A., Schmid, H. Michel, B. & Biebuyck, H. "Patterned delivery of immunoglobulins to surfaces using microfluidic networks.", Science, 276, 779-781 (1997)
- 319. Kapur, R, Spargo, B. J., Chen, M. S., Calvert, J. M. & Rudolph, A. S. "Fabrication and selective surface modification of 3-dimensionally textured biomedical polymers from etched silicon substrates", J. Biomed. Mater. Res., 33, 205-216 (1996)
- 320. Onnerffjord, P., Nilsson, J., Wallman, L. Laurell, T. Marko-Varga, G. "Picoliter sample preparation in MALDI-TOF MS using a micromachined silicon flowthrough dispenser.", Anal. Chem., 70, 4755-4760 (1998)
- Brooks, A. J., Halterman, M. W., Chadwick, C. A., Davidson, B. L.,
 Haak-Frendscho, M., Radel, C. Porter, C. & Federoff, H. J. "Reproducible and efficient murine CNS gene delivery using a microprocessor-controlled injector.", J. Neurosci. Meth., 80, 137-147, (1998)
- 322. Nilsson, P., Persson, B., Uhlen, M. & Nygren, P. A. "Real-time monitoring of DNA manipulations using biosensor technology.", *Anal. Biochem.*, **224**, 400-408 (1995)
- 323. Marrazza, G., Chianella, I. & Mascini, M. "Disposable DNA electrochemical sensor for hybridization detection.", *Biosensors*, 14, 43-51 (1999)
- 324. Wang, J., Cai, X., Rivas, G., Shiraishi, H. & Dontha, N. "Nucleic acis immobilization, recognition and detection at chronopotentiometric DNA chips.", *Biosensors*, 12, 587-599 (1997)
- 325. Sugawara, T. & Matsuda, T. "Photochemical surface derivatization of a peptide containing Arg-Gly-Asp (RGD).", J. Biomed. Mater. Res., 29, 1047-1052 (1995)

- 326. Van Nostrand's Scientific Encyclopaedia, 7th Edition, Ed. Van Nostrand Reinhold (New York), (1997)
- 327. Durig, U., Pohl, D. W. & Rohner, F. "Near-field optical scanning microscopy.", J. Appl. Physics, 59, 3318-3327 (1986)
- 328. Fischer, U. C., Durig, U. T. & Pohl, D. W. "Near-field optical scanning microscopy in reflection.", *Appl. Phys. Lett.*, **52**, 249-251 (1988)
- 329. Schemelinin, A., Lieberman, K. & Lewis, A. "Nanometer resolution near-field optical lithography.", *Abstracts of American Vacuum Society 41st National Symposium*, (1994)
- 330. Shchemelinin, A., Rudman, M., Lieberman, K.& Lewis, A. "A simple lateral force sensing technique for near-field micropattern generation." *Review of Scientific Instruments*, 64, 3538-3541 (1993)
- 331. Rudman, M, Lewis, A, Mallul, A, Haviv, V, Turovets, I, Shchemelinin, A & Nebenzahl, I "Near-field subwavelength micropattern generation - pipette guided argon fluoride excimer laser microfabrication." J of Applied Physics, 72, 4379-4383 (1992)